DESIGN AND ANALYSIS OF RECONFIGURABLE RF FRONT END AMPLIFIERS FOR WIRELESS BROADBAND MULTIMEDIA COMMUNICATION

Thesis

Submitted in partial fulfillment of the requirements for the degree of DOCTOR OF PHILOSOPHY

by VIGNESH R



DEPARTMENT OF ELECTRONICS AND COMMUNICATION ENGINEERING, NATIONAL INSTITUTE OF TECHNOLOGY KARNATAKA, SURATHKAL, MANGALORE -575025

AUGUST, 2022

DECLARATION

I hereby *declare* that the Research Thesis entitled **DESIGN AND ANALYSIS OF RECONFIGURABLE RF FRONT END AMPLIFIERS FOR WIRE-LESS BROADBAND MULTIMEDIA COMMUNICATION** which is being submitted to the *National Institute of Technology Karnataka, Surathkal* in partial fulfillment of the requirement for the award of the Degree of *Doctor of Philosophy* in **Department of Electronics and Communication Engineering** is a *bonafide report of the research work carried out by me*. The material contained in this research thesis has not been submitted to any University or Institution for the award of any degree.

Reg. No.: 187115/187EC017 Department of Electronics and Communication Engineering.

Place: NITK-Surathkal. Date: 5-08-2022

CERTIFICATE

This is to certify that the Research Thesis entitled **DESIGN AND ANAL-YSIS OF RECONFIGURABLE RF FRONT END AMPLIFIERS FOR WIRELESS BROADBAND MULTIMEDIA COMMUNICATION** submitted by **VIGNESH R** (Register Number: 187115/187EC017) as the record of the research work carried out by him, is accepted as the *Research Thesis submission* in partial fulfillment of the requirements for the award of degree of **Doctor of Philosophy**.

Dr. Sandeep Kumar Research Guide Assistant Professor Dept. of E & C Engg. NITK Surathkal - 575025

Chairman-DRPC

प्राध्याप्तिंहएवर सिम्ना आर्थ्यक्ष Date or A Beal) इी एवं सी विश्वाग / E & C Department एन आई टी के, सुरतकल/NITK. Surathkal मगलूर / MANGALORE - 575 025 •

Acknowledgements

In the course of my PhD journey, I'd want to take a moment to appreciate the several individuals who have selflessly assisted me along the way. I want to offer my heartfelt gratitude to Dr. Sandeep Kumar, my research supervisor. He has always been exceedingly encouraging and invested far more trust in me than I deserved. I am incredibly grateful for the excellent technical and personal advice he has given me over the years. There are times when I feel dejected, inferior, and demotivated. In those times, he just planted trust and hope in my heart to move and rise further. Also, I admire how he treats fellow students like me, and he is such a down-to-earth person and super friendly among his students. I have never experienced this much kind, humble, super friendly, and helpful person as a supervisor in my life. Without his invaluable instruction, assistance, and encouragement, this dissertation would not have been feasible. I am blessed by God to have him as my research supervisor.

I want to express my deepest gratitude to Prof. Hanjung Song, Department of Nanoscience and Engineering, Centre for Nano Manufacturing, Inje University, Gimhae, South Korea, for allowing me to use the lab resources and helping me in chip fabrication and measurement. Without his support and assistance, this research would not have been completed.

I want to express my gratitude to Prof. Laxminidhi. T, the former Head of the Department of Electronics and Communication Engineering, for his guidance and administrative assistance. I am thankful to Prof. Ashvini Chaturvedi, the current Head of Electronics and Communication Engineering Department, for his time and support. Along with him, I'd like to express my gratitude to Dr. Shyam Lal of the Electronics and Communication Engineering department and Dr. Yashwanth Kashyap of the Electrical and Electronics department for their insightful suggestions on how to improve the content and quality of my research work. I thank all of my fellow research scholars at NITK for their continuous support. I would also like to take this opportunity to express my gratitude to the rest of the faculty and staff of the E&C department at NITK Surathkal.

I want to express my heartful gratitude to my spiritual guru Smt. Desa Mangaiyarkarasi for her teachings of balancing the mind, body, emotions,

and soul through the ancient sacred Tamil Saiva literature poems. Her teachings uplifted me in all aspects of my life and multiplied my performance manifold. I am also indebted to her for the impact she created in my life, and it transformed me into a blissful person.

I want to express my sincere gratitude to my amazing parents R. Akilandammal and KS. Rajhamanicgam for their endless love, sacrifice, and support throughout my life. I want to offer my heartful appreciation to my wonderful sister Devekala for her love and support. I also want to express my gratitude to my uncles Palaniguru and Kannan and my aunts Lakshmi and Rajakumari for their moral support throughout my life. My cousins Suthaarshiny, Varshe, and Aravindh, have been a great source of joy. I am blessed to have them all in my life.

Finally, I want to express my deepest gratitude to my God Murugan, Rishis, Yogis, Siddhars, especially Arunagirinathar, Manikavasakar, Thirumoolar, for blessing me with excellent health, strength, and peace all the time. Dedicated to My Dear Parents

iv

Abstract

This research work focuses on the design and implementation of Microwave Low Noise Amplifiers (LNAs) with built-in techniques for wireless broadband multimedia communications. The RF front-end blocks account for the majority of the system's Noise Figure (NF). As a result, low noise designs are being researched to improve the device's signal-to-noise ratio (SNR). Enhancing the SNR and linearity of the front-end amplifiers significantly boosts receiver performance. The proposed LNAs' primary objective is to produce low NF, high flat gain, and excellent linearity while minimizing power consumption at the microwave and millimeter (mm) wave bands of operation.

Firstly, a Suspended Substrate Line (SSL) integrated CMOS LNA is proposed for low-power 5G wireless receiver systems which are operated in the range of 28GHz to 32GHz. It comprises two stages cascode topology for high gain, better reverse isolation, and improved stability at mm-wave frequencies. A novel SSL-based parallel series network is incorporated between the two cascode stages to provide the required network matching and help in enhancing the bandwidth. Prior to incorporation into the design, the proposed SSL is designed and characterized for optimum EM compatibility. The proposed CMOS LNA, which incorporates these novel approaches, will overcome the device's parasitic and electromagnetic (EM) losses. EM losses in bulk active and passive components have been incurred utilizing built-in approaches to improve LNA linearity. The proposed mm-wave LNA enables each component in its family to prevent EM wave leakage and its associated parasitic losses in the layout. The simulation and measurement results demonstrate the minimum NF of 2.5dB, the maximum gain of 25dB, and high linearity and stability in the desired band of operation.

Secondly, A novel Quasi circulator (QC) integrated LNA operating in 8GHz to 12GHz (X band) is implemented using a two-stage inductive degeneration common source (CS) topology with current reused techniques and variable impedance load. QC provides the minimum insertion loss of 0.9dB with good return and isolation losses. Incorporating QC and optimizing the LNA, the highest flat gain of 30dB with only 0.5dB variation across the entire 4GHz bandwidth and minimum NF of 1dB are achieved simultaneously. The proposed QC-LNA is highly linear, and third-order harmonics are suppressed considerably. It consumes low power and is highly compact.

Thirdly, an integrated LNA with a Folded Butterfly Stub Stepped Impedance Resonator (FBSSIR) is also developed by employing packaging technology for a satellite navigation receiver system in the range of 1.6GHz to 2.5GHz. The proposed FBSSIR achieves a more compact structure, controllable transmission zero, adjustable center frequency, and adjustable bandwidth by applying a new structural deformation of a stepped-impedance-resonator (SIR). The core LNA circuit incorporates the proposed FBSSIR as it acts as a filter and the input-output matching network. The proposed FBSSIR-LNA achieves good gain, decent NF, and good linearity by consuming decent power.

Finally, a switch-free dual-band reconfigurable LNA in the range of 18GHz to 40GHz (K/Ka-band) is also designed for ultra-wideband wireless applications. It is realized by incorporating inter-stage and output-stage Suspended-Substrate Coupled-Lines (SSCL). The cascaded CS topology with source degeneration is used in the respective amplifier stages. The proposed inter-stage SSCL splits the amplified input signal from the broadband driving stage into two parallel single band stages. The corresponding High-band (Ka) and Low-band (K) stages amplify two split-band signals. At the output, the proposed SSCL output stage combines the amplified two single bands. Also, the proposed SSCL provides the necessary network matching to the LNA. A single band of operation can be obtained by simply shutting off the drain voltage of the unused transistor band. The proposed LNA achieves an incredibly flat gain of 27dB, ultra-low NF of 1.2dB, and best linear performance with good third-order harmonics suppression at the same time.

All the proposed CMOS LNAs are fabricated using a commercial 65nm CMOS process operating with 1.2V supply. DoE and statistical analysis are additional novel contributions towards this thesis work for analyzing

the proposed LNA designs. Design of Experiment (DoE) is a new analysis technique to find the individual device parameter's contribution to the final gain, NF, and return loss. Statistical analysis is also performed to find the yield so that the robustness of the proposed designs is satisfied.

Keywords: Low Noise Amplifier; Suspended Substrate Line; Quasi Circulator; SSCL; FBSSIR;

viii

Contents

	Ack	nowledgements	i
	Abst	tract	V
	List	of figures	i
	List	of tables	i
	Non	nenclature	ĸ
	Abb	reviations	¢
1	INT	TRODUCTION	L
	1.1	RF Frontend	2
	1.2	Issues in RF Frontend	3
	1.3	Motivation	5
	1.4	Objectives	3
	1.5	Organization and contribution of thesis	3
2	\mathbf{LN}	A PERFORMANCE METRICS AND ANALYSIS)
2	LN 2.1	A PERFORMANCE METRICS AND ANALYSIS 9 Introduction 9)
2	LN 2.1 2.2	A PERFORMANCE METRICS AND ANALYSIS 9 Introduction 9 Noise 10)
2	LN 2.1 2.2	A PERFORMANCE METRICS AND ANALYSIS 9 Introduction 9 Noise 10 2.2.1 Thermal noise 11	• • •
2	LN 2 2.1 2.2	A PERFORMANCE METRICS AND ANALYSIS 9 Introduction 9 Noise 10 2.2.1 Thermal noise 11 2.2.2 MOS devices dominant noise sources 11))) 1 1
2	 LN2 2.1 2.2 2.3 	A PERFORMANCE METRICS AND ANALYSIS 9 Introduction 9 Noise 10 2.2.1 Thermal noise 11 2.2.2 MOS devices dominant noise sources 11 Basic LNA Topologies 13))
2	LN22.12.22.3	A PERFORMANCE METRICS AND ANALYSIS 9 Introduction 9 Noise 10 2.2.1 Thermal noise 11 2.2.2 MOS devices dominant noise sources 11 Basic LNA Topologies 12 2.3.1 Common source Topology 13))
2	LN22.12.22.3	A PERFORMANCE METRICS AND ANALYSIS 9 Introduction 9 Noise 10 2.2.1 Thermal noise 11 2.2.2 MOS devices dominant noise sources 11 Basic LNA Topologies 13 2.3.1 Common source Topology 13 2.3.2 Common Gate Topology 14	I I <t< td=""></t<>
2	LN22.12.22.3	A PERFORMANCE METRICS AND ANALYSIS9Introduction9Noise102.2.1Thermal noise112.2.2MOS devices dominant noise sources12Basic LNA Topologies132.3.1Common source Topology132.3.2Common Gate Topology142.3.3Cascode Topology16	I I <td< td=""></td<>
2	LN22.12.22.3	A PERFORMANCE METRICS AND ANALYSIS9Introduction9Noise102.2.1Thermal noise112.2.2MOS devices dominant noise sources12Basic LNA Topologies132.3.1Common source Topology132.3.2Common Gate Topology142.3.3Cascode Topology162.3.4S-Parameters17	I I I
2	 LN2 2.1 2.2 2.3 2.4 	A PERFORMANCE METRICS AND ANALYSIS 9 Introduction 9 Noise 10 2.2.1 Thermal noise 11 2.2.2 MOS devices dominant noise sources 11 2.2.2 MOS devices dominant noise sources 12 Basic LNA Topologies 13 2.3.1 Common source Topology 14 2.3.2 Common Gate Topology 16 2.3.3 Cascode Topology 16 2.3.4 S-Parameters 16 Noise Figure 16)) 1 1
2	 LNA 2.1 2.2 2.3 2.4 2.5 	A PERFORMANCE METRICS AND ANALYSIS9Introduction10Noise112.2.1Thermal noise2.2.2MOS devices dominant noise sources13142.2.2MOS devices dominant noise sources14152.3.1Common source Topology2.3.2Common Gate Topology2.3.3Cascode Topology2.3.4S-ParametersNoise Figure19Sensitivity21	 b b c c

		2.6.1 Harmonic distortion	21
		2.6.2 1-dB compression point	22
		2.6.3 Input intercept Point	22
	2.7	Figure of Merit	23
	2.8	Statistical Analysis	23
	2.9	DoE Analysis	24
	2.10	Fabrication and measurement process	25
3	BRO	OADBAND LNA DESIGN USING SSL	27
	3.1	Introduction	27
	3.2	Prior Work	28
	3.3	Conceptualization and Implementation	29
		3.3.1 SSL Structural Analysis	29
		3.3.2 Analytical Modeling of SSL Inspired Parallel-Series Network	35
		3.3.3 Built-In Techniques Based CMOS LNA Design	41
	3.4	Results Discussion	44
	3.5	Summary	51
	-		
4	\mathbf{QU}	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS-	
4	QU. TEN	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS	55
4	QU. TEN 4.1	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 55
4	QU. TEN 4.1 4.2	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 55 56
4	QU. TEN 4.1 4.2 4.3	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 55 56 57
4	QU. TEN 4.1 4.2 4.3 4.4	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 55 56 57 58
4	QU. TEN 4.1 4.2 4.3 4.4 4.5	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 55 56 57 58 60
4	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 56 57 58 60 62
4	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6 4.7	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	555 556 577 588 600 622 63
4	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6 4.7	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	555 556 577 588 600 622 633 655
4	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6 4.7	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 55 56 57 58 60 62 63 65 65
4	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6 4.7	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	555 556 577 588 600 622 633 655 666
4	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 55 57 58 60 62 63 65 66 67 72
4 5	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8 LNA	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 56 57 58 60 62 63 65 66 67 72
4 5	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8 LNA IGA	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	55 55 57 58 60 62 63 65 66 67 72 73
4 5	QU. TEN 4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8 LNA IGA 5.1	ASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYS- MS Introduction	 55 56 57 58 60 62 63 65 66 67 72 73 73

	5.2	Prior Work
	5.3	Impedance Match Design Using FBSSIR
		5.3.1 Fundamentals of Stepped Impedance Resonator (SIR) $\ldots $ 76
		5.3.2 FBSSIR design and analysis
	5.4	Design and consideration of LNA integrated with FBSSIR 85
		5.4.1 DoE Analysis $\ldots \ldots $
		5.4.2 Statistical Analysis
	5.5	Results and Discussion
	5.6	Summary
6	\mathbf{SW}	ITCHLESS RECONFIGURABLE LNA BASED ON SSCL FOR
	K/l	KA BAND101
	6.1	Introduction
	6.2	Prior Works
	6.3	SSCL design and consideration
		6.3.1 Coupled line Analysis $\ldots \ldots 103$
		6.3.2 Comparison of basic coupled line structure with proposed SSCL 110
	6.4	Design and Analysis of CMOS Switchless LNA
		6.4.1 Design of Broadband drive stage
		6.4.2 Design of High band (Ka) and Low band (K) LNA stages 118
		6.4.3 Consolidation of all Stages
		6.4.4 Noise Figure Analysis $\dots \dots \dots$
		6.4.5 Statistical Analysis
		$6.4.6 \text{DoE Analysis} \dots \dots \dots \dots \dots \dots \dots \dots \dots $
	6.5	Performance Analysis
		6.5.1 Experimental Verification
	6.6	Summary
7	CO	NCLUSION AND FUTURE WORKS 135
	7.1	Conclusion
	7.2	Future Works
Α	PPE	NDICES 137
	I: N	OISE FIGURE ANALYSIS
	A-1	Noise Figure Derivation

Bibliography	140
Publications Based on the Thesis	155

List of Figures

1.1	Simple block diagram of typical RF frontend	3
1.2	RF design Trade-offs	4
1.3	Parameters impacting RF performance	5
2.1	MOS transistor Dominant noise sources representation	12
2.2	Common source Topology with no degeneration	15
2.3	Common source Topology with inductive source degeneration	16
2.4	Common gate Topology	17
2.5	Cascode Topology with degeneration $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	18
2.6	Port network for evaluating S-Parameters	18
2.7	IIP_3, P_{1dB} representation (log scale)	23
2.8	CMOS fabrication process flowchart	26
3.1	3D Geometric view of proposed SSL	30
3.2	SSL component with (a) with splicing of layers (b) side view	31
3.3	Surface Current density of SSL (a. M1 and M2 layers (b) M5 and M6 $$	
	layers (c) M3 and M4 layers (d) Complete layers	32
3.4	Impedance performance of SSL with its (a) characteristic impedance	
	and (b) smith chart \ldots	34
3.5	Return loss Versus Frequency	35
3.6	Phase versus Frequency	36
3.7	Variation of resonant frequencies with respect to airgap between metal	
	layers of SSL	36
3.8	The far-field pattern of the slotted metal layer (a) E-theta and E-phi	
	at 27.2GHz and (b) H-theta and H-phi at 27.2GHz	37
3.9	Far-field pattern of a slotted metal layer (a) E-theta and E-phi at	
	32.6GHz and (b) H-theta and H-phi at 32.6GHz	37

3.10	SSL based parallel-series network with its (a) schematic and (b) layout	
	design \ldots	38
3.11	Smith chart plot of SSL parallel-series network	39
3.12	Reflection of power versus frequency	39
3.13	Transmitted power versus frequency	40
3.14	Input impedance versus frequency	41
3.15	Proposed schematic of built-in techniques-based CMOS LNA $\ \ . \ . \ .$	42
3.16	Photographs of LNA with (a) 3D layout and (b) die microchip photo-	
	graph	44
3.17	LNA measurement with its (a) experimental set-up and (b) on-wafer	
	calibration standards \ldots	45
3.18	Simulated return loss Vs frequency of LNA with varying SSL length $~~$.	46
3.19	Simulated Forward gain Vs frequency LNA with varying SSL Length $~$.	47
3.20	S-parameter Vs frequency of LNA with proper matching $\ldots \ldots \ldots$	47
3.21	Simulated input impedance curve versus frequency (a) Rectangular Plot	
	(b) Smith chart Plot	48
3.22	Noise figure versus frequency	49
3.23	Stability factor K of LNA	50
3.24	Measured 1-dB compressed point of LNA	51
3.25	Yield estimation of LNA with (a) Iteration count vs return loss (b)	
	Iteration count vs forward gain and (c) Iteration count vs noise figure .	52
4.1	Simple Circulator representation	57
4.2	Schematic of Proposed QC-LNA	58
4.3	Simulated S-parameters variations of QC	60
4.4	Proposed Flow chart for optimization using QN algorithm of QC-LNA	64
4.5	Microchip die photo of fabricated QC-LNA	65
4.6	Yield estimation of QC-LNA (a) Gain (b) NF (c) Return Loss $\ . \ . \ .$	66
4.7	Design variable contribution to (a) Gain (b) NF (c) Return Loss	67
4.8	Simulated input impedance using smith chart plot	68
4.9	Stability factor K of QC-LNA	69
4.10	Noise figure of designed QC-LNA	69
4.11	S Parameters of designed QC-LNA	70
4.12	Input power (P_{in}) vs. Third order harmonics (IP_3) and Output Power	
	$(P_{out} \text{ fundamental}) \dots \dots$	72

5.1	Block diagram of a transportation management system using IRNSS	
	receiver	74
5.2	(a) Schematic diagram of SIR (b) The equivalent circuit diagram of the	
	SIR and (c) The equivalent diagram of Folded SIR with a gap	77
5.3	Proposed FBSSIR with is (a) Schematic view (b) Fabricated view	78
5.4	Coupling Scheme of the FBSSIR	79
5.5	Flowchart of the design procedure for FBSSIR	80
5.6	The surface current distribution at (a) 1.6GHz (b) 2GHz and (c) 2.58GHz $$	81
5.7	S-parameters analysis by varying the stub height (j)	82
5.8	S-parameters analysis by varying the folded T-stub length (L2)	82
5.9	S-parameters analysis by Varying the butterfly stub length (L1) \ldots	83
5.10	Smith chart plot of the FBSSIR	84
5.11	Simulated and Measured S-parameters of FBSSIR	84
5.12	Proposed FBSSIR based LNA (a) Schematic (b) 3D layout	87
5.13	Small signal equivalent circuit of the proposed FBSSIR based LNA $~$	88
5.14	Design variable contribution to (a) Gain (b) NF (c) Return Loss	90
5.15	Yield estimation of FBSSIR-LNA (a) Gain (b) NF (c) Return Loss $~$.	91
5.16	Fabricated photograph view with (a) LNA prototype (b) FBSSIR inte-	
	grated LNA prototype	92
5.17	Noise Figure of the designed FBSSIR-LNA	93
5.18	Gain of the designed FBSSIR-LNA	93
5.19	Simulated Comparison of Pin vs. IP3 and Pout (fundamental) of LNA	
	with and without FBSSIR	94
5.20	Simulated smith chart plot of FBSSIR integrated LNA (a) Input impedance	
	(b) Output impedance	95
5.21	Measured Stability factor K and Δ of FBSSIR-LNA	96
5.22	Simulated and measured S-parameters of FBSSIR-LNA	96
6.1	Conventional coupled line structure (A)	04
6.2	Even-Odd mode Excitation in CL	04
6.3	Coupled power of conventional coupled line A	06
6.4	Proposed Coupled line structure B	07
6.5	Coupled power of proposed coupled line B without capacitor C_{\circ} 1	~. 08
6.6	Coupled power of proposed coupled line B according to different cou-	
0.0	pling coefficients without capacitor C_{\bullet}	09
	ro	50

6.7	Phase difference of proposed coupled line B without capacitor C_8 109	
6.8	Phase difference of proposed coupled line B with capacitor C_8 110	
6.9	(a) 1-D Microstrip Coupled Line (MCL) (b) 1-D Suspended Substrate	
	Coupled Line (SSCL) (c) 3-D geometry of SSCL	
6.10	Schematic of (a) Inter-stage MCL/SMCL/SSCL (b) Output-stage SSCL 112	
6.11	Simulated S-parameters and phase of MCL/SMCL as interstage $\ . \ . \ . \ 112$	
6.12	Simulated S-parameters and phase of SSCL as interstage $\ . \ . \ . \ . \ . \ . \ . \ . \ . \ $	
6.13	Simulated S-parameters and phase of SSCL as output stage $\ . \ . \ . \ . \ . \ 113$	
6.14	Proposed reconfigurable LNA topology	
6.15	Schematic of the proposed reconfigurable LNA	
6.16	Simplified small-signal model of first CS stage for input impedance 116	
6.17	Simulated S-Parameters performance of the separate broadband drive	
	stage	
6.18	Simulated S-Parameters performance of the separate Low-band (K) stage. $\underline{118}$	
6.19	Simulated S-Parameters performance of the separate High-band (Ka)	
	stage	
6.20	Simulated Noise Figure performance of the broadband, high band, and	
	low band stages	
6.21	Yield estimation of LNA (a) Gain (b) NF. (c) Return Loss $\ . \ . \ . \ . \ . \ . \ . \ . \ . \ $	
6.22	Design parameter contribution at (a) Low-band at 20GHz (b) High-	
	band at 35GHz	
6.23	Microchip die photo of fabricated LNA \hdots	
6.24	Measurement setup of the LNA $\ldots \ldots \ldots$	
6.25	Simulated smith chart plot of proposed LNA Input Impedance \ldots 126	
6.26	Simulated smith chart plot of proposed LNA Output Impedance $~~$ 127	
6.27	Stability factor K of LNA	
6.28	Noise Figure of the designed dual-band LNA $\hfill \ldots \ldots \ldots \ldots \ldots \ldots \ldots 128$	
6.29	Simulated and measured S-parameters of the designed LNA in dual-	
	band mode	
6.30	Simulated group delay and phase response of the designed LNA in dual-	
	band mode	
6.31	Measured S-parameters and NF of the designed LNA in high-band mode 129	
6.32	Measured S-parameters and NF of the designed LNA in low-band mode 130	
6.33	Measured K-band Linearity performance	

6.34	Measured Ka-band Linearity performance	131
A.1	Noisy multi-port (a) its equivalent noiseless multi-port network (b) with	
	equivalent current and voltage noise sources at the input	137

List of Tables

2.1	Reported works on LNA	10
2.2	Basic LNA topologies comparison	13
3.1	Optimized values of designed LNA components	43
3.2	Comparison of the LNA performance with prior works found in the	
	literature	53
4.1	Optimized values of designed QC-LNA components	65
4.2	Design Variables assigned for DoE Analysis	68
4.3	Comparison of the LNA performance with prior works	71
5.1	Comparison of state of art with resonators	85
5.2	Optimized component values of the proposed LNA circuit	86
5.3	Results summary of FBSSIR-LNA	97
5.4	Comparison of the FBSSIR-LNA performance with state of art	99
6.1	Drain Voltages states of single-band transistors	20
6.2	Optimized values of designed Switchless-LNA components 12	23
6.3	Comparison of the proposed reconfigurable LNA performance with state $\$	

Abbreviations

SSL	Suspended Substrate Line
LNA	Low Noise Amplifier
THD	Total harmonic distortion
MOSFET	Metal oxide semiconductor field effect transistor
NMOS	N-channel metal oxide semiconductor
PMOS	P-channel metal oxide semiconductor
CMOS	Complementary metal oxide semiconductor
EM	Electro Magnetic
RF	Radio Frequency
NF	Noise Figure
QC	Quasi Circulator
SoC	System-on-chip
QN	Quasi Newton
SD	Source degeneration
FBSSIR	Folded Butterfly Stub Stepped Impedance Resonator
SIR	Stepped Impedance Resonator
SNR	Signal to noise ratio
FoM	Figure-of-merit
DoE	Design of experiment
DC	Direct current
SSCL	Suspended Substrate Coupled Line
PVT	Process, supply voltage and temperature
IIP	Input Intercept Point
OIP	output Intercept Point
RFIC	Radio Frequency Integrated Circuit
PCB	Printed Circuit Board
ADS	Advanced Design System
DUT	Device Under Test
MSL	Microstrip Line
PDK	Process Design Kit
PNA	Programmable Network Analyzer
IRNSS	Indian Regional Navigational Satellite System
BJT	Bipolar Junction transistor

xxii

•

Notations

- f Frequency in Hz
- ω Angular frequency in radians per second
- f_c Cut-off frequency in Hz
- g_m Transconductance
- γ Excess drain noise coefficient
- δ Excess gate noise coefficient
- Q Quality factor
- c Correlation coefficient
- L Channel length of the MOSFET
- W Channel width of the MOSFET
- V_{GS} Gate-source voltage of the MOSFET
- V_{DS} Drain-source voltage of the MOSFET
- k Boltzmann Constant
- T Absolute temperature
- μ_n Mobility of electrons
- C_{ox} Gate-oxide capacitance
- g_{d0} Drain conductance for zero drain-source voltage
- g_g Frequency dependent gate conductance
- Z_0 Characteristic Impedance
- ε_{eff} Effective dielectric constant
- λ Wavelength

xxiv

•

Chapter 1

INTRODUCTION

The phrase "communication" refers to the sharing of information between two individuals or groups of individuals using a medium for transmission. The communication process will succeed when the receiver provides positive feedback to the sender on the received signal. The term "Multimedia" refers to the simultaneous use of different modes of communication such as audio, text, video, images, graphics, animation, and so on at the same time.

Telecommunications' basic purpose was connectivity with mobility. As Technology evolved the services provided with it began to grow. Currently, services are not limited to voice and text messaging. A Long-term Evolution (LTE) known as 4G communication with a new and updated infrastructure was introduced in 2009, which provided very high data rates of more than 28Mbps. The world is mostly using this advanced version of 4G communications, which provides multi-services such as mobile banking, multimedia browsing such as video streaming, video calling, etc. The latest survey on networking shows that more than 90% of network data traffic is of multimedia content.

The purpose of the Wireless Broadband Multimedia Communication Systems (WBMCS) is to provide users with radio access to broadband multimedia services that are either supported by customer premises networks or directly provided by public fixed networks. WBMCS will provide a non-wired, mobile (or at least movable) extension to wired networks for data speeds greater than 300 Mbps, with applications in wireless Local Area Network (LAN) or mobile broadband systems.

WBMCS is being researched in the microwave (3-30GHz) and millimeter wave (30-300GHz) bands to meet the required bandwidth. The topic of WBMCS research has garnered considerable attention as a result of the growing importance of multimedia and computer applications in communications. The latest development is the next generation communication system known as 5G developed from the LTE advanced, which provides multi services than 4G with a high data rate of 1 Gbps and accuracy.

1.1 RF Frontend

The RF front end is the heart of any wireless communication system. The term RF front end refers to modules located near the antenna. The block diagram of a typical RF front end is shown in Figure 1.1. It consists of a filter, amplifier, and converter. The amplifier used in the transmitter is called a Power amplifier because it focuses on multiplying the signal's power density before transmission. In contrast, the amplifier in the receiver is called Low Noise Amplifier (LNA) because it mostly determines the noise performance of the entire RF receiver. Three critical signal processes are performed by an RF front end, both in the transmitter and receiver signal paths(Baltus 2004):

- Frequency translation is used to convert a stream of signals to a frequency suitable for transmission or processing.
- Amplification of a signal to obtain an amplitude suitable for further transmission or processing.
- Signal filtering, to exclude undesired signal components created by other transceivers or the transceiver's signal-processing circuits.

The frequency conversion happens with the help of mixers and local oscillator signals. It can also be performed in two or more conversion stages, which reduces the performance requirements for the individual filter, gain, and mixer blocks. A straightforward antenna interface is seen here, with distinct, single-antenna transmits and receives antennas. In fact, it is generally beneficial to share the transmitter's antenna with the receiver to save money and space. Between the circuits and the antenna, this requires switches or duplex filters. Additionally, many systems today employ multi-band radios, necessitating the use of several antennas or numerous antenna feeds, which necessitates the addition of additional switches or, in some cases, variable capacitors.

Frequency conversion oscillators are frequently tunable to enable channel selection at the radio's front end, hence decreasing the dynamic range needs of the filters, IF (Intermediate Frequency) amplifiers, and data converters. These tunable oscillators are often controlled by a synthesizer loop, which significantly complicates the front end in terms of transistor count. Even yet, an RF front end is composed of a small number of active devices, frequently in the hundreds or thousands. The silicon area of the front-end lowers as IC technology advances, resulting in device shrinkage in physical dimensions. Current RF front-end integrated circuits are only a few square micrometers in size. Additionally, with enhanced transceiver architectures such as single-conversion receivers and greater radio digitization, the needed functionality, and hence the number of active devices lowers.

After the RF front ends, the signal will be sent to the RF back end, which refers to the module far away from the antenna or also known as the baseband processing module. The role of the receiver is more crucial than the sender in the communication process. The LNA is the most crucial part of the receiver module. This thesis focuses primarily on the Low Noise Amplifier design.



Figure 1.1: Simple block diagram of typical RF frontend

1.2 Issues in RF Frontend

Wireless Broadband Multimedia Communication Systems (WBMCSs), which support data rates greater than 20 Mb/s, are rapidly emerging (da Costa and Yang 2020). The RF design is quite challenging because a good understanding of multidiscipline knowledge is needed. Even if we focus on the RFIC design, mainly the receiver or transmitter, they also possess many trade-offs among noise, power, linearity, gain, frequency, and supply voltage. This is depicted as an RF design hexagon in Figure 1.2 (Razavi 2013). These trade-offs in the RFIC design create numerous architecture designs based on the requirement and applications used.



Figure 1.2: RF design Trade-offs

The RFIC performance depends on many factors and is mainly classified into two variables: device-level variables and board-level variables. The main parameters affecting the RFIC performance are depicted in Figure 1.3 (Das 2013). It is noted that the devices level variables such as device geometry. Process technology and parasitics depend purely on the silicon technologies such as CMOS, BiCMOS, and so on. Each technology has distinct advantages and disadvantages and is chosen only for the purpose of meeting the requirements and applications. This gives the researcher the opportunity to work with various device technologies in the RFIC design. The circuit and board-level variables also allow the researcher to illustrate numerous designs based on requirements and applications. In short, the research gap in RFIC design is depicted as:

- Demands for enhanced performance, cost savings, and functionality continue to create new obstacles.
- These trade-offs and advancements in technology make the RFIC design more challenging and never-ending scope in the future.

• As communication evolves rapidly, such as 5G, the need to invent new RFIC chips is mandatory. This gives a significant scope in working in the field of RFIC design.



Figure 1.3: Parameters impacting RF performance

1.3 Motivation

The Low noise amplifier (LNA) is the most crucial signal processing block in any RF receiver as it decides almost all the important parameters of the receiver. The RF design tradeoffs, as in Figure 1.2, allows the researcher to propose the new design trade-offs in the LNA to achieve high performance at reduced cost, power, and size. Moreover, the combination of device-level variables and circuit-level variables allows one to propose and test a new LNA design topology to enhance the receiver performance better.

The incorporation of the microwave techniques and components into the design of the LNA pave the way for the new dimensional performance analysis. These microwave techniques and components help in controlling the signal in the circuit, such as controlling the direction of the wave, splitting the wave, phase shifting of the wave, and so on. One such component is the suspended substrate line which allows the designer to optimize the performance of the LNA for better trade-offs. This urges the RF designer to work on developing new methodologies by using microwave techniques for the LNA design.

1.4 Objectives

- To design a wideband LNA by using a suspended substrate line in order to achieve high gain and low noise figure.
- To design and analyze Quasi-Circulator integrated LNA using Quasi Newton Algorithm optimization to achieve ultra-low-noise figure and stable flat gain over a broadband frequency range of operation.
- To design and analyze stepped impedance resonator integrated LNA to get yield estimation and robustness of the design.
- To design and analyze a switchless reconfigurable ultra-wideband LNA to achieve flat gain, ultra-low noise figure, good linearity, low power consumption, and compact size.
- To carry out all CMOS designs from layout to commercial 65nm CMOS chip fabrication.

1.5 Organization and contribution of thesis

The thesis's aim is to build novel LNA designs with built-in techniques for broadband applications. Based on the technological background and motivation, the primary interest is in building integrated circuits for LNA with the best power, noise, and impedance matching. Additionally, this LNA should have a high gain and linearity to enhance receiver performance.

Chapter 2 details the significance and definition of LNA performance metrics that are used to evaluate RF design, such as noise figure, gain, 1-dB compression point, Figure of merit, input intercept point, S-Parameters, and dynamic range. Additionally, a brief description and comparison of the conventional LNA topologies are provided.

Chapter 3 details the proposed techniques of the Suspended Substrate line (SSL) based LNA for a 5G communication system in the frequency range of 28-32GHz.
Primarily, structural analysis of the SSL is presented with the simulation results. Second, the analytical modeling of SSL inspired parallel series network with the layout and simulation results are provided. Then a detailed discussion on the proposed CMOS LNA based on the built-in techniques is given. Finally, the measurement and simulation results and their discussion are provided to compare state-of-the-art.

Chapter 4 discusses the Quasi circulator (QC) integrated LNA for X-band RF front-end systems working in the frequency range of 8-12GHz. The design of the Quasi circulator based on the CMOS process and its simulation results are given primarily. A detailed discussion of the CMOS LNA design topology is presented secondly. The optimization algorithm, namely the Quasi-Newton algorithm, is given afterward. Finally, a detailed discussion is provided for the simulation and measurement results with the state-of-the-art comparison table.

Chapter 5 details the wide bandwidth Folded Butterfly Stub Stepped Impedance Resonator (FBSSIR) integrated LNA for satellite navigation systems operating in the frequency range of 1.6-2.5GHz. Firstly, the proposed FBSSIR design fundamentals and analysis with the PCB layout and results are provided. Secondly, the proposed LNA design based on the integration with FBSSIR is provided in a detailed manner. Finally, a detailed discussion of the measurement results is provided with the stateof-the-art comparison table.

Chapter 6 discusses the switchless reconfigurable CMOS LNA based on the Suspended Substrate Coupled line (SSCL) operating in the full K/Ka-band (18-40GHz) is presented. Firstly, the coupled lines are discussed in-depth with analysis and simulation results to show the potential of the proposed SSCL with the conventional coupled lines. Secondly, the design topology of CMOS LNA is discussed in detail with simulation results. Finally, a detailed discussion of the measurement results is provided with the state-of-the-art comparison table.

The conclusion of this thesis is given in Chapter 7, highlighting the contribution of the work and throwing hints at possible future works. All the LNA designs proposed are fabricated and tested with measurement results. They are also compared with the respective state of art LNA in each chapter.

Chapter 2

LNA PERFORMANCE METRICS AND ANALYSIS

2.1 Introduction

When designing a complete wireless radio receiver, it is critical to evaluate the low noise amplifier and other blocks to have a complete understanding of their performance. The performance of a receiver design is determined by a number of factors, including the receiver's sensitivity, selectivity, and susceptibility to reception mistakes. The RF design engineer's primary objective is to maximize the front-end performance, with a particular emphasis on the first amplifying device. To reliably retrieve the contained information, high-end receivers require a good LNA to discriminate the desired signal from the noise in the environment. Certain properties of the LNA are controllable by the designer and have a direct effect on the receiver's sensitivity: noise figure, gain, bandwidth, linearity, and dynamic range. Controlling these properties, on the other hand, demands a grasp of the active device, impedance matching, fabrication, and assembly details in order to design an amplifier with the fewest possible trade-offs.

The set of variables affecting LNA performance at the device and board design levels is depicted in Figure 1.3. It is the designer's responsibility to reduce the impact of external variables while identifying the most appropriate trade-offs between conflicting qualities in order to maximize receiver sensitivity, selectivity, and information integrity. The trade-offs at the device level are determined by the process technology, transistor shape, and package parasitics. All of these parameters have a substantial effect on noise performance and should be taken into account when designing the

Reported Works	Technology	Frequency	Gain	\mathbf{NF}	Power	Area
		(GHz)	(dB)	(dB)	(mW)	(mm^2)
Qin and Xue $2017a$	$65 \mathrm{nm}$	16-30	10.2	5.7	19	0.37
	CMOS					
Qin and Xue $2017b$	$65 \mathrm{nm}$	8-29	10.7	5.6	12.1	0.3
	CMOS					
Davulcu et al. 2018	130nm SiGe	8-12	20	1.4	100	0.72
	CMOS					
Çalışkan <i>et al.</i> 2019	130nm SiGe	6-12	10	1	20	0.4
	BiCMOS					
Wang and Zhang 2020	130nm SiGe	22 - 47	22	4.3	10	0.13
	BiCMOS					
Yu et al. 2021	$65 \mathrm{nm}$	22-40	28	3.2	32	0.4
	CMOS					

LNA. In fact, implementing a high-performance LNA is difficult due to parameter trade-offs. The recent reported works in the LNA are provided in the Table 2.1 for getting insights into the performance reported so far.

Table 2.1: Reported works on LNA

2.2 Noise

In communication systems, noise is defined as an unwanted signal because it reduces the overall system's sensitivity. There are numerous sources of noise, each with its own unique method of noise generation. Shot noise, flicker noise, and thermal noise are the three primary causes of noise in integrated circuits. Shot noise is primarily created by electric charges hopping across a potential barrier and is characteristic of nonlinear devices such as diodes and transistors. The only source of shot noise in MOS devices is the DC gate leakage current, and thus it is not regarded as a significant issue (Lee 2004). This is in contrast to bipolar transistors, which can drastically reduce the receiver's overall performance due to base and collector shot noise.

Flicker noise (alternatively called pink noise) originates as a result of charges being trapped in the flaws and impurities in the channel area of MOS devices. Larger MOS devices, on average, exhibit less flicker noise. (Lee 2004) provides the spectral density of this noise.

$$\overline{i_{fn}^2} = \frac{K \cdot g_m^2}{fWLC_{ox}^2} \tag{2.1}$$

Here, K denotes a device constant, g_m denotes the MOS device's transconductance, f denotes the operating frequency, C_{ox} denotes the gate-oxide capacitance per unit area, and W and L denote the MOS device's width and length, respectively.

2.2.1 Thermal noise

Thermal noise is generated when carriers in a conductor are agitated, and its spectral density is defined by the following number, referred to as available noise power (Lee 2004).

$$P_{NA} = kT\Delta f \tag{2.2}$$

where k denotes the Boltzman constant $(1.38 \times 10^{-23} J/K)$, T is the absolute temperature in Kelvins, and Δf denotes the noise bandwidth in Hz. P_{NA} is -174dBm for a 1 Hz noise bandwidth at room temperature (290K) and is frequently referred to as the system's noise floor. The noise floor is a critical parameter in determining the receiver's sensitivity (Lee 2004).

2.2.2 MOS devices dominant noise sources

Figure 2.1 (Molavi 2005) illustrates the various sources of noise in a MOS device. In the saturation area, the MOS device behaves like a trans-conductor; in the triode region, it behaves like a resistor. As a result, thermal noise associated with carriers in the channel will be identical to the noise associated with carriers in a conductor. (Van der Ziel 1986) deduced the expression for the drain current noise (also known as channel thermal noise) in MOS devices, which is given as

$$\overline{i_{nd}^2} = 4kT\gamma g_{do}\Delta f \tag{2.3}$$

where g_{d0} is the drain conductance at zero drain-source voltage and γ is a technologydependent parameter with a value of around 2/3 for long-channel devices at saturation (it is bigger in short-channel devices and has a value of 3 for 180 nm technology) (Molavi 2005). A comprehensive investigation of the noise characteristics of a MOS device indicates that channel thermal noise does not capture all of the noise produced by the device (Shahani *et al.* 1997). To simulate the additional noise, a frequencydependent gate conductance can be introduced:

$$g_g = \frac{\omega^2 C_{g_s}^2}{5g_{d0}} \tag{2.4}$$

And an equivalent gate current noise is expressed as:

$$\overline{i_{ng}^2} = 4kT\delta g_g \Delta f \tag{2.5}$$

where δ denotes the gate noise coefficient, which is a technology-dependent metric. Its value is 4/3 in the case of long channel and 6 in the case of 180 nm technology (Van der Ziel 1986). The correlation coefficient between gate current noise and channel thermal noise is provided as:

$$c = \frac{\overline{i_{ng}i_{nd}^*}}{\sqrt{\overline{i_{ng}^2}} \cdot \sqrt{\overline{i_{nd}^2}}}$$
(2.6)

where c is a complex number with a theoretical value of roughly 0.395 j for long channel devices as defined in Van der Ziel Aldert (1986) and -0.55 j for 180 nm technology. Another type of noise in MOS devices that may contribute to the total noise of LNA is the distributed resistance of the poly-silicon gate (Lee 2004). The value of this resistance is as follows:

$$R_g = \frac{R_{\Box}W}{3n^2L} \tag{2.7}$$

where R_{\Box} is the gate terminal's sheet resistivity, n is the number of gate fingers, and W and L are the width and length of the MOS device, respectively.



Figure 2.1: MOS transistor Dominant noise sources representation

2.3 Basic LNA Topologies

For comparison with the most relevant parameters, familiar LNA topologies have been considered. The three familiar topologies of the LNA are Common Source (CS), Common Gate (CG), and Cascode. Based on the reported works on these topologies it is noted that to construct narrow-band systems, the common source topology is more appealing. Stability and linearity are enhanced by the cascode connection of the same topology. On the other hand, the common gate topology's $1/g_m$ input impedance makes it suitable for wide-band applications. Table 2.2 compares the three basic topologies(Razavi 2013). The basic topologies can be modified to enhance their

PARAMETERS	Common Source	Common Gate	Cascode
Gain	moderate	low	high
Noise Figure	best	better	good
Sensitivity to PVT	more	less	less
Reverse isolation	low	good	best
Bandwidth	low	very wide	wide

 Table 2.2:
 Basic LNA topologies comparison

performance accordingly.

2.3.1 Common source Topology

The Figure 2.2 represents the basic common source topology. We can modify the common source topology by having inductive load, resistive feedback, capacitive feedback, inductive degeneration, and so on. The resistive termination and voltage degradation are the bottlenecks in this topology and is quite useful in broadband matching than narrow-band. The noise factor at low frequency for this topology can be expressed as (Razavi 2013).

$$Nf = 2 + 4\frac{\gamma}{\alpha}\frac{1}{g_m R} \tag{2.8}$$

where α is $\frac{g_m}{g_{d0}}$. The noise factor exceeds 3dB even for low value of γ . This topology has an extremely high noise factor, which is incompatible with typical front-end architectures. Additionally, it is susceptible to transistor parasitics. The solution to this lies in degeneration of source through inductor known as inductive source degeneration common source topology as shown in Figure 2.3. This is traditionally used for narrow band application but some works shown that by proper optimizing and matching we can make this for broadband application too. In this topology, simultaneous high gain and low noise can be achieved. Additionally, it provides excellent impedance matching without the use of a real resistor. From (Perrott 2005) and (Reddy 2019), the total drain noise current is depicted as

$$\overline{\frac{i_{ndg}^2}{\Delta f}} = \overline{\frac{i_{nd}^2}{\Delta f}} \left(|\eta|^2 + 2\operatorname{Re}\left\{ cX_d + Z_{gsw} \right\} + X_d^2 \left| Z_{gsw} \right|^2 \right)$$
(2.9)

where

$$\frac{\overline{i_{nd}^2}}{\Delta f} = 4kT\gamma g_{d0}, \ \eta = 1 - \left(\frac{g_m Z_{deg}}{Z_{deg} + Z_g}\right), \ Z_g = R_S + j\omega L_g, \\
Z_{deg} = j\omega L_{deg}, \\
X_d = \frac{g_m}{g_{d0}}\sqrt{\frac{\delta}{5\gamma}} \ and \ Z_{gsw} = \frac{1}{j\omega C_{gs}} \|\frac{Z_{deg} + Z_g}{1 + g_m Z_{deg}}.$$

The auxiliary inductor (L_g) offers freedom to set resonance frequency (ω_0) and Z_{in} separately. For technology below 180 nm it equation 2.9 can be modified as

$$\frac{\overline{i_{ndg}^2}}{\Delta f} = \frac{\overline{i_{nd}^2}}{\Delta f} \left(|\eta|^2 + 2 \operatorname{Re} \left\{ -j|c|X_d^* Z_{gsw} \right\} + X_d^2 |Z_{gsw}|^2 \right)$$
(2.10)

where $\eta = 1/2, Z_{gsw} = \frac{1}{2} (2Q_{in} - j)$ and Q_{in} denotes the quality factor of the input RLC network and is expressed as

$$Q_{in} = \frac{1}{2\omega_0 R_S C_{gs}} = \frac{\omega_0 \left(L_g + L_{deg}\right)}{2R_S}$$
(2.11)

Equation 2.10 is simplified as

$$\frac{\overline{i_{ndg}^2}}{\Delta f} = \frac{\overline{i_{nd}^2}}{\Delta f} \frac{1}{4} \left(1 - 2|c|X_d + (4Q^2 + 1)X_d^2 \right)$$
(2.12)

The noise generated by the input source is specified as

$$\overline{i_{\text{nout}}^2} = (g_m Q_{\text{in}})^2 \overline{v_{nR_S}^2}$$
(2.13)

The noise factor can also be defined as

$$Nf = \frac{\text{Total output noise power}}{\text{Output noise due to input source}}$$
(2.14)

Noise factor of inductive degeneration common source stage is written as

$$Nf = \frac{(g_m Q_{in})^2 \overline{v_{nR_s}^2} + \overline{i_{ndg}^2}/\Delta f}{(g_m Q_{in})^2 \overline{v_{nR_s}^2}} = 1 + \left(\frac{\omega_0}{\omega_t}\right) \gamma \underbrace{\left(\frac{g_{d0}}{g_m}\right) \left(\frac{1}{2Q_{in}}\right) \left(1 - 2|c|X_d + \left(4Q_{in}^2 + 1\right)X_d^2\right)}_{\text{Naise condition factors}}$$
(2.15)

Noise scaling factor



Figure 2.2: Common source Topology with no degeneration

2.3.2 Common Gate Topology

The common gate stage's low input impedance makes it an appealing choice for LNA design. The basic CG topology is shown in Figure 2.4. Without the use of a resistor, this topology provides impedance matching. While CG architecture has a lower noise factor than resistive terminated CS, it worsens at higher frequencies. The noise factor



Figure 2.3: Common source Topology with inductive source degeneration

can be expressed as (Razavi 2013).

$$Nf = 1 + \frac{\gamma}{\alpha} \frac{1}{g_m R_S}$$

$$\Rightarrow Nf = 1 + \frac{\gamma}{\alpha} \left(\because \frac{1}{g_m} = R_S \right)$$
(2.16)

This topology can also be modified to enhance the noise factor performance in the desired frequency range.

2.3.3 Cascode Topology

The cascode topology is a combination of CS and CG topologies and is shown in Figure 2.5. To minimize the noise contribution of M_{CG} , the inductor L_{deg} is used as source degeneration. This also allows the input to get isolated from the load, such as inductive load or resistive load. The real part and imaginary part impedance matching at the input is done by the degeneration inductor. The minimum noise factor of the cascode topology is expressed as (Nguyen *et al.* 2004).

$$Nf = 1 + \frac{2}{\sqrt{5}} \frac{\omega}{\omega_T} \sqrt{\gamma \delta \left(1 - |c|^2\right)}$$
(2.17)

where

$$\omega_T = \frac{g_m}{C_{gs}}$$



Figure 2.4: Common gate Topology

c is channel thermal noise correlation coefficient, δ is the gate noise coefficient and γ is a technology dependent parameter.

2.3.4 S-Parameters

Generally, we use Z, Y, and h-parameters while evaluating two-port networks at low frequency. To determine these parameters, open and short circuit tests must be performed, but at higher frequencies, this is extremely difficult due to the presence of parasitic capacitance and inductance. Due to the impossibility of conducting short and open circuit testing and the potential of causing damage to the circuit during these tests, (Collin 2001) recommends using an alternative method to characterize the network at high frequencies. One typical technique is the use of S-parameters (S stands for scattering), which define the four variables as incident (reflected) input (output) voltage (or power) waves. S-parameters are defined in such a way that they take advantage of the fact that a transmission line terminated at its characteristic impedance does not reflect any power at its termination (Lee 2004). Consider the



Figure 2.5: Cascode Topology with degeneration



Figure 2.6: Port network for evaluating S-Parameters

block diagram of a two-port network illustrated in Figure 2.6, where Z_o denotes the impedance of the source and load terminations and E_{i1} and E_{r1} denote the magnitudes of the incident and reflected voltage waves, respectively. The coefficients of the S-parameters are as follows:

$$b_1 = S_{11}a_1 + S_{12}a_2$$

$$b_2 = S_{21}a_1 + S_{22}a_2$$
(2.18)

where

$$a_1 = \frac{E_{i1}}{\sqrt{Z_0}}, \ a_2 = \frac{E_{i2}}{\sqrt{Z_0}}, \ b_1 = \frac{E_{r1}}{\sqrt{Z_0}} \ and \ b_2 = \frac{E_{r2}}{\sqrt{Z_0}}$$

The square of the magnitudes of a_i and b_i are equal to the incident and reflected power at both ports after normalization to the square root of Z_o . Now, if we end the second port with Z_o , which equalizes a_2 , and connect port one to a power source, we have the following relationships:

$$S_{11} = \left. \frac{b_1}{a_1} \right|_{a_2=0} = \frac{E_{r1}}{E_{i1}} \tag{2.19}$$

$$S_{21} = \left. \frac{b_2}{a_1} \right|_{a_2=0} = \frac{E_{r2}}{E_{i1}} \tag{2.20}$$

where S_{11} is referred to as the input reflection coefficient and is a practical measure of the LNA's input port impedance matching, and S_{21} is the amplifier's forward gain. On the other hand, if port one is connected to Z_o and power is supplied from port 2, the following relationships exist:

$$S_{12} = \frac{b_1}{a_2}\Big|_{a_1=0} = \frac{E_{r1}}{E_{i2}}$$

$$S_{22} = \frac{b_2}{a_2}\Big|_{a_1=0} = \frac{E_{r2}}{E_{i2}}$$
(2.21)

where S_{12} denotes the reverse transmission or gain, and S_{22} denotes the network's output reflection coefficient. Using these definitions, we can predict that a good amplifier should have a large S_{21} for maximum gain, low S_{11} and S_{22} for optimal input and output matching, and a very low S_{12} for stability and reverse isolation. S_{21} is > 10 dB, S_{12} is < -40 dB and the input and output reflection coefficients are less than -10 dB; however, these values may vary depending on the application.

2.4 Noise Figure

The noise figure (NF) is a metric that indicates how the signal's signal-to-noise ratio (SNR) degrades as it travels through the receiver front-end. 'Harold Friis' defined the noise factor (Nf) mathematically as the ratio of the system's input SNR to its output

SNR. Nf is denoted by (Razavi 2013).

$$Nf = \frac{SNR_{i/p}}{SNR_{o/p}} \tag{2.22}$$

Additionally, the noise factor can be stated as

$$Nf = \frac{P_{sig}/P_{nR_S}}{SNR_{o/p}} \tag{2.23}$$

Where P_{sig} is the power of the input signal and P_{nR_s} is the power of the source resistance noise per unit bandwidth. As a result:

$$P_{sig} = P_{nR_S} \times (Nf) \times SNR_{o/p} \tag{2.24}$$

Because the power of the input signal is dispersed over the channel bandwidth, the total mean square power can be expressed as

$$P_{\text{sig,tot}} = P_{nR_S} \times (Nf) \times SNR_{o/p} \times B \tag{2.25}$$

where B is the channel bandwidth. The noise performance of a system determines the dynamic range's lower limit. The Nf value is typically expressed in decibels (dB) and is referred to as the noise figure (NF).

$$NF = 10\log_{10}(Nf) \tag{2.26}$$

Now the ultimate equation for NF from 2.25 is obtained as

$$NF = P_{sig,\min}(dB) - (-174(dBm/Hz)) - SNR_{o/p,\min}(dB) - 10\log_{10}B \qquad (2.27)$$

NF can be specified for both individual blocks and the full receiver. NF_{LNA} , for example, determines the inherent noise of LNA, which is amplified with the signal. Friis deduced the noise factor for a cascaded system of m stages in 1947, and it is denoted by

$$Nf_{\text{tot}} = Nf_1 + \frac{Nf_2 - 1}{A_{p1}} + \frac{Nf_3 - 1}{A_{p1}A_{p2}} + \dots + \frac{Nf_m - 1}{A_{p1}A_{p2}\dots A_{p(m-1)}}$$
(2.28)

where Nf_i and A_{pi} are the noise factor and power gain of the i^{th} stage, respectively. As seen in equation 2.28, the noise contribution of each step reduces as the gain of the preceding stage grows. Thus, the gain and noise factor of the first stage determines the overall noise performance of the receiver. In practice, because the LNA is the first active block in the receiver chain, it is responsible for high gain, low noise, and the ability to handle significant unwanted signals.

2.5 Sensitivity

It is the signal level at which a receiver/LNA can identify an acceptable signal with acceptable quality. Acceptable quality is defined as a sufficient SNR. Generally, it should be as low as feasible for a strong receiver. It is mathematically written as (Razavi 2013):

$$P_{\rm sen}(dBm) = -174(dBm/Hz) + NF(dB) + SNR_{o/p,\min}(dB) + 10\log_{10}B \quad (2.29)$$

where B denotes the RF system's bandwidth in Hz and $SNR_{o/p,\min}$ denotes the system's/minimum amplifier's SNR in dB at the output.

2.6 Distortion and intermodulation

While linear models can approximate the small-signal analysis of RF circuits, realworld circuits display some degree of nonlinearity, resulting in harmonic distortion (Razavi 2013).

2.6.1 Harmonic distortion

When a sinusoidal signal (i.e., $A \cos \omega t$) is applied to a nonlinear memoryless system, the output frequency is not only fundamental but also integer multiples of the input frequency (also called as harmonics). The general mathematical expression of a nonlinear system is as follows:

$$y(t) = \alpha_1 x(t) + \alpha_2 x^2(t) + \alpha_3 x^3(t) + \dots$$
(2.30)

where x(t) and y(t) are the non-linear system's input and output, respectively. The gain of nonlinear systems now decreases as a function of $(\alpha_1 A + \frac{3}{4}\alpha_3 A^3)$, where A is the magnitude of the input signal. Gain compression is the process by which gain deviates from its ideal characteristics. The 1-dB compression point and third order intercept point (IP_3) are the most often used non-linearity indicators.

2.6.2 1-dB compression point

The 1-dB compression point is defined as the point at which the actual power gain deviates by 1 dB from its ideal value. The peak voltage of 1-dB compression is determined mathematically by

$$A_{in,1dB} = \sqrt{0.145 \left| \frac{\alpha_1}{\alpha_3} \right|} \tag{2.31}$$

2.6.3 Input intercept Point

Above this point, the circuit becomes highly nonlinear due to the presence of intermodulation components. It should be as high as possible for a decent receiver/LNA. If a weak signal is accompanied by two strong interferers that exhibit third-order non-linearity, one of the intermodulation (IM) products will fall within the band of interest, thereby contaminating the desired component. The term 'intercept point' (IP) was used to describe signal distortion caused by third-order intermodulation between two neighboring interferers. It is determined via a two-tone test in which two signals of identical amplitude are used. The level of the input signal at which the third-order IM product has the same power as the fundamental is referred to as the two-tone input-referred third-order intercept point (IIP_3) , while the corresponding output level is referred to as the output third-order intercept point (OIP_3) . The dynamic range of a receiver is defined as the ratio of the highest input level that it can tolerate to the lowest input level that it can detect. Generally, it should be as high as feasible for a proficient receiver. The maximum level of the input signal determines the amount of distortion produced by the circuit. The circuit's noise contribution determines the lower limit on the input (P_{MDS}) . The Figure 2.7 illustrates the calculation of IIP_3 , P_{1dB} and dynamic range.



Figure 2.7: IIP_3, P_{1dB} representation (log scale)

2.7 Figure of Merit

The Figure of Merit (FoM) is a quantitative measure of equipment, system, or method's performance in comparison to its alternatives. In engineering, figures of merit are frequently developed for certain materials or devices to ascertain their relative value for a given application. Likewise, the figure of merit for LNA performance analysis is given by relating all the LNA metrics discussed above, such as gain, NF, power consumption, area, and IIP_3 . (Wang and Zhang 2020) define the FoM in the following equations 2.32, 2.33.

$$FoM = \left(\frac{Gain \,[abs] \times B.W \,[GHz]}{P_{dc} \,(mW) \times (NF \,[abs] - 1)}\right)$$
(2.32)

$$FoM_{IP3} = \left(\frac{Gain[abs] \times B \cdot W[GHz] \times IIP3(mW)}{P_{dc}(mW) \times (NF[abs] - 1)}\right)$$
(2.33)

2.8 Statistical Analysis

Given the Process Voltage and Temperature (PVT) volatility in device parameters, yield analysis is critical for calculating the real yield and confidence levels. The area under a normal Gaussian curve over a specified number of standard deviations is the confidence level. Monte Carlo yield analysis techniques have long been widely utilized and accepted as a method for yield estimation. The procedure is as simple as conducting a series of trials. Each trial is generated by randomly generating values for yield variables according to statistical distribution parameters, simulating the output, and comparing it to stated performance specifications. The strength of the Monte Carlo method is that its accuracy is independent of the number of statistical variables used and makes no simplifying assumptions regarding the probability distribution of component parameter values or performance responses. The disadvantage of this method is that each trial requires a full network simulation and that a large number of trials is necessary to acquire a high degree of confidence and an accurate estimate of yield. Fortunately, the simulator takes advantage of cutting-edge approaches to dramatically improve the Monte Carlo method's efficiency while maintaining its universality. Statistical analysis or yield analysis estimates the yield by performing a series of trials. Each trial results from randomly generating yield variable values according to statistical distribution specifications, performing a simulation, and evaluating the result against the stated performance specifications. We have performed the Statistical analysis using Keysight Advanced Design System (ADS). The confidence level is the area under a normal Gaussian curve over a given number of standard deviations (σ). $(\sigma = 1)$ gives 68.3%, $(\sigma = 2)$ gives 95.4% and $(\sigma = 3)$ gives 99.7%, of confidence levels respectively. The equation relating number of trials, confidence level, percent error is given by (Miller 2008) :

$$N = \left(\frac{C_{\sigma}}{\varepsilon}\right)^2 \cdot Y(1 - Y) \tag{2.34}$$

Here, N denotes sample size, C_{σ} is the confidence expressed as a number of standard deviations, ε is the percent error, and Y denotes the yield.

2.9 DoE Analysis

Design of Experiments (DOE) is a data-driven technique for a resilient design that enables you to understand how each design variable contributes to attaining the preceding aim. This method is very helpful for the circuit and design designers to improve the yield (Miller 2008). A typical DoE includes three primary steps:

- Plan the experiment
 - Assess the experimental resource budget

- Identify the input and response variables
- Assign levels (values) to input variables
- Perform the experiment and collect response data
- Analyze the data using statistical methods

This methodology can be applied sequentially to improve the statistical performance of a given circuit or system. Due to the inherent trade-off between predictive accuracy and the number of input variables, a screening experiment is used to discover variables that significantly contribute to performance variation. Following that, a refinement experiment might be performed to focus attention on the desired statistical response. In general, DOE procedures are designed to account for the types of errors that occur in any experiment. However, because circuit and system simulators produce similar findings for any analysis with equal input values, the complexity associated with designing, running, and analyzing tests is decreased. Because the experiment is being conducted on a computer, a more full characterization of the input/output interactions may be accomplished. Finally, because the computer conducts the onerous bookkeeping responsibilities during the trial, the likelihood of human error is significantly reduced.

DOE is mostly used to characterize an unknown process. In-circuit or system simulation, the unknown process is used to forecast the behavior of the under-test design (DUT). A straightforward strategy for characterizing a device under test (DUT) is to perturb each input variable (factor) in turn and record the ensuing output response. This strategy, however, fails when the response to a change in one element is dependent on the value of another factor. This computer simulation is done through the Advanced Design System RF platform software by Keysight.

2.10 Fabrication and measurement process

The Fabrication of RF circuits is mainly classified into three types namely CMOS technology, Monolithic Microwave Integrated Circuits (MMIC) and Microwave Integrated Circuit (MIC) or Printed Circuit Board (PCB) fabrication. In that PCB is a board level design using the RF components, while the other two CMOS and MMIC is chip level fabrication. The general CMOS fabrication process is given in the figure 2.8. The Measurement process is usally done through the Vector Network analyzer for



Figure 2.8: CMOS fabrication process flowchart

the calculation of the S-parameter, impedance calculation and Linearity. Nowadays other parameters such as Noise figure can also be calculated using a advanced vector network analyzer such as PNA-X (N5245B) without the need of the other additional equipment. Micro probing station is also needed to probe the die level chip to the PNA.

Chapter 3

BROADBAND LNA DESIGN USING SSL

3.1 Introduction

As 5G wireless networks evolve, the performance of a radio's front end becomes an increasingly crucial component of the RF receiver signal path, particularly in relation to the low-noise amplifier (LNA). With the advent of new process technologies for LNAs such as silicon germanium (SiGe), gallium arsenide (GaAs), and silicon on insulator (SOI), designers must re-evaluate the performance trade-offs associated with LNA parameters such as noise, gain, sensitivity, bandwidth, and power in order to utilize them effectively (Schweber 2018).

The front end's relevance cannot be emphasized because it significantly dictates the system's ultimate performance in weak signal circumstances and achievable bit error rate. If the LNA performance is inadequate, the remaining circuitry and receive channel management design attempts to achieve 5G performance will be in vain. Though LNA determines the overall performance of the RF front-end receiver, the design of the LNA becomes a challenging task due to its operation on a very high frequency. Although there are many 5G bands available, special attention is given to the minimum atmospheric absorption band (28GHz) for integrated mm-wave SoC transceivers (Agiwal *et al.* 2016 and Rappaport *et al.* 2013). A broadband LNA with optimal power, gain, NF, and linearity would be preferable for building an efficient front-end receiver for 5G wireless communication.

3.2 Prior Work

Many works have been reported to design the 5G wireless systems using many topologies such as cascode topology, 4-stage CS, cascaded CG, and CS stages. The traditional design of LNA with these topology does not deliver good design tradeoffs. They either have a good gain or low NF. Recently many microwave components such as coplanar waveguide (CPW), coupler, etc are incorporated into the LNA design to enhance the performance. One such recent notable work based on this is reported in (Keshavarz Hedayati et al. 2018), where a 33GHz LNA for 5G wireless systems is designed in 28nm Bulk CMOS. This uses 2 stage cascode stage with microwave components such as CPW. The issue with the reported work is they work in narrowband with bandwidth less than 500MHz and have NF around 5dB. Additionally, metal transmission lines such as coplanar waveguides (CPW), microstrip lines (MSL), and stub lines are far more challenging to integrate into SoC integrated packages due to their electromagnetic (EM) waves fringing around silicon substrates. As a result, they may impair the performance of the LNA. As a result, this work proposes a unique suspended substrate line (SSL) with compatible SoC integrated solutions. SSLs incorporate air cavities into their structure to minimize electromagnetic (EM) losses into the substrates, and so improve the performance of an active component design. The 65nm technology PDK, on the other hand, takes a more suitable approach to SSL in order to ensure excellent electromagnetic compatibility. Samsung's 65nm technology is distinguished by a twin or triple well CMOS process on P-substrate, shallow trench isolation, low resistance nickel salicided polysilicon and diffusion, wire-bond pads or controlled collapse chip connections, and optional electrically programmable fuses. At millimeter-wave frequencies, the design accuracy of active and passive components must be extremely high in order to resolve parasitic connection concerns (Adabi et al. 2007) and (Sanduleanu *et al.* 2006).

With new built-in methodologies focusing on high-performance characteristics, LNA design becomes complicated. Additionally, this work prioritizes low loss, low power, and excellent electromagnetic compatibility by evaluating the effects of metal fill and layout design of LNA with SSL and T-shape Junction (MTEE). The authors of (Vigilante and Reynaert 2016), (Elkholy *et al.* 2018), and (Medra *et al.* 2014) recently demonstrated a variable gain in LNAs by changing bias supplies or tunable resistors. (Zhang and Zhao 2018) describes the implementation of a wideband low noise ampli-

fier operating at frequencies spanning from 21GHz to 27GHz for millimeter 5G base stations when the gain is compromised. A low noise amplifier for 5G wireless networks with a forward gain of 15dB and a noise figure of 1.5dB. It was designed in the 3.4GHz to 3.8GHz frequency band. However, this configuration is incompatible with millimeter-wave 5G communication (M Fan wei and J Tseng 2018). A low noise amplifier (LNA) has been reported for a 28GHz communication system for 5G applications, achieving a gain of 30.5dB with a gain ripple of 0.3dB between 24GHz and 30GHz and a minimum noise figure (NF) of 2.05dB at 25.5GHz (Zhou *et al.* 2018).

3.3 Conceptualization and Implementation

While there are numerous LNAs that employ an inductor and a transformer, a transmission line-based feedback LNA is designed to take advantage of the transmission line feedback path's noise cancelling and derivative superposition capabilities(Keshavarz Hedayati *et al.* 2018). This type of LNA exhibits a high degree of linearity and a low noise figure. Using the transmission line, we develop and incorporate unique SSL-based built-in approaches into the LNA design.

3.3.1 SSL Structural Analysis

Few authors such as (Zhen-Qi Huang and Quan-Rang Yang 1988), (Chu *et al.* 2019), (Wang and Ma 2019), and (Wang *et al.* 2017) have recently reported their work on Substrate Integrated Suspended Line (SISL) architectures. These structures were fabricated on Duroid substrates or Duroid/FR4 hybrids with low-loss tangents. However, these structures are needed to interface with other microwave components and are not included in the transceiver's active blocks. (Ke *et al.* 2018) describes the implementation of a dual-band LNA based on the SISL platform on a PCB board, achieving a high-power gain of 28.4dB and ultra-low noise of 1.1dB. However, this printed circuit board structure is incompatible with wideband applications. Additionally, it is incompatible with mobile receiver applications due to its larger area in mm. As a result, this research outlines the process of implementing the SSL structure using wafer technology. Figure 3.1 depicts the suggested suspended substrate line (SSL) in three dimensions, while Figure 3.2a depicts the slicing of individual layers. During the simulation, the left side of the figure serves as Port 1 and the right side serves as Port 2.



Figure 3.1: 3D Geometric view of proposed SSL

Three substrate layers are stacked in the following order: $Sub_1 - SiN$, $Sub_2 - SiN$, and $Sub_3 - SiN$. To maximize packaging wafer technological flexibility, each substrate is made up of dielectric silicon nitride (SiN) layers with a dielectric constant of 8.5, a loss tangent of 0.0036, a thickness of 100μ m, and a maximum operating frequency of 33GHz. M1 to M6 are the metal layers comprised of copper interconnects that are positioned on each side (top and bottom) of the substrate. The thickness of the top metal layer M1 and the bottom metal layer M6 is 12.5μ m, which is greater than the thickness of the other dielectric layers, which are 2.5μ m, in order to minimize the loss of the electromagnetic (EM) field created inside.

Figure 3.2b illustrates the cross-section of the proposed geometry. The EM fields generated in metal layers M3 and M4 are restricted to the air cavity alone in this case. The vias link the metal layers (M1-M2, M3-M4, M5-M6, and M1-M6). M1 and M6 serve as the ground, whilst M2 and M5 serve as the defective ground for metal layers M3 and M4. The air cavity that exists between all of the substrates traps the majority of the EM fields generated inside, significantly reducing loss and dispersion.



Figure 3.2: SSL component with (a) with splicing of layers (b) side view

As dielectric loss diminishes, the thermal noise generated by the dissipated energy decreases. These impacts are visible in Figure 3.2b cross-sectional area. Hollowed-out metal layers M2 and M5 increase the air cavity between the two substrates. The core substrate, which contains M3 and M4 metal layers, is slotted in such a way that the desired operating frequency is achieved. While the slot pattern in M3 is formed like a W-shaped with rotatory trays similar to the back of a refrigerator, it is noted that this new slotted design functions as a radiator and a way of heat convection. The slotted pattern's width and the airgap (h_l) between the substrate are predicted to be 0.7μ m and 500μ m, respectively. The vias are utilised to excite all layers between M1 and M6. To characterise the proposed SSL, full-wave electromagnetic simulations using the Methods of Momentum (MoM) are done on the RF ADS platform. The maximum pad radius of 3 via radii and the snap distance of 0.0004 is used to appropriately mimic



the parasitic. Surface current density is depicted in Figure 3.3 using post-processing

Figure 3.3: Surface Current density of SSL (a. M1 and M2 layers (b) M5 and M6 layers (c) M3 and M4 layers (d) Complete layers

EM simulations and is defined using physics model equations. SSL's characteristic impedance may be modelled using the equations 3.1 (Ma *et al.* 2017), (Chen *et al.* 2018), and (Vecchi *et al.* 2009).

$$Z_0 = 60 \ln\left[\frac{f(r)}{r} + \sqrt{1 + \frac{4}{r^2}}\right]$$
(3.1)

where $f(r) = 6 + (2\pi - 6)e^{-(30.666/r)^{0.7528}}$ and $r = \frac{L}{h+h_l}$, L is the length of the M3 layer, h is the thickness of the substrate and h_l is the air gap between substrates. The

effective dielectric constant (ε_{eff}) can be expressed as provided in equation 3.2.

$$\varepsilon_{eff} = \frac{1}{\left[1 + \frac{h}{h_l} \left(h_1 - h_{l1} \ln \frac{L}{h_l}\right) \left(\frac{1}{\sqrt{\varepsilon_r}} - 1\right)\right]^2}$$
(3.2)

Where

$$h_1 = \left(0.8621 - 0.1251 \ln \frac{h}{h_l}\right)^4$$
$$h_{l1} = \left(0.4986 - 0.1397 \ln \frac{h}{h_l}\right)^4$$

On layers M3 and M4, the W-shaped structure is embedded back to back with a rotatory tray slot structure. The maximum current intensity on layers M3 and M4 is depicted in Figure 3.3(c). This is evident in Figure 3.3(a) and 3.3(b) as a reflection of the current path on the M2 and M5 layers. The reflection is caused by radiation losses caused by the slotted pattern on layers M3 and M4. On M2 and M5, the hollow rectangular-shaped slot receives surplus current reflection on its surface, resulting in wideband performance. The validity of this assumption is demonstrated by the fact that the effective dielectric constants of suspended metal layers span a large range and may be stated as given in equation 3.3

$$\varepsilon_{eff} = \frac{1}{\left(1 - f_1 f_2\right)^2} \tag{3.3}$$

Where, $f_1 = 1 - \frac{1}{\sqrt{\varepsilon_r}}$ and $f_2 = \frac{1}{\sum_{i=0}^3 c_i \left(\frac{w}{h_l}\right)^i}$ and function C_i can be defined as equation 3.4

$$C_i = \sum_{j=0}^3 d_{ij} \left(\frac{h_l}{h}\right)^j \tag{3.4}$$

SSL's characteristic impedance is determined using the above-mentioned relationships, which are illustrated in Figure 3.4a. While Figure 3.4b illustrates the smith chart plot. From an observational standpoint, it is evident that impedance values are almost real for two resonant frequencies of 27.2GHz and 32.6GHz, respectively. And it is observed that the curve is almost entirely within the inductive zone. The proposed SSL's performance measures are initiated with EM wave simulation and analysis. Figure 3.5 illustrates the return loss simulation by altering the air gap between the substrates. To begin, the proposed SSL is designed to operate at a targeted operating frequency of 27GHz and a 500 μ m air gap distance. At a 500 μ m air gap, the desired resonance



Figure 3.4: Impedance performance of SSL with its (a) characteristic impedance and (b) smith chart

at 27.2GHz and 32.6GHz resonant frequencies, respectively. SSL's linearity versus frequency curve at a 500μ m air gap is depicted in Figure 3.6. The fluctuation of resonant frequencies with regard to the airgap between the metal layers of SSL is depicted in Figure 3.7. After accounting for the 0.5mm air gap, it is seen that the initial



Figure 3.5: Return loss Versus Frequency

resonant band (i.e., 27.2GHz) changed to a higher operating frequency. By contrast, the second resonant band (32.6GHz) lowered the operation frequency down. This is owing to the fact that radiation losses on metal layers are taken into account. This is illustrated in Figure 3.6 by the fact that phases are inverted at both 27.2GHz and 32.6GHz resonating frequencies. Figure 3.8 and Figure 3.9 illustrate the proposed SSL's simulated far-field pattern when phi is set to 90^o. At the first resonating frequency of 27.2GHz, E-phi and H-theta exhibit figure-of-eight forms, whereas E-theta and H-phi exhibit significant lobes oriented more toward 180^o. When the eight-shaped figure begins rotating in the direction of 45^o at the resonant frequency of 32.6GHz, E-theta and H-phi have large lobes that are inverted towards the direction of 0^o.

3.3.2 Analytical Modeling of SSL Inspired Parallel-Series Network

This subsection covers the design and analysis of an SSL-based parallel-series network, as seen in Figure 3.10a, as well as the layout design in Figure 3.10b. The parallel-series network is motivated by its dual-band operation, which is achieved by the selection of appropriate inductors and capacitors. However, designing high quality-factor on-



Figure 3.6: Phase versus Frequency



Figure 3.7: Variation of resonant frequencies with respect to airgap between metal layers of SSL



Figure 3.8: The far-field pattern of the slotted metal layer (a) E-theta and E-phi at 27.2GHz and (b) H-theta and H-phi at 27.2GHz



Figure 3.9: Far-field pattern of a slotted metal layer (a) E-theta and E-phi at 32.6GHz and (b) H-theta and H-phi at 32.6GHz

chip inductors at millimeter frequencies is a difficult task, resulting in the network's performance degradation. To circumvent such on-chip design issues, incorporating SSL into such an electrical network is an ideal solution. SSL has a high Q, a wide bandwidth, excellent temperature stability, and is also EM compatible. A parallel-series SSL-based network is made up of a parallel SSL-C network and a series SSL-C network.

This network aspires to have a broad bandwidth. To do this, the proposed network is analyzed for proper impedance matching using the smith chart method. The



Figure 3.10: SSL based parallel-series network with its (a) schematic and (b) layout design

millimeter-wave technique is based on the assumption of a parallel-series network and the formulation of impedance values using the EM signal of operations. The input source is set to 50Ω to ensure compatibility with the parallel SSL-C network. Similar to the series SSL-C network, a parallel SSL-C network has a shunt output impedance that matches the series SSL-C network. As illustrated in Figure 3.11, the impedance values of both combined networks are estimated using a smith chart for the frequency range of 27GHz to 34GHz. The technique for selecting the optimal load is determined by the matched series SSL-C network's impedance. The optimal impedances are determined between 27GHz and 34GHz, where 27GHz is the network's highest operational frequency. The smith chart graph demonstrates that half of the curve is composed of inductive impedances, while the other half is composed of capacitive impedances. At the operating frequency of 27GHz, the actual impedance is estimated as $(0.904 + j0.512)\Omega$ multiplied by the SSL characteristic impedance. At the last operating frequency, the impedance is $Z_0 \times (1.035 - j0.544)\Omega$. The ADS platform's



Figure 3.11: Smith chart plot of SSL parallel-series network



Figure 3.12: Reflection of power versus frequency

EM full-wave simulation generates a return loss curve versus frequency, as depicted

in Figure 3.12. The return loss was seen in both inductor-based and SSL-based networks. The parallel-series network with an inductor displays dual-band performance at two resonant frequencies of 27GHz and 32.5GHz, despite the fact that the output resistance is greater than the input resistance.



Figure 3.13: Transmitted power versus frequency

On the other hand, an SSL-based parallel-series network operates over a broad frequency range of 27GHz to 34GHz, with a minimum of 30.5GHz. Due to the network's wideband performance, a wide variety of impedance values can be tuned for the proposed LNA architecture. On-chip inductors do not degrade performance; rather, they provide dual-band operation. Because our goal is wideband, we picked SSL. Additionally, inductors are large and generate more noise than a transmission line. The S_{21} performance of the SSL-based parallel series network is shown in Figure 3.13, where the insertion loss fluctuates between -0.4dB at 27GHz and -0.3dB at 34GHz, with a minimum of -0.06dB at 30GHz, producing an inverted U-shape. S_{21} requires that passive components do not induce attenuation of the forward signal and should remain at 0dB. Figure 3.14 illustrates the input impedance of the real and imaginary values of an SSL-based parallel-series network. The values of the input impedance indicate that the network has adequate matching constraints and performs satisfactorily within the required operating band.



Figure 3.14: Input impedance versus frequency

3.3.3 Built-In Techniques Based CMOS LNA Design

This subsection discusses the design and implementation of a CMOS LNA that incorporates the proposed built-in approaches in order to attain the desired wideband performance. The proposed schematic CMOS LNA architecture is depicted in Figure 3.15. The proposed design features a two-stage cascode topology with an on-chip suspended substrate line (SSL), a T-shape junction (MTEE), inductors, and a MOM capacitor. SSL, MTEE, inductors, and Metal oxide Metal (MoM) capacitors are investigated and characterized using EM full-wave simulations and then optimized for a large Q value. SSL's distinctive individual performance is demonstrated in the preceding section, which demonstrates dual-band operation. SSL transmission lines, designated SSL_1 and SSL_2 , are used at the back-end of transistor M2 and M4 to minimize parasitic losses and act as a shield to prevent EM waves from fringing into the lossy silicon substrate, resulting in a high Q value (Vecchi et al. 2009). These SSL lines provide the necessary resistance to prevent the current pathway from reversing from cascode topology and to avoid power losses. Other transmission line components, such as SSL_3 , SSL_4 , and MoM capacitors, have been combined to create a parallel-series network that works as a bridge between the two cascode topologies. It utilizes input



Figure 3.15: Proposed schematic of built-in techniques-based CMOS LNA

inverted L-shaped (IIL) and output inverted SSL-based L-shaped (IIL-SSL) networks to achieve perfect alignment. Though there are other methods for matching inputs, we utilize a simple inverted L shape network because it is straightforward and appropriate for this design. A millimeter-wave complete simulation approach is used to solve both IIL and IIL-SSL networks. The T-shape junction (MTEE) supplies power and connects the SSL transmission lines with little loss. A precise design of the SSL and T-junction is critical for optimizing the architecture utilizing the PDK technology, as parasitic interconnects present a significant obstacle. The transistor layout was created in order to minimize interconnect parasitics, optimize device sizes, and design accurate mm-wave circuits for maximum performance in 65nm CMOS. Each stage of the proposed LNA adopts a cascode topology to improve reverse isolation, gain, and stability at mm-wave frequencies. The inductor L_3 (500pH) with a quality factor of 27 connects the drain-source interconnection's common gate (CG) and common source (CS) transistors for increased gain and to cancel out parasitic capacitances at these nodes. The second stage is meant to provide a higher gain than the first stage while maintaining a slightly lower NF due to the absence of source degeneration inductor L_2 . The intra-stage inductors L_3 , L_5 (500pH and 122pH with a quality factor of 27)
Comp	Value	Comp	Value	Comp	Value	Comp	Value
L_1	$475.8~\mathrm{pH}$	C_1	1 fF	C_7	$2.56~\mathrm{fF}$	R_1	$21.2~\mathrm{K}\Omega$
L_2	$613.5~\mathrm{pH}$	C_2	$276 \ \mathrm{fF}$	C_8	$5~\mathrm{fF}$	R_2	$21.2~\mathrm{K}\Omega$
L_3	$500 \mathrm{pH}$	C_3	$0.0021~{\rm fF}$	C_9	$36~\mathrm{fF}$	$M2_W$	$39.37 \mu \mathrm{m}$
L_4	325.4 pH	C_4	$0.0101 \ \mathrm{fF}$	C_{10}	$35~\mathrm{fF}$	$M3_W$	$75.3125 \mu \mathrm{m}$
L_5	$122.215 \ {\rm pH}$	C_5	$189.27~\mathrm{fF}$	C_{dc}	$2.2~\mathrm{fF}$	$M4_W$	$144.025 \mu\mathrm{m}$
L_6	229.28 pH	C_6	$0.0109 \; {\rm fF}$	$M1_W$	$63.875 \mu \mathrm{m}$		

 Table 3.1: Optimized values of designed LNA components

and transistor sizes are changed and optimized to transform the source impedance for noise impedance optimization. The aspect ratios of the device designs are employed to optimize the circuit, i.e., $M1 = 63.87/0.065 \ \mu\text{m}$, $M2 = 39.34/0.065 \ \mu\text{m}$, $M3 = 75.31/0.065 \ \mu\text{m}$, and $M4 = 144/0.065 \ \mu\text{m}$. M1 and M2 have 30 fingers, while M3 and M4 have 32; the finger width is set to 1μ m. Matching the input impedance Z_{in} employs inductive degeneration at the first CS stage: L_2 of 613.5pH with a quality factor of 18 degenerates the source of M1 and nearly equalizes the real part of Z_{in} to 50Ω ; while L_1 of 475.8pH with a quality factor of 30 in series with the gate nearly cancels out the imaginary part. L_1 's quality factor is deliberately selected higher than L_3 's because the input inductor's Q has a significant effect on NF. Between the output of the first cascode stage and the matching network is incorporated the capacitor C_{dc} . This is to prevent V_{DD} from being shorted to ground via L_4 , $MTEE_2$, and SSL_3 .

Tuning is used to optimize the load inductors and capacitors for interstage and output conjugate matching. The wideband gain response is obtained by adding the parallel series SSL-capacitor circuit between the first stage's CG and the second stage's CS and allowing it to function at the interstage matching network's resonant frequency. Two frequencies with zero input reactance can be produced using a good parallel to series interstage matching approach. In the 28-32GHz LNA, the parallel SSL-capacitor tank comprised of C_5 and SSL_3 not only resonates at 28.5GHz but also exhibits a broad response spectrum. The second cascode stage provides a larger gain than the first cascode stage while contributing the least amount of noise to the overall circuit. The parameter's value of the proposed LNA design is depicted in Table 3.1. All SSL's width (W) and Length (L) are chosen to be 25μ m and 100μ m respectively. Similarly, all the magic Tee MTEE dimensions W_1 , W_2 and W_3 are chosen to be 25μ m, 25μ m, and 50μ m, respectively.

3.4 Results Discussion

This section discusses the simulation and measurement results. Figure 3.16a and 3.16b illustrate the proposed LNA's layout and die microchip photograph. LNA has an estimated area of $0.35 \times 0.22mm^2$. The LNA is implemented in 65nm PDK technology and is produced utilizing a Magna Hynix Samsung RF 65nm process. The chip is fabricated layer by layer using substrates and MOM capacitors.



Figure 3.16: Photographs of LNA with (a) 3D layout and (b) die microchip photograph

On the silicon substrate, SSL and T-junctions are fabricated using the metalinsulator-metal (MIM) arrangement in three modes. The six-layer copper interconnects are produced utilizing a 65nm method that utilizes nanoclustering silica. These



Karl Suss (KSM) microprobe system





Figure 3.17: LNA measurement with its (a) experimental set-up and (b) on-wafer calibration standards

six layers are stacked on silicon wafers alongside other LNA components in 65nm to give flexibility and prevent parasitic losses caused by the CMOS mm-wave process. The passive components are mounted on the silicon chip using the quasi-mode layer method. Full-wave simulations and measurements of LNA performance parameters have defined performance parameters with optimal dimensions to achieve high performance over a wideband of operation. Figure 3.17a illustrates the experimental setup

for LNA measurement. S-parameters are measured for on-wafer LNA utilizing microprobe sets such as the Karl Suss (KSM) microprobe system, Keysight's PNA-X (N5245B) technology up to 50GHz, and Cascade Microtech microprobe tips with a 100 μ m pitch. Prior to performing measurements, the PNA and micro-probes must be calibrated. Microprobe systems have been calibrated using the short-open-loadthrough technique. Figure 3.17b illustrates an on-wafer calibration standard. Gold short circuits (Short), gold open pads (Open), and gold plus thin film resistors are included in the calibration sets (Load or Match). These calibration sets are used to accurately get the measured results of the Device Under Test (DUT). Figure 3.18



Figure 3.18: Simulated return loss Vs frequency of LNA with varying SSL length

illustrates the simulated return loss versus frequency plot obtained by increasing the SSL duration from 80μ m to 120μ m. As the length increases, it is noted that the upper sidebands are affected. This is because the surface current density converges at the SSL's edges, increasing the amount of current flowing through the return path. After examining the simulation results, the optimal value of L is deemed to be 100μ m. To determine the optimal forward gain of the LNA, the SSL and input-output matching networks are characterized. By varying the length of SSL from 80μ m to 120μ m, the maximum peak gain of 25dB is achieved for the optimized value of 100 µm, and that can be seen in Figure 3.19. It is also found that at a cut-off frequency of 28GHz,



Figure 3.19: Simulated Forward gain Vs frequency LNA with varying SSL Length

maximum gain of 21.5dB is achieved. However, substantial gain is required in 5G



Figure 3.20: S-parameter Vs frequency of LNA with proper matching

wireless networks to achieve intermodulation below the noise floor level. As a result, LNA designs incorporate IIL and IIL-SSL networks as input and output matching networks.



Figure 3.21: Simulated input impedance curve versus frequency (a) Rectangular Plot (b) Smith chart Plot

To acquire a wide impedance bandwidth, these networks are solved using a millimeterwave technique. Figure 3.20 illustrates the variations in return loss and the wide bandwidth achieved between 28GHz and 32GHz. Simultaneously, the forward gain plot in



Figure 3.22: Noise figure versus frequency

the Figure 3.20 achieves a gain of 26dB at 28GHz and a gain of 20dB at 32GHz. Additionally, the highest peak gain of 26dB is attained across the intended wide operating band. Gains decrease from 28GHz to 29.5GHz and 30GHz to 32GHz, respectively. This is because of the loading effect caused by cascade stages and SSL-based parallelseries networks that are implemented into the LNA design. Figure 3.20 also illustrates measured return loss and forward gain. The disparity between the measured and simulated values of S_{11} at low frequencies is most likely due to an inaccurate assessment of electromagnetic parasitic effect and concentric inductor impedance.

In Figure 3.21a, the real and imaginary input impedances of the LNA design are plotted, while the smith chart impedance is shown in Figure 3.21b. The above Smith chart plot demonstrates that the U-shaped pattern varies between the first and fourth quadrants. The bandwidth of the LNA design is posses variable inductance mixing both inductance and capacitance and has 50Ω matching limitations. Apart from this, the noise figure of the LNA is determined by varying the dc supply voltage from 1.2V to 2V, as illustrated in Figure 3.22. The optimal noise figure of the LNA is between 3.1dB and 2.5dB when supplied with 1.2V. The rationale for the improved noise figure at lower supply voltage is related to the cascode LNA topology's utilization of inductive load. At higher frequencies, the inductance drops dramatically, allowing for stronger resonant behavior at L_4 , C_4 , and L_6 , C_6 . As a result, these product components compensate for internal parasitics, resulting in the lowest noise figure possible for this design. For the RF frequency range, the stability is determined by using the Stability



Figure 3.23: Stability factor K of LNA

factor K (SFK) that depends on S-parameters. To achieve stability, it is required to have SFK > 1 and $\Delta < 1$. Both SFK and Δ are given by the equations given below

$$SFK = \frac{1 - |S_{11}|^2 - |S_{22}|^2 + |S_{11} S_{22} - S_{12} S_{21}|^2}{2|S_{12} S_{21}|}$$
(3.5)

$$\Delta = |\mathbf{S}_{11} \; \mathbf{S}_{22} - \mathbf{S}_{12} \; \mathbf{S}_{21}| \tag{3.6}$$

LNA stability is measured and plotted in Figure 3.23, where a positive stable point is achieved when L is at 100μ m, while for other values of L, the stability falls below 0. The linearity of LNA is measured and plotted in Figure 3.24. The linearity parameter a 1dB compression point (IP_{1dB}) of -19dBm, is achieved at the frequency of 30.5GHz.

The yield estimation of LNA utilizing Monte-Carlo simulation for return loss,



Figure 3.24: Measured 1-dB compressed point of LNA

forward gain, and noise figure is presented in Figure 3.25(a), (b), and (c). Under 1000 iterations, the yield percentage is 53.9% with a tolerance of 2% on all components. The figure of merit (FoM) of the CMOS LNA is calculated to be 31.04 at low noise and 18.11 at high noise, respectively. The expression to compute the FoM is given in equation 2.32 Table 3.2 compares the state-of-the-art of LNA to previously published ones.

3.5 Summary

This work discusses a CMOS LNA with a broad bandwidth of 28GHz to 32GHz for a 5G wireless cellular front-end receiver system. The proposed work is the product of extensive experimentation with numerous built-in techniques, including SSL, SSL parallel-series network, and full-wave mm-wave simulation. The SSL is meant to operate in two bands with 50 Ω matching limitations. The LNA is constructed using integrated methodologies to achieve high-performance system-on-chip (SoC) IC solutions. Additionally, the fabricated LNA has a wide bandwidth and a maximum forward gain of 26dB. At mm-wave frequencies, a minimum noise figure of 2.5dB is attained with a 1.2V supply voltage. Measurements and simulations are also used to



Figure 3.25: Yield estimation of LNA with (a) Iteration count vs return loss (b) Iteration count vs forward gain and (c) Iteration count vs noise figure

determine the appropriate linearity and stability parameters. The analysis of LNA revealed that the proposed CMOS LNA had the best performance and excellent electromagnetic compatibility, making them ideal for 5G cellular front-end receivers.

Parameters	Technology	Topology (µm)	Frequency (GHz)	Gain (dB)	NF (dB)	Power (mW)	IP_{1dB} (dBm)	Area (mm^2)	FoM
Keshavarz Hedayati et al. (2018)	28nm CMOS	2-stage cascode	33	18.6	4.9	9.7	-25.5	0.23	8.1
Qin and Xue $(2017b)$	65nm CMOS	CG-CS	7.6-29	10.7	5.6	12.1	I	0.3	10.52
Qin and Xue $(2017a)$	65nm CMOS	cascode	15.8-30.3	10.2	5.7	12.4	-0.5	0.18	10.88
Yu et al. (2017)	65nm CMOS	4-stage CS	54.4-90	17.7	7.4	19	-15.4	0.37	10.3
Huang $et al.$ (2009)	65nm CMOS	2-stage cascode	50-60	16	3.8	10	I	0.25	15.9
Ameen et al. (2017)	65nm CMOS	2-stage CS	26-30	11	1.79	I	I	I	I
This Work	65nm CMOS	2-stage cascode with SSL	27.8-32.5	25	2.5	7.2	-19	0.08	31

literature.
n the
und ii
\mathbf{fo}
works
prior
with
performance
ne LNA
of tl
Comparison
Table 3.2:

– Not calculated

Chapter 4

QUASI CIRCULATOR INTEGRATED LNA FOR X-BAND SYSTEMS

4.1 Introduction

The most important X-band applications include satellite communication, radio astronomy, broadcasting satellites, and radar. The majority of these applications require a low noise receiver to achieve the required sensitivity. The LNA is a critical component of the receiver circuit, amplifying the received signal while minimizing additional noise. The noise figure indicates the amplifier's performance; the lower the NF, the better. The design of the LNA circuit is critical in the receiver. It must consume little power without impairing gain or NF.

Also, the requirement for low-cost, durable, integrated SoC transceivers led the path for developing integrated RF front-end systems that include a circulator, LNA, and PA. One such solution is the RF front end, which is made of nano-ferrite material reported in Jadormio *et al.* 2017. Nonetheless, integrating a CMOS LNA with a ferrite-based circulator results in a significant insertion loss. Generally, circulators were constructed utilizing magnetic ferrite passive devices. A CMOS-based QC is used in this application, a novel design that enhances its performance in terms of gain and low-noise to the amplifier. The LNA is critical in the receiver chain block because it manages the signal-to-noise ratio (SNR), which requires a high gain to amplify a weak signal. Considering both the integrated SoC and performance of LNA, a novel

Quasi circulator integrated LNA, which maintains the highest flat gain, ultra-low NF, low power, and high linearity, is proposed for a full X-band of 8-12GHz in this work.

4.2 Prior Work

The authors in Lu et al. 2016 describe a viable design for an X-band LNA with 44dB gain and 7.5dB NF. The authors in Shang et al. 2012 report a 55dB increase of LNA with 4.9dB NF. These high gain LNAs are nonlinear and have an extremely high NF, which has an effect on the receiver system as a whole. The authors in Meghdadi et al. 2018 describe a very linear X-band LNA with a gain of only 13.6dB. Although the works Kanar and Rebeiz 2014 and Caliskan et al. 2019 describes ultra-low noise LNAs with NFs of 0.77dB and 1.2dB, they have a gain deficit of roughly 10dB. A UWB LNA with a maximum gain of 18.57 dB and a noise figure of 2.4 dB is attained in Kumar and Deolia 2019. This optimises its design using a specific sort of optimization (particle swarm optimization). The disadvantage is that it did not consider harmonics such as third-order intercepts or 1-dB compression points when determining the linearity of the constructed circuit. The authors in Arshad et al. 2015 describes a sub-10mW wideband LNA for UWB applications. While the harmonics and linearity are superior to the work in Kumar and Deolia 2019, the gain is only 10.3dB and the minimum NF is 3.68dB. A UWB LNA with double capacitor cross-coupled feedback is stated to have a gain of 21 dB and a noise figure of 2.38 dB in Ghaneei Aarani et al. 2019. The work in Meghdadi et al. 2020 describes a reconfigurable transceiver device for X band phased arrays with 13.5dB gain and 8.2dB NF. The authors in Dai 2017 presents a low power LNA operating at 0.6V and producing just 11.2dB gain and 3.8dB NF. The authors in Wang and Teng 2016 proposes a variable gain LNA with an interstage reflection type attenuator for the X band. Numerous LNAs with similar topologies are described in Liu et al. 2011, Chang et al. 2013 and Schwindt et al. 2005. The commonality in all of the above-mentioned works is that they have either a high gain with a high NF or an ultra-low NF with a low gain. The aforementioned publications mentioned above concentrated exclusively on LNA design. In X-band wireless transceivers, the CMOS integrated LNA with circulator is used. A CMOS-based quasi circulator for WLAN applications is disclosed in Hsieh et al. 2014. It employs a phase cancellation approach for narrowband operation at 2.4 GHz.

4.3 Circulators

Circulators are fundamental three-port nonreciprocal functional blocks that are used to segregate incident and reflected waves in microwave and millimeter systems. The incident signal circulates in only one direction, either clockwise or anticlockwise, to reach the next port. Ports are connected in the reverse direction. Circulator operates in duplex mode by using the same antenna for the transmitter and receiver. The simple circulator pictorial representation is depicted in Figure 4.1. The circulator's three primary parameters are return loss, isolation loss, and insertion loss. The circulator's S-parameters are a three-dimensional matrix comprised of nine entries numbered S_{11} to S_{33} . S_{21} (insertion loss), S_{23} , and S_{32} are critical LNA design parameters for the circulator (isolation). To avoid signal cross-over in duplex mode, the transceiver requires strong isolation, yet it requires low insertion loss to receive a signal. In conjunction with LNA design, the circulator must accomplish the aforementioned two objectives. This is a difficult endeavor, and the proposed Quasi circulator must accomplish these objectives, which will be detailed in greater detail in the following section.



Figure 4.1: Simple Circulator representation

4.4 Quasi circulator

The authors in Shin *et al.* 2008 used the term "quasi circulator" to refer to the circulator's electronic realisation. A quasi circulator is analogous to a circulator, except those two of the three ports are completely isolated in both clockwise and anticlockwise directions. A quasi circulator is one in which power is transferred from the port antenna to the port receiver and from the port transmitter to the port antenna, but not from the port receiver to the port transmitter. The proposed active Quasi circulator with a frequency range of 8 to 12 GHz is depicted schematically in Figure 4.2 as a stage-1. Ports 1, 2, and 3 are used to connect the antenna, receiver, and transmitter, accordingly. Transistors M_{N1} and M_{N2} are included in the current reused cascode stage, which reduces the insertion loss from the antenna to the receiver and improves the receiver's isolation from the antenna. Transistors M_{P1} and M_{P2} are included in another current reusing cascode stage to improve the circuit's stability. L_1 , L_2 , C_1 ,



Figure 4.2: Schematic of Proposed QC-LNA

 C_2 give input impedance matching to the antenna, while C_4 and L_4 offer transmitter impedance matching. The antenna signal is sent into M_{N1} 's common source stage, which amplifies it. It is transmitted to the receiver via M_{N2} . If sufficient optimization is performed, a minimum gain of roughly 1dB can be attained by gently easing the isolation loss in the receiver path. The M_{N3} with C_6 functions as a magnitude adjustment circuit, taking into account fluctuations in Process, Voltage, and Temperature (PVT) (Hsieh *et al.* 2014). By adjusting the bias voltage of M_{N3} , one can alter the loading capacitance (capacitance in the path of C_6 to ground via M_{N3}), i.e. the magnitude response from receiver to transmitter. This M_{N3} operating in a weak inversion zone will isolate the receiver from the transmitter by enabling the majority of the signal from the antenna to travel through C_3 and a little portion of the signal to the ground via the loading capacitance discussed previously. The signal from the transmitter to the antenna is routed through capacitor C_4 to the source terminal of M_{P1} . It will connect to the antenna through C_5 . The antenna's input impedance, Z_{in} , is provided below.

$$Z_{\rm in} = \frac{g_{mn1}L_2}{C_{gsn1} + C_2} + j\left(\omega_0 \left(L_1 + L_2\right) - \frac{1}{\omega_0 \left(C_{gsn1} + C_2\right)}\right)$$
(4.1)

Here C_{gsn1} and C_{gmn1} are the gate-source parasitic capacitance and transconductance of M_{N1} . Quasi circulator has three ports whose S-parameter matrix is given below in equation. All the S-parameters are essential for the Quasi circulator.

$$[S]_{3\times3} = \begin{bmatrix} S_{11} & S_{12} & S_{13} \\ S_{21} & S_{22} & S_{23} \\ S_{31} & S_{32} & S_{33} \end{bmatrix}$$
(4.2)

The S-parameters are divided into three categories: return loss (S_{11}, S_{22}, S_{33}) , insertion loss (S_{21}, S_{13}) , and isolation $(S_{12}, S_{23}, S_{32}, S_{31})$. S_{22}, S_{21}, S_{23} , and S_{32} are significant in relation to the LNA design that is attached to port 2 of this QC. Figure 4.3 illustrates the simulated S-Parameter plots of the proposed quasi circulator. At the antenna port S_{11} , the receiver port S_{22} , and the transmitter port S_{33} , the simulated input return losses are all greater than 12dB. Across the whole X-band, the simulated insertion losses S_{21} (from the antenna to the receiver) and S_{13} (from the transmitter to the antenna) are 1.1dB and 1.5dB, respectively. Across the full X-band frequency of operation, the isolation between S_{32} (receiver to transmitter), S_{12} (receiver to antenna), S_{23} (transmitter to receiver), and S_{31} (antenna to transmitter) is 40dB, 37dB, 17dB, and 14dB, respectively.



Figure 4.3: Simulated S-parameters variations of QC

4.5 LNA design

Figure 4.2 depicts the proposed QC-LNA. The antenna signal is available at port 2 of the quasi-circulator and is sent to the LNA block through a capacitor C_3 . This block implements an input network using an inductive degeneration common source structure. This architecture is capable of delivering a high gain at a low NF and a low power consumption (Yueting *et al.* 2012). The inductor (L_s, L_g) , capacitor c_t and gate-source parasitics of M_1 (C_{gs1}) create the required input matching condition. The Z_{in1} expression is given below.

$$Z_{in1} = \frac{g_{m1}L_s}{C_m} + j\left(\omega_0 \left(L_s + L_g\right) - \frac{1}{\omega_0 C_m}\right)$$
(4.3)

Here g_{m1} is the transconductance of M_1 and $c_m = c_t + C_{gs1} + C_{gs2}$. The transistors M_1 and M_2 are biased with the same current. In DC mode, this LNA circuit block operates as a single stage; in AC mode, it operates as a two-stage. C_r will act as a grounding path for the M_2 source in AC mode, allowing for high gain. The signal is supplied to M_2 's gate through L_a , which avoids signal leakage to ground. By avoiding signal leakage, the optimal value of L_a with a high Q factor ensures good gain. As a

result, this inductor La is more sensitive to gain. L_a and C_a form a parallel tank to provide a high impedance to the signal for grounding at the operational frequency. L_b and C_b , in turn, produce a series tank with a low impedance at the operating frequency required to transmit the signal to M_2 's gate. This then connects the first and second sections of the reused structure. Parallel placement of transistor M_3 to M_1 ensures stability and cancels out third-order harmonics when suitably sized and biased. For a transistor M_1 , the third-order intercept point IIP_3 and drain current are calculated as follows (Nikbakhsh *et al.* 2018).

$$IIP_{3} = 20\log\sqrt{\frac{4}{3}\left|\frac{g_{m1}}{g_{m1}''}\right|}$$
(4.4)

$$I_{d1} = g_{m1}V_{gs1} + g'_{m1}V_{gs1}^2 + g''_{m1}V_{gs1}^3 + \cdots$$
(4.5)

Here g_{m1} is the transconductance of M_1 , also g'_{m1} and g''_{m1} are the second and thirdorder nonlinear coefficients of this transistor, respectively. According to Taylor series

$$g_{m1} = \frac{\partial I_{d1}}{\partial V_{gs1}} \tag{4.6}$$

$$g'_{m1} = \frac{1}{2!} \times \frac{\partial^2 I_{d1}}{\partial V_{gs1}^2}$$
(4.7)

$$g''_{m1} = \frac{1}{3!} \times \frac{\partial^3 I_{d1}}{\partial V^3_{gs1}}$$
(4.8)

According to equation 4.4, as g''_{m1} is decreased, IIP_3 is increased, and the circuit linearity is ameliorated. To keep g''_{m1} about zero for a wider range of V_{gs1} the transistor M_3 is employed in parallel to M_1 . The M_3 works in the weak inversion region. It is biased and sized in a way that the positive peak of the g''_{m3} curve can cancel the negative peak of the g''_{m1} curve. Thus, this improves the IIP_3 of the circuit. The optimum value of g_m for every transistor is required to achieve high gain, and it depends on V_{gs} and W. By optimizing these two parameters, the best g_m could be achieved. The gain contribution by g_m through W will be shown in the next section. The LNA is designed so that it is unconditionally stable (SFK > 1) in the specified frequency range. The M_5 in parallel with inductor L_x acts as a variable impedance if it is biased in the deep triode region. To have a maximum gain, load matching is critical, and that can be done if the M_5 is biased in the deep triode region. By choosing the proper value of V_{B9} , L_x , and W of M_5 that can be achieved, resulting in a high gain at the output. The expression for the total gain is given below.

$$A_{V,T} = A_{v-s1} \times A_{v-s2} \times A_2 \tag{4.9}$$

$$A_{v-s1} = -\frac{g_{m1}}{2} \times \frac{sL_a \left\| \frac{1}{sC_a} \right\| \frac{1}{sC_a} \left\| \left(sL_b + \frac{1}{sC_b} + \frac{1}{sC_{tot,c}} \right) \right.}{\left(L_g + L_s \right) C_m s^2 + \left(sL_s \left\| \frac{1}{sC_{ps1}} \right) g_{m1} + 1}$$
(4.10)

$$A_{v-s2} = \frac{-g_{m2}C_{b2}}{C_{b2} + C_{tot,n}} \times \frac{C_b}{s^2 L_b C_b C_{gs2} + C_{gs2} + C_b} \times sL_d \left\| \frac{1}{sC_{pd2}} \right\| r_{02} \left\| \frac{C_{b2} + C_{tot,n}}{sC_{b2}C_{tot,n}} \right]$$

$$A_2 = \frac{g_{m4}\omega_0 L_x}{1 + \omega_0 L_x \mu_n C_{ox} \left(W_5/L_5\right) \left(V_{B9} - V_{th5}\right)}$$

$$(4.12)$$

Here A_{v-s1} , A_{v-s2} , A_2 are the gain of the first stage, second stage, and gain due to M_5 , respectively. R_1 - R_4 is large enough to be ignored in all equations. Here g_{m1} , C_{ps1} and C_{pd1} are the transconductance of M_1 , the parasitic capacitance at source, and drain, respectively. Here g_{m2} , C_{ps2} , C_{pd2} , and r_{02} are the transconductance of M_2 , the parasitic capacitance of source, the parasitic capacitance of drain, and channel resistance of the M_2 . The second stage current reused structure, and the variable impedance load stage is modeled as $C_l tot, c$ and $C_l tot, n$.

4.6 Integrated CMOS LNA Design

Design optimization is an essential step to achieve the best performance of the QC-LNA circuit. The design goals of QC-LNA are:

- To achieve a flat gain across the entire X-band with minimum variation. It will be instrumental in the place where the signal strength is feeble.
- To achieve the noise figure less than 1.5dB to make a high SNR value.
- To achieve minimum insertion loss in the Quasi-circulator receiver path to avoid the incoming signal strength drop.
- To achieve better isolation above 20dB to avoid cross-interference during the duplex mode.

The Quasi-Newton (QN) control algorithm is utilised in this example because it is an effective way for solving nonlinear equations with a limited number of iterations. Each

objective is defined as a function of a nonlinear equation. The generalized algorithm procedure is reported in (Medra *et al.* 2014).

$$F(x) = 0 \tag{4.13}$$

Here F(x) is considered as a function of gain, NF, and return loss. If x_k is an approximate solution of equation 4.13 then the quasi-newton iterative formula can be described as follows:

$$\begin{cases} x_{k+1} = x_k + B_k^{-1} F(x_k), \\ B_{k+1}(x_{k+1} - x_k) = F(x_{k+1}) - F(x_k), \\ B_{k+1} = B_k + \Delta B_k, \quad \operatorname{rank}(\Delta B_k) = m \ge 1. \\ (k = 0, 1, \ldots), \end{cases}$$
(4.14)

Here B_k is Broyden rank. This formula is called quasi newton equation when $B_{k+1} = B_k + (y_k - B_k s_k) \frac{s_k^T}{s_k^T s_k}$ The design steps of the algorithm are shown as a flowchart shown in Figure 4.4. The optimized component values are listed in Table 4.1. Let us consider the circuit's gain; it depends on W_2 , L_b , W_3 , W_5 , and so on. The gain expression is given in equation 4.9 is taken as F(x). The initial value is provided to all the parameter values, such as the width of transistors, inductors, and capacitors which is x_k and it is input to the algorithm. It will compute the gain value and compare it with the goal (30dB). Generally, it will be less, then it will calculate $x_{(k+1)}$ and update the matrix, like this the cycle continues till $F(x_{(k+1)})$ reaches 30dB, which is our goal. The final $x_{(k+1)}$ matrix gives all the parameters values. Like this, all three goals are computed at the same time. By this way the optimized parameter value in the circuit design is computed with a minimum number of iterations also with a minimal computational time.

4.7 Performance Evaluation of QC-LNA

The LNA is implemented using RF 65nm process design technology and fabricated using the Magnachip Hynix Samsung process. The fabricated chip is a layer-bylayer formation using substrates and metal oxide metal (MoM) capacitors. The sixlayer copper-interconnects are fabricated using nano-clustering silica. These six-layers are stacked up on silicon wafers along with other components of LNA to provide



Figure 4.4: Proposed Flow chart for optimization using QN algorithm of QC-LNA

flexibility, which in turn avoids parasitic losses by CMOS mm-wave process. The coils were realized using an RFIC inductor toolkit pre-configured for an RF 65nm CMOS process technology for the Electromagnetic design system in ADS v. 2019. An octagonal inductor with width/spacing is chosen as $9\mu m/3\mu mm$, and 2.3 of the number of turns have been employed. The stacking multi-layer process is used for mounting the passive components on the silicon die. The microchip die photograph is shown in Figure 4.5. The total core area of the chip is $0.84 \times 0.52mm^2$. The full-wave simulation and measurement readings of LNA have characterized performance parameters with optimized dimensions to achieve high performance over a wideband

Comp	Value	Comp	Value	Comp	Value	Comp	Value
M_{N1W}	$99~\mu{ m m}$	C_1	1.6 pF	C_a	$0.19 \mathrm{nF}$	L_d	2.1 nH
M_{N2W}	$99.3~\mu\mathrm{m}$	C_2	$14 \ \mathrm{fF}$	C_b	$0.53 \ \mathrm{nF}$	L_a	2 nH
M_{N3W}	$110~\mu{\rm m}$	C_3	1.2 pF	C_r	$0.85~\mathrm{nF}$	L_b	4 nH
M_{P1W}	$5.8~\mu{ m m}$	C_4	$0.3 \mathrm{ pF}$	L_1	$1.7~\mathrm{nH}$	L_x	$1.9~\mathrm{nH}$
M_{P2W}	$1~\mu{ m m}$	C_5	$0.5 \ \mathrm{pF}$	L_2	$50 \mathrm{pH}$	R_1	$25~\mathrm{K}\Omega$
M_{1W}	$33.8~\mu\mathrm{m}$	C_6	$0.1 \ \mathrm{pF}$	L_3	1.1 nH	R_2	$50~{ m K}\Omega$
M_{2W}	$23.3~\mu\mathrm{m}$	C_t	$91~\mathrm{fF}$	L_4	$0.9 \ \mathrm{nH}$	R_3	$64~{ m K}\Omega$
M_{4W}	$99~\mu{ m m}$	C_{b1}	$0.2 \ \mathrm{nF}$	L_g	$1.7 \ \mathrm{nH}$	R_4	$50~{ m K}\Omega$
M_{5W}	$99~\mu{\rm m}$	C_{b2}	$0.7 \ \mathrm{nF}$	L_s	$64 \mathrm{pH}$	R_B	$50~{ m K}\Omega$

Table 4.1: Optimized values of designed QC-LNA components



Figure 4.5: Microchip die photo of fabricated QC-LNA

of operation.

4.7.1 Statistical Analysis

As discussed in section 2.8 the yield analysis is essential in estimating the actual yield and the confidence levels. The designed LNA is tested for the error tolerance of 5% and 1% under Gaussian distribution. The results for Gain, NF, return loss are shown in Figure 4.6 (a), (b), and (c) respectively. It is noted that the stable gain of 30dB, NF of 1.5dB, and return loss of 12dB are achieved with 1% tolerance. It results in a yield estimation of 100%, giving a 100% confidence level for 1000 iterations. But for the 5% tolerance yield estimation, 13% of the iteration failed to result in the yield of 87%. The parameters' limit is set as a minimum 28dB gain, maximum 2dB NF, and a minimum 10dB return loss. But usually, CMOS process PVT variations are less than 1% only, which means the designed QC-LNA provides a stable and accurate design.



Figure 4.6: Yield estimation of QC-LNA (a) Gain (b) NF (c) Return Loss

4.7.2 DoE Analysis

As discussed in section 2.9 Design of Experiments (DOE) is a data-driven technique for robust design to understand the design variable's contribution in achieving the final goal listed before. The main design variables and their assigned alphabet are listed in Table 4.2. The results of the design variables vs. contribution to the specific goal such as NF, Gain, and return loss are shown in Figure 4.7 (a), (b), and (c). It is noted that M_{2W} , L_b , and their interaction contribute more than 80% of the gain. So, these two parameters are more sensitive to gain. It is also noted that M_{1W} contributes more than 80% of NF. M_{4W} contributes more than 35% of return loss. The design variables which contribute less than 1% are neglected in Figure 4.7 for better clarity. Extra care has been taken for these sensitive and more contributing parameters while fabricating to get the dedicated, accurate response in the measurement results.



Figure 4.7: Design variable contribution to (a) Gain (b) NF (c) Return Loss

4.7.3 Experimental Analysis

S-parameters are measured for on-wafer QC-LNA using microprobe sets such as the Karl Suss (KSM) microprobe system, Keysight's PNA-X (N5245B) technologies up

Assigned Alphabet	Design variable	Assigned Alphabet	Design variable
A	C_a	\mathbf{F}	M_{2W}
В	C_b	G	M_{3W}
\mathbf{C}	L_a	Н	M_{4W}
D	L_b	Ι	M_{5W}
Ε	M_{1W}		

Table 4.2: Design Variables assigned for DoE Analysis

to 50GHz, and Cascade Microtech microprobe tips with a $100\mu m$ pitch. Prior to performing measurements, the PNA and micro-probes must be calibrated. Microprobe systems have been calibrated using the short-open-load-through technique. Gold short circuits (Short), gold open pads (Open), and gold plus thin film resistors are included in the calibration sets (Load or Match). These calibration sets are made using the RF 65nm CMOS technology on a silicon substrate. The measurement setup is shown in the Figure 3.17a and 3.17b.



Figure 4.8: Simulated input impedance using smith chart plot

The input impedance of QC-LNA is shown as a smith chart in Figure 4.8. It is noted that the impedance is slightly capacitive due to the dominance of the capacitors C_1 and C_2 . Stability of the LNA is mandatory for the circuit to function properly.



Figure 4.9: Stability factor K of QC-LNA



Figure 4.10: Noise figure of designed QC-LNA

The expression for the Stability factor K is expressed in equations 3.5 and 3.6. LNA stability is measured and plotted in Figure 4.9. It is noted that the designed QC-LNA is unconditionally stable in the desired frequency range of 8 to 12GHz.

Figure 4.10 and Figure 4.11 illustrate the measured and simulated NF and Sparameters for QC-LNA. It is noted that the NF ranges between 1.25dB and 1.4dB at 8GHz and 12GHz. At 9.5GHz, it has a minimum of about 1dB. Gain of 30dB is reached with only a 0.5dB variance across the frequency range of 8-12GHz. When compared to the previously stated works, this is phenomenal performance in terms of gain for having such a little variance across the full 4GHz bandwidth.



Figure 4.11: S Parameters of designed QC-LNA

The return loss (S_{11}) is sufficiently well below 10dB to avoid any reflection. Other parameters, such as S_{12} and S_{22} , also show satisfactory performance. In Figure 4.12, the measured P_{in} vs. P_{out} (fundamental @ 9.3 GHz) and Pin vs. Pout (Third-order harmonics) to get the IIP_3 and IP_{1dB} is plotted. It is worth noting that the compression point for the Input 1 dB (IP_{1dB}) is -15dBm. At +10dBm, the third-order input intercept point (IIP_3) is obtained. Additionally, the plot demonstrates that the developed QC-LNA achieves good third-order output intercept point (OIP_3) and Output 1dB compression point (OP_{1dB}) values of +40dBm and +13.8dBm, respectively. The Table 4.3 compares the state of the art of the designed LNA to previously published ones, and it shows a phenomenal FoM of 57.4, which is the highest among all the reported works.

Parameters	Technology	Frequency (GHz)	Gain (dB)	NF (dB)	Power (mW)	$IP_{\mathrm{l}dB}~\mathrm{(dBm)}$	IIP_{3} (dBm)	Area (mm^2)	FoM
Hayati et al. (2020)	180nm CMOS	3-10.6	13.4	2.3	11.56	∞	I	0.55	8.38
Davulcu et al. (2018)	130nm SiGe BiCMOS	8-12	20	1.4	100	-0.75	5.25	0.72	2.2
Li et al. (2017)	130nm CMOS	3-12	11.5	4.75	8.5	I	2-	0.86	3.41
Lee and Kwon (2018)	110nm CMOS	3.10	17.6	3.7	2.9	I	-4.6	I	18.6
Çalışkan <i>et al.</i> (2019)	130nm SiGe BiCMOS	6-12	10	0.77	19.8	1.5	I	0.4	10.1
Ghaneei Aarani et al. (2019)	180nm CMOS	3.1-10.6	21	2.4	4.34	I	\$	1.05	49
Choi et al. (2021)	65nm CMOS	6.7 - 15.3	20	2.1	12.8	-17	6-	0.15	13.3
This Work	65nm CMOS	8-12	30	1.05	10.44 5	-15	+10	0.437	57.4

works.
prior
with
performance
٩V
E
the
of
Comparison
ŝ
4
Table

- Not calculated



Figure 4.12: Input power (P_{in}) vs. Third order harmonics (IP_3) and Output Power $(P_{out} \text{ fundamental})$

4.8 Summary

The integrated QC-LNA in 65nm CMOS is reported in this work using a two-stage current reuse approach improved using the QN algorithm. The statistical and DoE analyses are presented to evaluate the design's yield and robustness. This QC-LNA produces a constant gain of 30dB with a variance of only 0.5dB throughout the whole 4GHz bandwidth. It achieves an extremely low noise floor of 1dB. By dissipating only 10.44mW of power, the developed QC-LNA achieves IP_{1dB} of -15dBm, IIP_3 of +10dBm, OP_{1dB} of +13.8dBm, and OIP_3 of +40dBm. Losses and mismatches are decreased by integrating QC with LNA. The comparison to state of the art demonstrates that the developed QC-LNA produces reasonably good mean values for all parameters, resulting in the highest FOM.

Chapter 5

LNA INTEGRATED WITH RESONATOR FOR SATELLITE NAVIGATION SYSTEMS

5.1 Introduction

Due to the rapid growth of wireless communication infrastructure in recent years, there is a need for low-cost, robust, and high-performance receivers. Satellite navigation systems are one of the most efficient forms of wireless communication. For instance, a receiver for the Indian Regional Navigational Satellite System (IRNSS) can work in the L and S frequency bands between 1GHz and 3GHz. This study is primarily focused on producing an integrated FBSSIR LNA prototype for the IRNSS receiver, which will be used in smart city projects for intelligent transportation, vehicle monitoring, and precise timing. Figure 5.1 illustrates a typical block diagram of an IRNSS-based transportation management system. Transportation management requires a central server to oversee and monitor all operations, as well as an IRNSS receiver system to communicate between satellites and other devices. This system can monitor the functions of all buses and other vehicles via GSM/GPRS, which is controlled by a central server. The IRNSS satellite transmits continuous signals to the device's IRNSS receiver system and other components (Karthick and Kashwan 2015). The IRNSS receiver device's primary block is a filter and gain stage, often a Low Noise Amplifier (LNA). This work focuses on designing LNA integrated with resonator for satellite navigation systems.



Figure 5.1: Block diagram of a transportation management system using IRNSS receiver

5.2 Prior Work

The primary filters are constructed using either LC or transmission line technology. Inductors and capacitors have higher losses at RF frequencies, and at high frequencies, L might act like C and vice versa. Additionally, a high-value Q inductor increases the circuit's cost. The most cost-effective solution is to use transmission line-based filters. The first microstrip dual-mode resonators is reported in Wolff 1972. Other researchers such as Zhu and Wu 1999, Hong et al. 2007, Lee et al. 2000, Liao et al. 2007 and Zhang et al. 2008 later proposed dual-mode filters with varied geometry. (Hong et al. 2007) and (Lee et al. 2000) demonstrate E-type dual-mode resonators and a non-uniform T-type open stub, respectively. Second-order harmonics were generally seen at $2f_0$ in a standard E-type uniform dual-mode impedance resonator, where f_0 is the center frequency reported in Liao et al. 2007. Additionally, there are now four methods for fabricating multi-band Band Pass Filters, with a multimode resonator being the most fundamental of these four. In Luo et al. 2011 and Sun et al. 2013 ring multimode resonators are shown that combine two or more resonant modes of the multimode resonator. Zhu and Menzel 2003 and Balalem et al. 2009 discusses high-frequency selectivity filters in triple mode. Makimoto and Yamashita 1980 discusses the relationship between harmonic frequency and impedance ratio in a stepped impedance resonator. Jiang *et al.* 2016 presents a dual-band open and short stub-loaded stepped impedance resonator that employs a two-fourth-order cross-coupling method in the transmission path to create the quasi-elliptic response. Yang *et al.* 2005 describes a resonator based on a compact microstrip resonant cell (CMRC). In Hagag *et al.* 2019, they introduced a resonator based on an evanescent-mode cavity, in which they integrated cavity resonator-based LNA co-design and obtained tunable filtering capabilities. Shen *et al.* 2009 presents a three-mode stepped impedance resonator with a sharp skirt and a broad stopband.

The Works reported in Lu *et al.* 2010, Arsalan and Wu 2019, Fouad 2003, Gonzalez 1997, Gray *et al.* 2008 and Fouad 2001 discuss on LNA design without a resonator for wireless applications, and the majority of these works achieved narrowband operation. For instance, Lu *et al.* 2010 describes a microstrip line-based LNA that achieves 2.2-2.6GHz. In Arsalan and Wu 2019, an S-band LNA is developed for satellite navigation systems using microstrip lines with a frequency range of 2.3-2.6GHz. Fouad 2003 discusses an RF cascode BJT-LNA with shunt series input matching. Aneja *et al.* 2021 presents a dual-band microwave integrated circuit (MIC)-based LNA working at 1.1GHz and 2.4GHz that achieves 24.4dB and 20.1dB with 2.6dB and 3dB NF. Bisht *et al.* 2020 discusses a CMOS-based dual-band LNA operating at 900MHz and 2.4GHz with a power gain of 9.3dB and 10.6dB and a noise figure of 3.7dB and 3.1dB.

The limitations of the work discussed above are that all of the resonators have extremely narrow bands and hence cannot be used for wideband applications. Typically, a resonator is built for filtering purposes in a receiver or transmitter; however, creating and testing a resonator on an individual basis is acceptable. However, compatibility concerns, such as impedance mismatch, arise when the resonator is integrated with a Low Noise Amplifier or a Power Amplifier. Except for Hagag *et al.* 2019, none of the works reported and discussed above designed anything other than the resonator.

As a result, this work proposes a novel resonator named Folded Butterfly-stub Stepped Impedance Resonator (FBSSIR) integrated with a core LNA circuit named FBSSIR integrated LNA. FBSSIR uses a Butterfly-open stub and a Folded T-type open stub in the Stepped Impedance resonator's central plane. As two stubs are introduced, two transmission zeros can be obtained simultaneously. The proposed FBSSIR is compact and suppresses the spurious passband compared to the one given in Shen *et al.* 2009. This also provides the required input and output matching to the core LNA circuit.

5.3 Impedance Match Design Using FBSSIR

This section describes the impedance match design approach using folded butterfly stub stepped impedance resonator. Section 5.3.1 elaborates on the stepped impedance resonator structure's fundamentals, while the proposed FBSSIR structure and its analysis are discussed in Section 5.3.2.

5.3.1 Fundamentals of Stepped Impedance Resonator (SIR)

The basic geometry of SIR and its equivalent circuit diagram is shown in Figure 5.2(a) and (b). To understand about fundamentals of a stepped impedance resonator, a $(\lambda/2)$ SIR consists of a low-impedance part (LIP) with characteristic impedance Z_1 and an electrical length of 2θ in the middle while two high-impedance parts (HIP) with characteristic impedance Z_2 and electrical length of θ at both sides is considered. The mathematical relationship between the resonant frequencies, characteristic impedance ratio, and the electrical length of SIR can be represented using equations 5.1 and 5.2 (Chu and Chen 2008).

$$\theta = \arctan\sqrt{R_Z} \tag{5.1}$$

$$\frac{f_{S1}}{f_0} = \frac{\pi}{2\arctan\sqrt{R_Z}} \tag{5.2}$$

Where f_0 is the resonant or central frequency of the SIR, f_{S1} is the first spurious resonant frequency of the SIR, θ is the electrical length of each part, and $R_Z = Z_2/Z_1$ is the characteristic impedance ratio of HIP and LIP. The characteristic impedance of each part with the microstrip as the fundamental structure is determined by the formula given in equations 5.3 and 5.4.

$$Z_0 = \frac{120\pi}{\sqrt{\varepsilon_{re}}} \frac{1}{W/h + 1.393 + 0.667\ln(W/h + 1.444)}$$
(5.3)

$$\varepsilon_{re} = \frac{\varepsilon_r + 1}{2} + \frac{\varepsilon_r - 1}{2} \left(1 + \frac{12h}{W} \right)^{-\frac{1}{2}}$$
(5.4)

Where W is the width of the conductor, h is the thickness of the grounded dielectric substrate and ε_r is the dielectric constant of the substrate. If f_0 and f_{S1} are provided,

the physical size of SIR can be calculated with Equations 5.1, 5.2, 5.3 and 5.4. The input impedance relation for each part of SIR can be calculated as in equations 5.5 and 5.6 (Hou *et al.* 2020).

$$|Z_{in}(l)| = \left| Z_0 \frac{Z_L + jZ_0 \tan\beta l}{Z_0 + jZ_L \tan\beta l} \right|$$
(5.5)

$$|Z_{in}| = \left| Z_0 \frac{Z_L + jZ_0}{Z_0 + jZ_L} \right|$$
(5.6)

Where Z_0 is the characteristic impedance, Z_L is the load impedance and $\theta = \beta l; \beta = \frac{2\pi}{\lambda_g}$ and $\lambda_g = \frac{\lambda_0}{\sqrt{\varepsilon_{re}}} = \frac{c}{f_0\sqrt{\varepsilon_{re}}}$. To make a folded structure with a gap using SIR, a folded open-loop SIR can be modeled as a coupling capacitor C_m in the equivalent circuit shown in Figure 5.2(c). Now the characteristic impedance Z_0 come along with the electric length of the series step impedance part, and the separated LIP are brief to θ' and $2\theta' + \theta'_s = 2\theta$.



Figure 5.2: (a) Schematic diagram of SIR (b) The equivalent circuit diagram of the SIR and (c) The equivalent diagram of Folded SIR with a gap.

To increase transmission zero and compact size, SIR is bent and incorporated with the folded T-type open stub and butterfly open stub in the central plane with SIR to make a proposed FBSSIR which will be discussed in the next subsection.

5.3.2 FBSSIR design and analysis

The geometry of the proposed FBSSIR is shown in Figure 5.3a. It consists of a halfwavelength ($\lambda/2$) SIR, a Butterfly type open Stub, and a folded T-type open stub on its central plane. The physical dimension of the parameters shown in Figure 5.3a are: a=10mm; b=10.5mm; c=9mm; d=5.2mm; e=6.5mm; f=15mm; g=1.5mm; h=2mm; i=1mm; j=4mm and the gap S1=S2=0.2mm. The fabricated top view of the proposed FBSSIR is given in Figure 5.3b. The modeling and simulation of the FBSSIR structure are done using an advanced design system (ADS) v. 2019 RF simulator, while fabrication is done using basic fabricated methods of PCB packaging technology.



Figure 5.3: Proposed FBSSIR with is (a) Schematic view (b) Fabricated view

In order to calculate the transmission zeros and better return loss of FBSSIR, the proposed resonator coupling scheme is considered, which is shown in Figure 5.4. This coupling scheme can operate in three modes, two even and one odd mode. The coupling matrix K corresponding to the designed resonator is given in equation 5.7.


Figure 5.4: Coupling Scheme of the FBSSIR

$$K = \begin{bmatrix} 0 & K_{S1} & K_{S2} & K_{S3} & 0 \\ K_{S1} & K_{11} & 0 & 0 & K_{1L} \\ K_{S2} & 0 & K_{22} & 0 & K_{2L} \\ K_{S3} & 0 & 0 & K_{33} & K_{3L} \\ 0 & K_{1L} & K_{2L} & K_{3L} & 0 \end{bmatrix}$$
(5.7)

The above matrix K, $K_{S2} = K_{2L}$, $K_{S1} = K_{1L}$, and $K_{S3} = K_{3L}$ proves that the designed resonator is symmetrical. There is no coupling among the three modes because the split mode frequency and self-resonant frequency are equal. The generalized coupling matrix for the above with transmission zeros at $\pm 2j$ and return loss of 20dB is obtained by using the optimization technique from Shen *et al.* 2009 and given in equation 5.8.

$$K = \begin{bmatrix} 0 & 0.451 & 0.779 & 0.451 & 0\\ 0.451 & 1.05 & 0 & 0 & 0.451\\ 0.779 & 0 & 0 & 0 & -0.779\\ 0.451 & 0 & 0 & -0.964 & 0.451\\ 0 & 0.451 & -0.779 & 0.451 & 0 \end{bmatrix}$$
(5.8)

The diagonal elements in matrix K of equation 5.7 and 5.8 are found using equation 5.9

$$K_{i,i} = \frac{(f_0^2 - f_i^2)}{(\Delta f \cdot f_i)}$$
(5.9)

The parameters $(f_0 = 2GHz)$ and $(\Delta f = 900MHz)$ are considered for the central frequency and bandwidth of the resonator, respectively; f_i is the resonant frequency of the i^{th} resonator where the lower band $(f_1 = 1.6GHz)$ and upper band $f_3 =$





Figure 5.5: Flowchart of the design procedure for FBSSIR

the location of the transmission zeros in the upper and lower stopbands, respectively. The transmission zeros are affected by the width of the butterfly stub and T-type stub. The core frequency and the position of two transmission zeros that affect the bandwidth are primarily determined by the L_1 and L_2 lengths. As the butterfly stub is employed, L_1 consists of the angle suspended by the circular sector (60⁰), the breadth (2mm), and the circular sector's outer radius (5mm). L_2 is 20.7mm in length. The variables mentioned here have been optimised, and the flow chart



depicted in Figure 5.5 illustrates the design approach. Figure 5.6 depicts the intended

Figure 5.6: The surface current distribution at (a) 1.6GHz (b) 2GHz and (c) 2.58GHz

FBSSIR's simulated surface current distribution pattern for the three modes. The resonator's surface current density is approximated at 1.6GHz, where current begins to flow from input to output port, while at 2GHz, the signal flows exclusively through the SIR structure, as illustrated in Figure 5.6(a) and 5.6(b). While the maximal current density is on the Butterfly stub at 2.6GHz, this indicates that the signal is not reaching the output because it controls the transmission zeros, hence controlling the upper stopband, as illustrated in Figure 5.6(c).

The SIR is designed at a center frequency of 2GHz using the fundamentals and formula in the preceding subsections, and a folded T-stub and butterfly stub are embedded in the plane. The proposed FBSSIR's performance is evaluated using the insertion loss plot given in Figure 5.7. As shown in Figure 5.7, the stub height (j) is changed to obtain the optimal value. Below 4mm, there is a significant loss in signal transmission in the upper sideband, while there is no change above 4mm; thus, the optimal value for height (j) in FBSSIR is 4mm. Additionally, Figure 5.8 and Figure 5.9 exhibit plots of the S-parameters. After varying L_2 between 18.7mm and 21.7mm, it



Figure 5.7: S-parameters analysis by varying the stub height (j)



Figure 5.8: S-parameters analysis by varying the folded T-stub length (L2)

is obvious that the optimal value is 20.7mm, as L_2 less than the optimal value has less bandwidth, influencing both the lower and higher bands. And L_2 above the chosen value produces a response that is close to the optimum value response. Similarly, L_1 is altered between 10 and 13 mm, and their S_{11} and S_{21} responses are illustrated in Figure 5.9.



Figure 5.9: S-parameters analysis by Varying the butterfly stub length (L1)

Interestingly, 10mm produces the best S_{21} reaction and the worst S_{11} response, whereas 13mm produces the best S_{11} response and the worst S_{21} response. As a result, the optimal value of 11mm is chosen to meet our bandwidth need. The proposed FBSSIR is fabricated on a double-layer copper printed circuit board with a ϵ =4.6. The Smith chart is used to plot the resonator's input and output impedances, as seen in Figure 5.10. It is worth noting that the curve begins at the purely actual frequency of 1.64GHz and ends at another real frequency of 2GHz, which is the capacitive region's centre frequency. The curve then shifts to the inductive area between 2GHz and 2.5GHz. Figure 5.11 illustrates the measured and simulated values of the Sparameters. Due to the symmetry of the resonator, $S_{11} = S_{22}$ and $S_{21} = S_{12}$. From 1.6GHz to 2.55GHz, the return loss is substantially below -10dB, covering a nearly 1GHz wide bandwidth with a center frequency of 2GHz. The simulated and measured values agree well in terms of return loss, although there is a 3 to 4dB difference in S_{21} owing to manufacturing wear and tear. The constructed FBSSIR has a surface area of $37 \times 19mm^2$. The FBSSIR is compared to other published works in Table 5.1. This



Figure 5.10: Smith chart plot of the FBSSIR



Figure 5.11: Simulated and Measured S-parameters of FBSSIR

work achieves the highest fractional bandwidth (FBW=50%). The equation 5.10 are used to compute the Fractional Bandwidth:

$$FBW\% = \frac{f_2 - f_1}{f_c} \times 100 \tag{5.10}$$

Where f_2 is the upper cutoff frequency, f_1 is the lower cutoff frequency and f_c is the center frequency. Here we have $f_2 = 2.58$ GHz, $f_1 = 1.6$ GHz and $f_c = 2$ GHz which results in FBW% of about to be 50 in our case.

Parameters	Central freq	FBW%	Insertion	Return	Area
	(GHz)		loss(dB)	loss(dB)	(mm^2)
Chu and Chen 2008	2.13/4.87	14/11.2	0.73/0.9	32/33	23.1×23.1
Hou <i>et al.</i> 2020	2.38/4.87	6.7/8	2/1.4	—	_
Luo <i>et al.</i> 2010	1.84/2.9	8.1/6.8	1.7/1.6	> 10	_
Zhang et al. 2007	3.4/6.7	4.8/4	3.8/3.7	18.7/14.6	-
Hagag et al. 2019	2.4	16.6	2	> 10	_
This Work	2	50	2.5	> 15	37×19

Table 5.1: Comparison of state of art with resonators

5.4 Design and consideration of LNA integrated with FBSSIR

This section discusses the design and analysis of an integrated LNA based on FB-SSIR. In general, the function of a resonator is to operate as a filter and to provide impedance matching, but the purpose of integrating FBSSIR into an LNA is to investigate a novel impedance matching design and to desensitise the LNA's passive components. The FBSSIR is a circuit that serves as an input and output matching network as well as a filter for this core LNA circuit. The transistor utilised in this application is the BFP420, which is a low-noise device based on a Siemens grounded emitter transistor (SIEGET). This device was chosen due to its 1.1dB minimum noise figure (NF_{min}) , 21dB transconductance (G_m) , and 25GHz transition frequency (f_T) . The LNA is designed to operate at a normal V_{DD} of 3.3V and consumes very little power. The collector current (I_C) is limited to a maximum of 20mA. We employ a two-stage cascade LNA with an emitter grounding topology. Passive components

such as resistors R_1 to R_4 and inductors L_1 and L_3 are used to create the major bias networks for each stage. Both the input and output are capacitive coupled (i.e., C_1 and C_7) to prevent the adjacent circuitry from being affected by DC. Figure 5.12a illustrates the suggested schematic for the FBSSIR integrated LNA circuit. The initial stage is designed to have the lowest feasible noise figure because it establishes the circuit's sensitivity and noise figure. The base DC is supplied via the inductor L_1 , which increases the LNA's Third order harmonics (IP_3) performance. Prior to cascading the first stage to the second, it is prudent to establish an interstage network to ensure maximum power transmission between the first and second stages. Rather than employing a conventional LC network as an interstage, this research proposes a novel interstage matching network based on stub matching. To improve stub matching, we employ butterfly stub B_1 with a width of feed line W=0.5mm, an outer radius of circular sector $R_o = 1.63$ mm, and an angle of 78°. The $B_1 - C_4$ network is added to match the first and second stages' output and input impedances. The second stage optimises IP_3 and gain via R_4 and L_4 . Additionally, it compensates for the gain loss caused by the inclusion of FBSSIR. The capacitors C_3 and C_2 in the first stage and C_5 and C_6 in the second stage give the AC grounding. The complete circuit is built and analysed

Comp	Value	Comp	Value	Comp	Value
R_1	$5.6~{ m K}\Omega$	R_2	$240~\Omega$	R_3	$6.8 \text{ K}\Omega$
R_4	190 Ω	L_1	33 nH	L_2	$43 \mathrm{~nH}$
L_3	$24 \mathrm{nH}$	L_4	56 nH	C_{1-7}	$100 \mathrm{pF}$

Table 5.2: Optimized component values of the proposed LNA circuit

for the sensitivity of the passive components utilised in the LNA design, and lastly, using DoE and statistical analysis, it is optimised for the optimal performance. Table 5.2 contains the optimal values for the LNA components. Figure 5.12b depicts the isometric representation of the three-dimensional layout. The layout is created with packaging components and assessed using the ADS and the Momentum microwave. v2019. The three-dimensional layout view reveals that the top and bottom layers are constructed of copper, and that vias connect the top and bottom layers throughout the circuit. The vias carry the surplus signal density to the ground bottom layer. Two types of vias are used: one with a drill diameter of 0.2mm and a pad diameter



Figure 5.12: Proposed FBSSIR based LNA (a) Schematic (b) 3D layout

of 0.3mm, and another with a drill diameter of 0.5mm and a pad diameter of 1mm. It's worth noting that the vias are more densely packed around the transistors to help

dissipate heat and offer a path for grounding. Due to the thinness of the signal path, 0.2mm diameter vias are used in and around the transistor. In other regions, vias with a diameter of 0.5mm are employed. The small-signal equivalent high-frequency



Figure 5.13: Small signal equivalent circuit of the proposed FBSSIR based LNA

model of the FBSSIR LNA is shown in Figure 5.13. The input impedance $Z_{in-Total}$ of the circuit can be calculated using equation 5.11.

$$Z_{in-Total} = Z_{FBSSIR} + Z_{BJT-LNA} \tag{5.11}$$

The input impedance of FBSSIR is similar to the SIR as the Folded butterfly stub only alters the location of zeros which is βl and is given by the equation 5.5.

Using the High-frequency small-signal model of BJT $Z_{BJT-LNA}$ is calculated. Here r_b is base spreading resistance and represents the resistance of the base-emitter junction. r_{π} represents the dynamic resistance for small signal analysis. r_e represents the parasitic emitter resistance. C_{μ} is the depletion capacitance of the collector base junction. C_{π} is the capacitance of the base emitter junction. $g_m v_{\pi}$ is the amplification factor.

$$Z_{BJT-LNA} = \frac{v_i}{i_b} \tag{5.12}$$

$$v_i = r_b . i_b + \left[\frac{1}{S.C_M} \|r_{\pi}\right] . i_b + i_e . r_e$$
 (5.13)

Where $C_M = C_{\pi} + C_{\mu}$ is the equivalent parasitic capacitance including the Miller

effect and $i_e = i_b \left[1 + g_m \left[\frac{1}{S.C_M} \| r_\pi \right] \right]$ By substituting this in equation 5.13

$$\frac{v_i}{i_b} = r_b + \left[\frac{1}{S.C_M} \|r_\pi\right] + \left[1 + g_m \left[\frac{1}{S.C_M} \|r_\pi\right]\right] \cdot r_e \tag{5.14}$$

Simplifying the equation 5.14

$$Z_{BJT-LNA} = r_e + r_b + \frac{r_{\pi} \cdot [1 + r_e g_m]}{1 + SC_M \cdot r_{\pi}}$$
(5.15)

By substituting equation 5.5 and equation 5.15 in equation 5.11, we get the total input impedance of FBSSIR LNA given by the equation 5.16 below.

$$Z_{in-Total} = \left| Z_0 \frac{Z_L + j Z_0 \tan \beta l}{Z_0 + j Z_L \tan \beta l} \right| + r_e + r_b + \frac{r_\pi \cdot [1 + r_e g_m]}{1 + S C_M \cdot r_\pi}$$
(5.16)

5.4.1 DoE Analysis

The Design of Experiments (DoE) is examined in order to produce a robust design and to comprehend the contribution of the design variable to accomplishing the final goal specifications as discussed in section 2.9. Only passive components are taken into account while determining the design variable. The DoE analysis is performed on the LNA core circuit with and without the FBSSIR to demonstrate the FBSSIR's potential. The results of the comparison of design variables versus contribution to the specific aim, such as NF, Gain, and return loss, are given in Figure 5.14(a), (b), and (c). It is worth noting that when there is no FBSSIR, R_2 and R_4 contribute virtually entirely to the gain. By integrating FBSSIR, the gain sensitivity is lowered by 30% from R_2 . The gain sensitivity reduction caused by R_2 is shared by other passives such as L_2 to L_4 , R_1 , and R_3 . Generally, sensitivity to a single component value is not recommended because it has a detrimental effect on the circuit's performance.

Similarly, it is noted that R_2 provides about 100% of NF in the absence of FBSSIR. However, when the FBSSIR is added, the NF's singular dependent on R_2 becomes desensitised and spreads to the other passives. The same holds true for the return loss depicted in Figure 5.14(c). The point of observation is that the FBSSIR is used for both input and output filtering; the sensitivity of the passive components is desensitised, resulting in consistent output results.



Figure 5.14: Design variable contribution to (a) Gain (b) NF (c) Return Loss

5.4.2 Statistical Analysis

Due to the fact that device parameters are vulnerable to fluctuation due to Process Voltage and Temperature (PVT), yield analysis is critical for calculating the real yield and confidence levels as discussed in section 2.8. The proposed FBSSIR integrated LNA is evaluated for error tolerance of 5% and 1% under Gaussian distribution. Gain, NF, and return loss values are depicted in Figure 5.15 (a), (b), and (c), respectively. The stable gain of 22dB, the NF of 2.8dB, and the return loss of 12dB are all achieved with a tolerance of 1%. It yields a yield estimate of 100% after 1000 iterations. However, for the 5% tolerance yield estimation, 32% of iterations failed to yield 68%. The parameters are limited to a minimum gain of 20dB, a maximum NF of 3.5dB, and a minimum return loss of 10dB. However, PVT changes are often less than 1%,

indicating that the planned FBSSIR-LNA is stable and robust.



Figure 5.15: Yield estimation of FBSSIR-LNA (a) Gain (b) NF (c) Return Loss

5.5 Results and Discussion

This section discusses the performance of the FBSSIR integrated LNA by presenting simulation and measurement findings. The circuit is compared before and after the addition of the FBSSIR to demonstrate the potential and importance of the FBSSIR as a filtering step in the input and output to evaluate the LNA's performance. The LNA is implemented and constructed utilising PCB packaging techniques on a FR4 substrate. Figure 5.16a and figure 5.16b illustrate the top view of the manufactured LNA and FBSSIR integrated LNA, respectively. S-parameters for the FBSSIR integrated LNA



(b)

Figure 5.16: Fabricated photograph view with (a) LNA prototype (b) FBSSIR integrated LNA prototype

are determined using a PNA E8363 network analyzer connected to Agilent technologies RF probes up to 40GHz. Prior to performing measurements, the PNA and probes must be calibrated. For probe systems, the approach and method of short-open-load-through calibration have been used. Figure 5.17 shows the simulated and measured NF plots with and without FBSSIR. The NF of the LNA decreases by 7.5 dB in the FBSSIR passband, which ranges from 10.2dB to 2.7dB at 1.8GHz. Outside of the 1.6GHz to 2.5GHz frequency range, even the NF of the circuit with FBSSIR increases by 10dB. Due to the correct impedance matching of the components, the FBSSIR integrated LNA significantly reduces the NF from 10dB to 2.5dB. The antenna signal is routed through the FBSSIR and then to the LNA. By filtering, the FBSSIR eliminates noise and provides a pure signal to the first stage of the LNA. A modest decrease in



Figure 5.17: Noise Figure of the designed FBSSIR-LNA



Figure 5.18: Gain of the designed FBSSIR-LNA

signal strength is detected, lowering the gain of the circuit by 4dB. However, the second stage LNA circuit may easily compensate for this increase. The output side

FBSSIR filters out noise generated by the device's passives and actives.

Figure 5.18 illustrates the gain comparison with and without FBSSIR. It is worth noting that, while the gain decreases slightly when FBSSIR is used, the slope of the gain decay from 1.8GHz to 2.5GHz is slightly reduced.



Figure 5.19: Simulated Comparison of Pin vs. IP3 and Pout (fundamental) of LNA with and without FBSSIR

Figure 5.19 plots the simulated input and output power (fundamental at 1.8 GHz) and input and output power (third-order harmonics) for the designed LNA with and without FBSSIR to determine the IIP_3 and IP_{1dB} . It is worth noting that for the circuit with FBSSIR, the Input 1 dB (IP_{1dB}) compression point is obtained at -23dBm. At 0dBm, the third-order input intercept point (IIP_3) is reached. Additionally, the plot demonstrates that the developed LNA achieves good OIP_3 and OP_{1dB} values of +25dBm and +2.3dBm, respectively. Third-order harmonics are decreased slightly when FBSSIR is used, resulting in a +5dBm improvement of IIP_3 from -5dBm when FBSSIR is not used to 0dBm when FBSSIR is used. This LNA's strong linearity results in an outstanding spur-free dynamic range, which is ideal for wireless receiver construction.

The input and output impedances of the FBSSIR integrated LNA are depicted in Figure 5.20a and Figure 5.20b, respectively, as a smith chart. Around the resonant



Figure 5.20: Simulated smith chart plot of FBSSIR integrated LNA (a) Input impedance (b) Output impedance

frequency of 2GHz, both the input and output impedances align with the real axis. They exhibit identical half capacitive and half inductance curves as a result of the FBSSIR's perfect impedance matching at the input and output. The measured stability factor K and the Δ described in equation 3.5 and equation 3.6 is plotted in the



Figure 5.21: Measured Stability factor K and Δ of FBSSIR-LNA

figure 5.21. It is worth noting that the developed LNA is perfectly stable over the necessary frequency range of 1.6GHz to 2.5GHz.



Figure 5.22: Simulated and measured S-parameters of FBSSIR-LNA

Figure 5.22 depicts the measured and simulated entire S-parameters of LNA. The gain is greater than 22dB over the whole frequency range of 1.6GHz to 2.5GHz. The disparity between measured and simulated values is primarily due to fabrication wear and tear, namely soldering. In the 1.6GHz to 2.5GHz frequency band, the return loss (S_{11}) is sufficiently low (far below 12dB) to avoid reflection. Other S-parameters, such as S_{12} and S_{22} , also gives satisfactory performance. The output of the LNA with and without FBSSIR is shown in Table 5.3. The state of art comparison with reported works is shown in Table 5.4.

Parameters	Without FBSSIR LNA	With FBSSIR LNA
Freq(GHz)	1.6-2.5	1.6-2.5
$\operatorname{Gain}(\mathrm{dB})$	30	16
$S_{11}(\mathrm{dB})$	<-2	<-12
NF(dB)	10	2.7
$IP_{1dB}(dBm)$	-24	-23
$IIP_3(dBm)$	-5	0
$OIP_3(dBm)$	+25	+25
$OP_{1dB}(dBm)$	+2.9	+2.3
$\operatorname{Area}(cm^2)$	3.3×1.6	10.7×2

Table 5.3: Results summary of FBSSIR-LNA

5.6 Summary

The FBSSIR integrated LNA in PCB technology with a two-stage cascaded topology is provided in this chapter. The statistical and DOE analyses are presented to evaluate the design's yield and resilience. To begin, the suggested FBSSIR is constructed and validated using full-wave electromagnetic simulations. It functioned as an excellent noise filter and desensitised the LNA passive settings. This FBSSIR integrated LNA achieves gain in excess of 22dB throughout the 1.6-2.5GHz frequency range. It achieves a low NF of 2.7dB at 1.8GHz and fluctuates between 3-4.5dB between 1.6GHz and 2.5GHz. The FBSSIR integrated LNA is very linear, with an IP_{1dB} of -23dBm, an IIP_3 of 0dBm, an OP_{1dB} of +2.3dBm, and an OIP_3 of +25dBm, and a power dissipation of 26.5mW.

Parameters	Frequency (GHz)	Gain (dB)	NF (dB)	$IIP_3^*(dBm)$	Area (cm^2)	$P_{DC}(\mathrm{mW})$	FoM
Yang $et \ al. (2005)$	2-2.5	12	1.7	I	2.4×1.8	30	0.28
Hagag et al. (2019)	2.8-4	14	4.6	11	8×3	06	0.05
Djoumessi and Wu (2009)	2.2-2.5	22	1.5	-10	3×3	7.5	1.7
Kumar and Pathak (2014)	2.3-2.5	12	0.7	I	6×5.5	42	0.1
Iyer and Pathak (2014)	2.3-2.5	7	4.4	I	8×6	35.1	0.01
Kumar and Pathak (2015)	2.4	11	4	I	12×3.4	56	0.01
Li et al. (2020)	0.3-6	16.5	3.1	-4	0.06×0.04	48	0.83
Chaghaei <i>et al.</i> (2020)	0.1-2	21	4.9	-9.4	0.02×0.02	1.8	5.8
This Work	1.6-2.5	27	2.7	0	10.7×2	26.5	0.4

art.
of
state
with
performance
Y
Ą
Ę
FBSSIR
$_{\mathrm{the}}$
of
Comparison
<u> </u>
5.4
Table

- Not calculated * Simulated Results

Chapter 6

SWITCHLESS RECONFIGURABLE LNA BASED ON SSCL FOR K/KA BAND

6.1 Introduction

Applications in the K-Ka band (18-40GHz), including as 5th generation (5G) networking, radar, and astronomy detection, primarily rely on wideband mm-wave front-ends. Low noise amplifiers (LNAs), as one of the most crucial active components, are the primary block of mm-wave receiver front-ends. We intend to obtain a low noise figure, a high gain with flatness, good linearity, and low power consumption for LNAs in order to enhance the dynamic range and sensitivity of the receiver system, which complicates LNA design. This work aims on achieving the best trade-offs among the LNA performance parameters by comparing the reported works in the entire bandwidth of 22GHz.

6.2 Prior Works

Extensive research on wideband LNAs has been conducted in recent decades. A distributed amplifier (DA) is the most promising topology for realising an ultra-wideband LNA (Liang et al. 2021) and (Fang and Buckwalter 2019). However, DA is not ideal in many cases because to its poor bandwidth at unity gain, high NF, and greater area. Other strategies for increasing bandwidth have been reported in (Hu et al. 2018), (Hu et al. 2019), (Nikandish and Medi 2015) and (Park et al. 2021) including shunt-resistive feedback, dual inductive peaking, enhancing dynamic load, a frequencydependent feedback loop (FDFL), and a split common gate transistor (SCGT). These proposed strategies for increasing bandwidth decrease other RF factors such as noise performance. The primary disadvantage of wideband LNAs is that they amplify the unwanted signal while reducing the receiver's sensitivity. Another method is to use reconfigurable LNAs to construct wideband receivers with high interference rejection, and numerous multiband LNAs are described in the literature such as (Yu and Neihart 2013). A reconfigurable multimode LNA is proposed in (Shaheen *et al.* 2021), but it has a lower gain of around 12dB and a high NF of up to 5dB due to the inclusion of a switched multitap transformer in the input matching network and a tunable load based on electrical and magnetic tuning operating at three single bands (24/28/39)GHz).

(Nawaz et al. 2020) describes a 28/60 GHz reconfigurable LNA based on RF switches for mm-wave transceivers. Additionally, this LNA has a meager gain of 14dB and an NF of 4.6dB. They still have a 1dB ON-loss and a 0.7dB OFF-loss despite employing fine sophisticated RF switches. Although (Lee and Nguyen 2021) presents a 44/60 GHz reconfigurable LNA based on Q-enhanced metamaterial transmission lines, they have a high NF of 5.6dB and IIP_3 of -14dBm. The article's distinctive feature is that it achieves a 30dB rejection between two bands. Although it achieves a respectable NF of 2.8dB and low power consumption, a dual-band switchable LNA based on LC tunable tank and tunable stub based on Hetero-junction Bipolar Transistor (HBT) switch is published in (Nawaz et al. 2019). However, it suffers from low gain. (Liang et al. 2020) describes a tri-band reconfigurable LNA working at 24/33/50 GHz that utilises a triple coupling transformer (TCT) and extra gain boosting to reach gain of up to 32dB with a 6dB minimum noise figure. This LNA has a high NF value and low linearity. All of these published works operate at mm-wave frequencies and have their own design trade-offs.

(Wang and Zhang 2020) describes a two-stage LNA operating between 22GHz and 47GHz with a peak gain of 22.2dB and a noise figure of 4.3dB employing linked L-type interstage matching inductors. This LNA offers one of the best design trade-

offs available, with the drawback of a low IIP_3 . (Lee and Nguyen 2018) describes a dual-band concurrent single-ended input to differential output LNA working at 18-25 GHz and 33-46 GHz with a gain of 16dB and a noise figure of 4.3dB. While this LNA has a low gain and a high NF, it also has the unique ability to manage stop band rejection and passband gain balancing concurrently. (Yu et al. 2021) describes a 21-41 GHz, 28.5dB gain, and 3.2dB NF LNA using the triple coupled transformer approach. Additionally, this LNA offers one of the best design trade-offs; nonetheless, no other LNA has achieved ultra-low NF while maintaining good linearity. (Li et al. 2018) describes an LNA with a gain of 12.8dB and an NF of 1.4dB working between 14GHz and 31GHz. Among the works presented, this is the only LNA with an NF less than 1.5dB yet a gain less than 13dB. (El-Aassar and Rebeiz 2020) presents a 19.5-28.5 GHz variable low power LNA with a gain of 7.8-23.2 dB and a 0.4-1.6 V supply; nevertheless, they have not achieved ultra-low NF. The literature reviewed above is suitable for wideband mm-wave receiver systems, but they all have limitations in terms of LNA design trade-offs in terms of gain, NF, linearity, power consumption, area, and sensitivity. All of the reported works except (Lee and Nguyen 2021) use one or more components such as tunable inductors, varactors, transformers, and RF switches, which require additional power supplies and are difficult to implement at high frequencies due to their associated loss and tradeoffs.

Considering the drawbacks of the bandwidth enhancement and other reconfigurable techniques for the ultra-wideband LNA, this research proposes a new switchless reconfigurable LNA based on the inter-stage and output-stage Suspended Substrate Coupled Line (SSCL) for the full K-Ka band (18-40) GHz. An 8-20 GHz switchless dual-band reconfigurable LNA and a 6-18 GHz switchless dual-band reconfigurable power amplifier (PA) are reported in (Xie *et al.* 2020) and (Choi *et al.* 2018) using a coupled line-based diplexer which is similar to the SSCL used in this research.

6.3 SSCL design and consideration

6.3.1 Coupled line Analysis

A coupled line (CL) is formed when a two-transmission line is placed in such proximity that energy from one line transfer to the other. This proximity will alter the electromagnetic fields (so the voltages and currents alter) of the propagating wave, thereby altering the transmission line's characteristic impedance. When the two-transmission line is placed as coupled lines it forms a four-port network. Figure 6.1 illustrates the



Figure 6.1: Conventional coupled line structure (A)



Figure 6.2: Even-Odd mode Excitation in CL

traditional microstrip coupled line. Coupled line working principles are the same even if we change the type of the transmission line used like suspended substrate line (SSL), strip-line, shielded microstrip line, Fin-line, and so on. So, the traditional microstrip coupled line analysis is discussed to better understand the proposed SSCL's dual-band activity mechanism. Port 1 is the input port, the output which comes directly from port 1 is through port or port 2. The output port where the coupled signal comes out is known as coupled port or port 3. The final port is known as the isolated port or port 4 where ideally no signal comes out from input. In a coupled line structure, the load impedance Z_{load} is considered to be the same as the coupled line's characteristic impedance Z_0 . In this case, the output voltage of the coupled port (port 3) and through port (port 2) can be determined as a function of the input voltage V_{in} denoted in equations 6.1 and 6.2 (Pozar 2011).

$$V_{\text{through}} = V_{\text{in}} \cdot \frac{\sqrt{1 - K^2}}{\sqrt{1 - K^2}\cos\theta + j\sin\theta} = T \cdot V_{\text{in}}$$
(6.1)

$$V_{\text{coupled}} = V_{\text{in}} \cdot \frac{jK\sin\theta}{\sqrt{1-K^2}\cos\theta + j\sin\theta} = C_F \cdot V_{\text{in}}$$
(6.2)

Where $T = \frac{\sqrt{1-K^2}}{\sqrt{1-K^2}\cos\theta + j\sin\theta}$ and $C_F = \frac{jK\sin\theta}{\sqrt{1-K^2}\cos\theta + j\sin\theta}$

Voltage coupling factor C_F of port 3 as a function of the coupling coefficient, K and electrical length θ can be given by:

$$20\log|C_F| = 20\log\left|\frac{jK\sin\theta}{\sqrt{1-K^2\cos\theta} + j\sin\theta}\right|$$
(6.3)

Here K is the coupled line's coupling coefficient, and θ is the coupled line's electrical length.

There are two modes for the coupled line depending on the current flow mechanism in an electromagnetic context. In the first mode, current flows downward on the conductor with a contra-flow current back up the other conductor due to displacement current coupling between the two conductors. This is referred to as the odd mode current, and it has an odd mode characteristic impedance Z_{0o} associated with it. In the other mode, the current flowing between each center conductor through displacement current has the same polarity and is connected to a common ground. As a result, this current is referred to as even mode current, and it is accompanied by an even mode characteristic impedance Z_{0e} (Pozar 2011). The pictorial representation of the odd mode and even mode is depicted in Figure 6.2. The coupling coefficient K as a function of odd mode impedance Z_{0o} and even mode impedance Z_{0e} can be given by:

$$K = \frac{(Z_{0e} - Z_{0o})}{(Z_{0e} + Z_{0o})} \tag{6.4}$$

The relation between the characteristic impedance Z_0 and odd and even mode impedance is given as:

$$Z_0 = \sqrt{Z_{0e} Z_{0o}}$$
(6.5)

From equations 6.1 and 6.2, we can see that at very low frequencies or terse line length ($\theta \ll \pi/2$), nearly all input power is transmitted to port 2, with none being coupled to port 3. The maximum coupled power transmitted to port 3 is achieved at $\theta = \pi/2$ or $l = \lambda/4$, as shown in Figure 6.3.



Figure 6.3: Coupled power of conventional coupled line A

From Figure 6.3, the maximum coupled power is approximately -3dB with a longcoupled line ($\theta = \pi/2$) for K = 0.7. For this reason, the traditional coupled line structure in Figure 6.1 may not be applicable in the proposed LNA. The scattering matrix [S] of a four-port coupled-line coupler can be written as:

$$[S]_{4\times4} = \begin{bmatrix} 0 & -j\sqrt{1-K^2} & K & 0\\ -j\sqrt{1-K^2} & 0 & 0 & K\\ K & 0 & 0 & -j\sqrt{1-K^2}\\ 0 & K & -j\sqrt{1-K^2} & 0 \end{bmatrix}$$
(6.6)

The modified structure of the coupled line is shown in Figure 6.4. The isolated port (port 4) is grounded initially (without C_8). The shunt capacitance C_7 at port



Figure 6.4: Proposed Coupled line structure B

2 (through port) behaves as a short circuit for ideal high-band frequencies operation. The input impedance matching network of the low-band LNA absorbs this C_7 during the low-band frequency operation. The through port output signal is mirrored with an 180⁰ phase shift and re-enters the coupled line. The signal is then coupled to the isolated port, mirrored with an 180⁰ phase shift, and finally emerges from the coupled port (port 3) with the initial coupled signal. The output voltage of the isolated port (port 4) and coupled port (port 3) as a function of the input voltage V_{in} can be expressed as:

$$V_4^- = C_F \cdot V_2^+ = C_F \cdot \Gamma \cdot V_2^- = C_F \cdot \Gamma \cdot T \cdot V_{\text{in}}$$

$$(6.7)$$

$$V_{3}^{-} = C_{F} \cdot V_{\text{in}} + T \cdot V_{4}^{+} = C_{F} \cdot V_{\text{in}} + T \cdot \Gamma \cdot V_{4}^{-}$$
(6.8)

On simplifying the above equation

$$V_3^- = C_F \cdot V_{\rm in} + C_F \cdot T^2 \cdot V_{\rm in} \tag{6.9}$$

Coupled power of port 3 as a function of the coupling coefficient K and electrical length θ can be expressed as:

$$20\log\left|\frac{V_3^-}{V_{in}}\right| = 20\log\left|C_F + C_F \cdot T^2\right|$$
(6.10)

It is noted that from equation 6.9, the coupled port output signal consists of two

parts: $C_F V_{in}$ and $C_F T^2 V_{in}$. The phase difference between the two parts is identical to the coupled line's double electrical length, as shown in Figure 6.5. When the



Figure 6.5: Coupled power of proposed coupled line B without capacitor C_8

coupled line's electrical length is $\lambda/4$, the phase difference between the two parts of the coupled power is 180⁰; then, the minimum coupled power is obtained. Figure 6.5 and Figure 6.6 illustrate the coupled power in port 3 of the coupled line shown in Figure 6.4 according to different coupling coefficient K and electrical length (θ) of the coupled line. When the electrical length is $30^{0} \sim 45^{0}$ or $135^{0} \sim 150^{0}$, maximum coupled power can be obtained for a definite K. Here, a maximum coupled power of -4dB can only be achieved for K= 0.7, which is not sufficient to be used in the interstage of the LNA. The phase difference between the two parts of the coupled power $C_F.V_{in}$ and $C_F.T^2.V_{in}$ is the cause of inadequate coupled power, and it is shown in Figure 6.7.

To compensate for the phase difference between two parts of the coupled power, we added a shunt capacitance C_8 to the isolated port. The phase response after adding C_8 is shown in Figure 6.8. From Figure 6.7 and Figure 6.8, without the capacitor C_8 , there is 180⁰ phase difference when the electrical length is 90⁰ or $\lambda/4$, causing minimum coupling, but after adding shunt capacitance C_8 to the isolated port, the phase difference becomes 0⁰ at $\lambda/4$ electrical length which maximizes the coupling.



Figure 6.6: Coupled power of proposed coupled line B according to different coupling coefficients without capacitor C_8



Figure 6.7: Phase difference of proposed coupled line B without capacitor C_8



Figure 6.8: Phase difference of proposed coupled line B with capacitor C_8

6.3.2 Comparison of basic coupled line structure with proposed SSCL

The proposed coupled line is first tested with the traditional microstrip line (MCL) and Shielded microstrip line (SMCL) shown in Figure 6.9(a) as an inter-stage. The schematic of the proposed structure shown in Figure 6.10(a) is simulated, and the simulation results in Figure 6.11 show that the coupling power suffers a loss of around -4.8dB because of the described reason in the above subsection 6.3.1. The losses occur due to dispersion, fringing effects, and other EM losses due to mm-waves. In Chapter 3 section 3.3.1 a wafer-level integrated suspended substrate line (SSL) platform is developed to avoid these losses in our previous work. Based on the SSL developed the coupled lines is developed named as Suspended Substrate Coupled Lines (SSCL). The 1-D and 3-D schematics of the proposed SSCL are shown in Figure 6.9(b) and Figure 6.9(c). It consists of two substrates made of Silicon Nitride (SiN) with a thickness of 25μ m marked as M1, M2, M3, and M4. The metal layers M1 and M4 act as the ground, while the metal layer M3 act as a signal trace.



Figure 6.9: (a) 1-D Microstrip Coupled Line (MCL) (b) 1-D Suspended Substrate Coupled Line (SSCL) (c) 3-D geometry of SSCL

The metal layer M2 act as a signal reflection path. When the signal passes through the metal layer M3 at such high mm-wave bands, an EM field is developed, which causes a fringing effect around the M3 layer. Additional losses such as dispersion loss, dielectric loss also occur. But here, as the signal trace path has a reflection path (M2) and ground layers (M1, M4), the developed EM field is confined to the air cavity as discussed in chapter 3 section 3.3.1. This allows us to have the minimum loss, and we receive high coupled power at the output.

The schematic of the proposed SSCL as interstage and output-stage are shown in Figure 6.10(a) and (b), respectively. The length (L) of the coupled line is 193 μ m with $30^{0} \sim 40^{0}$ electrical lengths over 18-40 GHz. Width (W) and separation distance (S) is 10 μ m and 4.93 μ m, respectively. The coupling coefficient (K) is 0.72 approximately. Shunt capacitance of $C_{7} = 0.01$ pF is added to the through-port, and Shunt capacitance



Figure 6.10: Schematic of (a) Inter-stage MCL/SMCL/SSCL (b) Output-stage SSCL



Figure 6.11: Simulated S-parameters and phase of MCL/SMCL as interstage

of $C_8 = 0.24 \text{pF}$ is added to the isolation port to increase coupled power. From the simulation results, as shown in Figure 6.12, over 25-40 GHz, the coupled power is approximately -1.5dB which is about 3dB higher than the traditional MCL as shown



Figure 6.12: Simulated S-parameters and phase of SSCL as interstage



Figure 6.13: Simulated S-parameters and phase of SSCL as outputstage

in Figure 6.11. By this, the input broadband signal (18-40) GHz is successfully split into (18-24) GHz at the through-port and (25-40) GHz at the coupled port. The odd

mode characteristic impedance Z_{0o} and even mode characteristic impedance Z_{0e} are calculated to be 20.16 Ω and 127.34 Ω , respectively. Thus, the proposed SSCL can be applied in the reconfigurable LNA's interstage. Figure 6.10(b) is the schematic of the proposed output-stage SSCL, and the simulation result is shown in Figure 6.13. It is the symmetric structure of the interstage SSCL with optimized S, L, and Capacitors. Insertion loss is less than 3dB over 18-24 / 25-40 GHz. This stage combines the signal from the high-band and low-band stages and couples the power to the output as a broadband signal. The phase response of the couplers are also shown in the same Figure 6.11, 6.12 and 6.13 respectively.

6.4 Design and Analysis of CMOS Switchless LNA

The simplified block showing the proposed dual-band switchless reconfigurable LNA topology is shown in Figure 6.14. It consists of three amplifier stages: a broadband



Figure 6.14: Proposed reconfigurable LNA topology

driving stage, a high-band (Ka) stage, and a low-band (K) stage, as well as two coupled stages based on SSCL, referred to as an inter-stage SSCL and an outerstage SSCL. All three amplifier stages employ the common source (CS) topology. Transmission lines are utilised in place of spiral inductors in this design to conserve chip space and minimise other associated losses. To obtain a compact chip area, the lengthy transmission wires are bent. Instead of employing a normal quarter-wave long
transformer as the bias circuit for the proposed LNA, transmission lines and bypass capacitors are employed to behave like short circuits within the desired bandwidth and reduce chip size. Figure 6.15 depicts the proposed reconfigurable LNA in its



Figure 6.15: Schematic of the proposed reconfigurable LNA

whole. This reconfigurable Switchless LNA is capable of operating in three modes: dual-band, high-band (Ka), and low-band (K). V_{d3} is disabled for the Ka-band stage, V_{d2} is disabled for the K-band stage, and both are enabled for the dual-band stage. The following sections discuss the circuit architecture of the broadband drive stage and the high-band low-band stage.

6.4.1 Design of Broadband drive stage

As illustrated in Figure 6.15, the broadband drive stage is the first block in the proposed reconfigurable LNA topology. It is where the input RF signal is received. This stage is critical for the reconfigurable LNA's input matching and noise performance. It comprises of two cascaded CS stages with transistors M_1 and M_2 working at V_{g1} = 0.43V. Capacitors (C_1, C_2) and a transmission line of length L_1 and width W_1 comprise the input matching network through which the signal flows and is sent to the transistor M_1 's gate. M_1 and M_2 receive the drain and gate voltages through transmission lines of lengths L_3 , L_2 , and L_5 , L_4 . Source degeneration is incorporated through the use of the transmission lines L_{s1} and L_{s2} to M_1 and M_2 in order to get the ideal impedance for the lowest NF value near to the complex conjugate of the input impedance. This achieves both low NF and input impedance matching. The capacitor C_3 is employed to prevent direct current from flowing from the first to the second stage.



Figure 6.16: Simplified small-signal model of first CS stage for input impedance

The input impedance (Z_{in}) of the circuit, which depends mainly on this stage, can be calculated using the simplified small-signal model of the first CS stage is shown in Figure 6.16. Z_{OX} denotes the combined impedances of all the preceding stages. For simplicity, impedance in transmission lines such as L₂, L₁, and L₃ is considered inductance. In Figure 6.16 g_m denotes the transconductance of the transistor M₁, whose value is proportional to the width of the transistor, C_{gs} and C_{ds} denotes gatesource and drain-source capacitance, C_{Mi} and C_{Mo} denotes equivalent input and output Miller capacitance due to gate-drain capacitance $C_{gd}.C_{Mi} = C_{gd}(1 - A_v), C_{Mo} =$ $C_{gd}\left(1 - \frac{1}{A_v}\right), A_v$ denotes the gain of the corresponding stage. The input impedance of the circuit $\left(Z_{in} = \frac{V_{in}}{i_{in}}\right)$ is calculated as follows:

$$V_{in} = i_{in} \left[j\omega \left(L_2 + L_{S1} \right) + \frac{1}{j\omega C_{in}} \right] + g_m \cdot V_{gs} \cdot j\omega L_{S1}$$

$$(6.11)$$

Here $C_{in} = C_{gs} + C_{Mi}$ is the equivalent input capacitance at the gate terminal. Substituting $V_{gs} = i_{in}/j\omega \cdot C_{in}$ in equation 6.11 and simplifying gives

$$Z_{in} = \frac{V_{in}}{i_{in}} = \frac{g_m \cdot L_{S1}}{C_{in}} + j \left[\omega \left(L_2 + L_{S1} \right) - \frac{1}{\omega C_{in}} \right]$$
(6.12)

Similarly, the gain of the stage A_v can be calculated and shown in equation 6.13.

$$A_{V} = \frac{\frac{-g_{m}}{j\omega C_{in}} \cdot \left[j\omega L_{3} \| Z_{ox} \| \frac{1}{j\omega C_{out}} \right]}{j \left[\omega \left(L_{2} + L_{S1} \right) - \frac{1}{\omega C_{in}} \right] + \frac{g_{m} \cdot L_{S1}}{C_{in}}}$$
(6.13)

The simulated S-parameters of the broadband drive stage are shown in Figure 6.17. The gain (S_{21}) is above 20dB till 33GHz, and it falls to 16dB at 40GHz. The return loss (S_{11}) is well below -10dB from 18-40 GHz.



Figure 6.17: Simulated S-Parameters performance of the separate broadband drive stage.

6.4.2 Design of High band (Ka) and Low band (K) LNA stages

Simulated results for the broadband driving stage indicate a gain fluctuation of 7dB between 18-40 GHz. Though it is difficult to maintain almost constant gain across the whole bandwidth, the high-band and low-band stages, each of which consists of two transistors, are used to keep gain fluctuation below 2dB. The low-band stage is composed of transistors M_5 and M_6 in a two-stage cascaded configuration running at $V_{g3} = 0.4$ V. This stage is intended to operate exclusively in the K band, between 18 and 24 GHz. Figure 6.18 illustrates the low-band driving stage's simulated S-parameters. The gain (S_{21}) is 10dB to 8dB in the 18-24GHz passband, but drops to 2dB at 40GHz. From 18-24 GHz, the return loss (S_{11}) is well below -10dB, while strong reflections are observed above this range. The high-band stage is composed of transistors M_3 and



Figure 6.18: Simulated S-Parameters performance of the separate Low-band (K) stage.

 M_4 in a two-stage cascaded configuration running at $V_{g2} = 0.47$ V. This stage is meant to operate exclusively in the Ka-band, which covers the frequency range of 25-40 GHz. Figure 6.19 illustrates the simulated S-parameters of the high-band drive stage. Gain (S_{21}) is between 12dB and 10dB in the 25-40 GHz passband. From 25-40 GHz, the return loss (S_{11}) is significantly below -10dB, and high reflections are observed in



Figure 6.19: Simulated S-Parameters performance of the separate High-band (Ka) stage.

the 18-24 GHz range. To reduce instabilities caused by probable oscillation loops in the complicated structure of the multi-device amplifier or by the device's nonlinear nature, R-C resistive feedback is introduced at the transistor M_4 's gate and drain using (R_1, C_{10}) . It is not recommended to connect the coupled port (port 3) of the output stage SSCL directly to the 50 Ω output load. This is because the SSCL is generated by stacking layer by layer on the substrate, making a direct connection to the output load difficult. Additionally, the impedance of SSCL varies with frequency, which is primarily inductive. In very extreme circumstances, the impedance may be completely inductive, resulting in circuit instability. To circumvent them, we have connected an R-C ($R_2 - C_{17}$) circuit between the output stage SSCL and the output load. In the worst-case scenario, the coupled R-C network cancels out any abrupt change in inductive impedance.

6.4.3 Consolidation of all Stages

After developing and testing each stage individually, such as the SSCL interstage, the SSCL output stage, the broadband stage, and the high and low band stages. As

Band	$V_{d2}(V)$	$V_{d3}(V)$	Total Current(mA)
K (18-24) GHz	0	1.2	13.8
Ka (25-40) GHz	1.2	0	14.5
K and Ka $(18-40)$ GHz	1.2	1.2	21.4

illustrated in Figure 6.14, all stages are interconnected. The output of the broadband

Table 6.1: Drain Voltages states of single-band transistors.

drive stage is connected to port 1 of the SSCL interstage, where it is separated into two single bands designated as Ka-band and K-band. Based on the SSCL principle outlined in the preceding section. The Ka-band is coupled via SSCL's port 3, while the remaining K band is sent to the respective amplifier stages via port 2. Following amplification by the respective stages, the output stage SSCL is used to combine two single-band signals. As previously explained, SSCL at the output combines the signals from ports 2 and 3 and outputs the broadband signal from port 1. The total dc drain current and the drain voltage states of single-band transistors required to accomplish switch-free dual-band operation are shown in Table 6.1.

6.4.4 Noise Figure Analysis

To examine the whole LNA's noise figure, the following method is used to transform any noisy multi-port network into an equivalent noise-free network plus standard current and voltage noises at the input port. When all internal noise sources are represented as current noises, a general noisy multi-port can be explained as a hybrid of a noise-free multi-port and some current noises at the ports, as illustrated in appendix A-1

. The derivation of the noise factor F of the multi-port network is also depicted deeply in appendix $\hbox{\rm A-1}$

Figure 6.20 illustrates the simulated NF of the various broadband, high band, and low band stages. The broadband stage NF has a U-shaped response curve with a maximum of 1.15dB at 40 GHz. This demonstrates that the technique outlined previously optimised the NF to a value of only 1.15dB. The low band stage's NF is around 1.6dB in the K band and grows above this band. The NF of the high band stage is



Figure 6.20: Simulated Noise Figure performance of the broadband, high band, and low band stages

around 1.2dB in the Ka band and up to 7dB in the K band. The reason for having high NF outside the desired frequency range in the single band operation lies in the design and optimization of SSCL.

6.4.5 Statistical Analysis

Due to the fact that device parameters are vulnerable to fluctuation due to Process Voltage and Temperature (PVT), yield analysis is critical for calculating the real yield and confidence levels as discussed in section 2.8. The designed LNA is evaluated for 5% and 1% error tolerances under Gaussian distribution. NF, Gain, and return loss values are depicted in Figure 6.21 (a), (b), and (c), respectively. The stable gain of 25dB, the NF of 1.6dB, and the return loss of -11dB are all achieved with a tolerance of 1%.

It resulted in a yield estimation of 100%, with a confidence level of 99.7% for 1000 iterations. However, with the 5% tolerance yield estimation, 10% of iterations failed to yield 90%. The parameters are limited to a minimum gain of 20dB, a maximum NF of 2dB, and a minimum return loss of -10dB. However, CMOS process PVT variations



Figure 6.21: Yield estimation of LNA (a) Gain (b) NF. (c) Return Loss

are often less than 1%, indicating that the proposed reconfigurable switchless LNA is stable and accurate.

6.4.6 DoE Analysis

Design of Experiments (DOE) is a data-driven technique for a resilient design that elucidates the role of each design parameter in accomplishing a specific objective, such as a flat gain of 25dB, NF of 1.5dB with low return loss, and high linearity as discussed in section 2.9. The proportion contribution of each design parameter to the specified goal, such as NF, Gain, and return loss, is presented in Figure 6.22.

To validate the LNA at both the K and Ka bands, it is tested at 20GHz and 35GHz, as seen in Figure 6.22(a) and (b). With the exception of the percentage sensitivity,



Figure 6.22: Design parameter contribution at (a) Low-band at 20GHz (b) High-band at 35GHz

the identical parameter from the broadband drive stage contributes to design goals. Additionally, it is noted that the low-band stage parameter M_{5W} contributes a small gain at 20GHz, whereas the high-band stage parameter M_{3W} contributes a small gain at 35GHz. For the sake of clarity, the design factors that contribute less than 1% are omitted from Figure 6.22. Extra attention has been taken during fabrication to ensure that these sensitive and contributing aspects produce trustworthy, precise measurement findings. The optimized parameters of the designed switchless LNA is depicted in Table 6.2.

Comp	Value	Comp	Value	Comp	Value	Comp	Value
M_{1W}	$88 \ \mu m$	M_{2W}	$179~\mu{\rm m}$	M_{3W}	$36 \ \mu m$	M_{4W}	$24 \ \mu m$
M_{5W}	$192~\mu\mathrm{m}$	M_{6W}	184 $\mu {\rm m}$	W_1	$10~\mu{ m m}$	C_1	2.6 pF
C_2	$0.14 \mathrm{\ pF}$	C_3	1.6 pF	C_4	0.24 pF	C_5	$0.48 \mathrm{\ pF}$
C_6	$0.44 \mathrm{\ pF}$	C_7	$0.01 \ \mathrm{pF}$	C_8	$0.24 \mathrm{\ pF}$	C_9	$0.15 \mathrm{\ pF}$
C_{10}	$0.5 \ \mathrm{pF}$	C_{11}	$0.18 \mathrm{\ pF}$	C_{12}	1.6 pF	C_{13}	0.02 pF
C_{14}	$0.6 \ \mathrm{pF}$	C_{15}	$0.47 \ \mathrm{pF}$	C_{16}	$0.8 \ \mathrm{pF}$	C_{17}	$0.8 \ \mathrm{pF}$
R_1	$1~{ m K}\Omega$	R_2	$47 \ \Omega$	L_1	$430~\mu\mathrm{m}$	L_2	$157~\mu\mathrm{m}$
L_3	$510~\mu{\rm m}$	L_4	$520~\mu{\rm m}$	L_5	$500~\mu{\rm m}$	L_6	$94~\mu{\rm m}$
L_7	$405~\mu{\rm m}$	L_8	$438~\mu{\rm m}$	L_9	$481~\mu\mathrm{m}$	L_{10}	$462~\mu\mathrm{m}$
L_{11}	$100~\mu{\rm m}$	L_{12}	$384~\mu\mathrm{m}$	L_{13}	$148~\mu{\rm m}$	L_{14}	$362~\mu{\rm m}$
L_{S1}	$35~\mu{ m m}$	L_{S2}	$2~\mu{ m m}$	L_{S3}	$36~\mu{ m m}$	L_{S4}	$23~\mu{ m m}$
L_{S5}	$305~\mu{\rm m}$	L_{S6}	$452~\mu\mathrm{m}$				

Table 6.2: Optimized values of designed Switchless-LNA components

6.5 Performance Analysis

The LNA is designed and built utilising RF 65nm process technology by Magnachip Hynix Samsung. The chip is fabricated layer by layer with substrates and metal oxide metal (MoM) capacitors. Copper interconnects with six layers were fabricated using nanoclustering silica. These six layers, along with the other components of the LNA, are stacked on silicon wafers to give flexibility and thereby prevent parasitic losses caused by the CMOS mm-wave process. The procedure of stacking many layers is used to mount passive components on the silicon chip. Figure 6.23 shows a microchip die picture of the produced LNA. The chip's entire core area is $0.61 \times 0.92mm^2$. Fullwave simulations and measurements of LNA performance parameters have defined performance parameters with optimal dimensions to achieve high performance over a wideband of operation.



Figure 6.23: Microchip die photo of fabricated LNA

6.5.1 Experimental Verification

For on-wafer LNA, S-parameters and noise performance are measured using microprobe sets such as Karl Suss (KSM) microprobe system, PNA-X (N5245B) Keysight's technologies up to 50GHz, and Cascade Microtech microprobe tips, with 100μ m pitch. Prior to performing measurements, the PNA and micro-probes must be calibrated. Microprobe systems have been calibrated using the short-open-load-through technique.



Figure 6.24: Measurement setup of the LNA

Gold short circuits (Short), gold open pads (Open), and gold plus thin film resistors (Load or Match) are included in the calibration sets. Figure 6.24 illustrates the complete measurement setup for the PNA-X. PNA-X employs a more precise cold source method than the conventional Y-factor method, in which the NF of the device under test (DUT) is determined from two independent measurements. The first measurement is the DUT's available gain. This is accomplished with extreme precision by measuring vector air corrected S-parameters. The second measurement is the noise power emanating from the DUT's output, with the input set at room temperature (25^oC). This noise figure option for the PNA-X has an integrated low noise receiver with three distinct gain levels. This enables the testing of a wide variety of devices with any combination of gain and NF without the need for additional hardware.

Vector Air corrected S-parameters compensate for the mismatch between the test system's unsatisfactory source match and the DUT's input impedance. When measuring the DUT's output noise power, mismatch correction is also implemented. The standard impedance tuner module is used to adjust the source mismatch created by the KSM probes and microprobe tips. Accurate 50Ω NF is estimated by adjusting the source impedance applied to the DUT's input and measuring the resulting NF. The Pasternack PE6085 50 Ω impedance convertor adapter connector is used to ensure that the output of S_{22} is properly matched. The comparison of state-of-art LNA with other reported ones is shown in Table 6.3 In Figure 6.25 and figure 6.26, the dual-band



Figure 6.25: Simulated smith chart plot of proposed LNA Input Impedance

LNA's input and output impedances are shown as a smith chart respectively. It is worth noting that the input impedance begins in the capacitive region denoted as m3 at 18GHz and ends at the actual axis at 20GHz. It then follows the inductive zone and makes another contact with the real axis at 28 GHz and 31 GHz, shown by the symbol m2. It then passes through the capacitive zone again and comes to rest on the real axis at the final frequency 40GHz denoted by m1. The output impedance is capacitive dominant from the starting frequency of 18GHz, denoted by m6, to 25GHz, denoted by m4. As a result, the impedance in the K band (18-24 GHz) is capacitive, whereas the impedance in the Ka band (25-40 GHz) is inductive. LNA stability is measured and plotted in Figure 6.27. It is noted that the designed reconfigurable switchless LNA is unconditionally stable in the desired frequency range of 18 to 40 GHz. The proposed reconfigurable switchless LNA's measured and simulated NF and S-parameters in all three modes, dual-band mode, high-band mode, and low-band mode, are illustrated here. The NF fluctuates between 1.05dB at 18GHz and 1.2dB



Figure 6.26: Simulated smith chart plot of proposed LNA Output Impedance



Figure 6.27: Stability factor K of LNA

at 40GHz in dual-band mode, with a minimum of roughly 0.9dB at 21GHz, as shown in Figure 6.28. As illustrated in Figure 6.29, the gain of 27dB is attained with only



Figure 6.28: Noise Figure of the designed dual-band LNA



Figure 6.29: Simulated and measured S-parameters of the designed LNA in dual-band mode

a 0.2dB fluctuation and a 2dB variation across the entire frequency ranges of 18 to 24GHz and 24 to 40GHz. When compared to the previously reported works, this is phenomenal performance in terms of gain for having such a little fluctuation across



Figure 6.30: Simulated group delay and phase response of the designed LNA in dualband mode



Figure 6.31: Measured S-parameters and NF of the designed LNA in high-band mode



Figure 6.32: Measured S-parameters and NF of the designed LNA in low-band mode

the full 22GHz bandwidth. The return loss (S_{11}) is substantially below 10dB, preventing reflection. Other parameters, such as S_{12} and S_{22} , also perform satisfactorily. The simulated group delay and phase response of the designed LNA in dual-band mode is shown in Figure 6.30 also shows satisfactory performance. As illustrated in Figure 6.31, a consistent gain of 22dB with a 2dB variation is attained in high-band mode in Ka-band (25 to 40GHz). NF of around 1.5dB is attained with a return loss of less than 10dB. As illustrated in Figure 6.32, steady gain of 25dB with a variation of 1dB is attained in low-band mode (18 to 24GHz). A NF of around 1.5dB is attained with a return loss of less than 10dB.

The linearity analysis was used to determine the harmonics of the basic signal, such as third-order harmonics. Due to the fact that we are dealing with mm-wave bands up to 40GHz, we have included an analysis of second harmonics. Figure 6.33 and figure 6.34 show the measured values at 20GHz (K-band) and 35GHz (Ka-band). At 20GHz, the input 1dB compression point (IP_{1dB}) is -17dBm, the output 1dB compression point (OP_{1dB}) is +7.1dBm, the input intercept third-order point (IIP_3) is 0dBm, the output intercept third-order point (OIP_3) is +25dBm, and the input intercept second-order point (IP_{1dB}) is -16dBm, the output 1dB compression



Figure 6.33: Measured K-band Linearity performance



Figure 6.34: Measured Ka-band Linearity performance

point (OP_{1dB}) is +6.4dBm, the input intercept third-order point (IIP_3) is 0dBm, the output intercept third-order point (OIP_3) is +23dBm, and the input intercept

second-order point (IIP_2) is +5dBm. This value indicates that our reconfigurable switchless LNA architecture is more effective at suppressing harmonics and delivering good linearity to the input. This is accomplished by the use of optimised source degeneration with transmission lines at each CS stage to maintain a gain of less than 28dB and the suggested SSCL at the interstage and output to mitigate the impacts of intermodulation distortion (IMD). When the IMD decreases, the IIP_3 automatically increases, resulting in a more linear device.

6.6 Summary

This research describes the design and fabrication of a switch-free reconfigurable dual-band LNA working between 18GHz to 40GHz in 65nm CMOS technology. To achieve frequency reconfigurable capabilities, we theoretically investigated and built Suspended Substrate Coupled Lines (SSCL) in the interstage and output stage. The interstage SSCL divides the dual-band input signals into single-band amplifiers and then combines them in the output port via the output-stage SSCL. The projected LNA's unit transistor is designed in a Common-Source (CS) architecture and takes advantage of the inductive effect of transmission lines to degenerate the source while maintaining low noise and input matching. The fabricated $0.56mm^2$ LNA device has a measured small-signal gain of 27-27.2 dB with approximately 1dB NF in the 18-24 GHz band and a measured small-signal gain of 25-27dB with approximately 1.2dB NF in the 24-40 GHz band. The statistical and DoE analyses are presented to evaluate the design's yield and robustness. The LNA is highly linear, with an IP_{1dB} of -16dBm, an IIP_3 of 0dBm, an OP_{1dB} of +6.4dBm, and an OIP_3 of +25dBm. The proposed chip consumes 25.7mW only in the dual band. The comparison to the state of the art demonstrates that the developed LNA produces reasonably good mean values for all parameters, resulting in the highest FOM when used in modern receiver systems.

with state of art.
A performance
econfigurable LN
if the proposed re
Comparison o
Table 6.3:

- Not calculated

Chapter 7

CONCLUSION AND FUTURE WORKS

7.1 Conclusion

Chapter 1 and chapter 2 presented the LNA basics with the design trade-offs, the basic LNA topology, and the performance metrics. The performance metrics such as Noise, Noise figures, S-parameters, linearity analysis, and so on have been discussed briefly. In addition to that, two new analyses, such as statistical analysis and DoE analysis, have been elaborated on in this chapter.

Chapter 3 presented a high gain with good linearity low noise amplifier is designed using a suspended substrate line for a 5G wireless cellular front-end receiver system operating at 28-32GHz. The proposed LNA design used optimization techniques and achieved high-performance SoC IC solutions with EM compatibility. The LNA is fabricated using 65nm CMOS technology which operates at 1.2V, and the chip area is $0.35 \times 0.22mm^2$.

Chapter 4 presented a Quasi Circulator integrated low noise amplifier is designed for the entire X band covering 8-12GHz for high survivability RF front end systems. The designed QC-LNA achieves an excellent flat gain of 30dB with ultra-low-noise figure 1dB in the entire 4GHz bandwidth and provides good linearity. The LNA is fabricated using 65nm CMOS technology which operates at 1.2V, and the chip area is $0.84 \times 0.52mm^2$.

Chapter 5 presented a folded butterfly stub stepped impedance resonator integrated LNA designed for satellite navigation systems operating at 1.6-2.5GHz. The LNA is fabricated using PCB technology which operates at 3.3V, and the board area is $10.7 \times 2cm^2$. The LNA offers a decent gain of 22dB, a noise figure of 2.7dB, and good linearity. The FBSSIR is also fabricated separately and analyzed with the size of $3.3 \times 1.6cm^2$.

Chapter 6 presented a switchless reconfigurable low noise amplifier for full K/Kaband using suspended substrate coupled lines operating at 18-40GHz is designed for modern wireless systems. The designed LNA achieves high-performance SoC solutions such as the highest flat gain of 27dB and an ultra-low noise figure of 1.2dB (Maximum) in the entire 22GHz bandwidth. It also offers excellent linearity performance, especially in suppressing the third and second-order harmonics. The band shifting is controlled by the drain voltage of the K band and Ka-band stages. The LNA is fabricated using 65nm CMOS technology and consumes 25.7mW (in dual-band mode), with the chip area of $0.61 \times 0.92mm^2$.

7.2 Future Works

The proposed CMOS LNAs considered all the essential parameters it should consider but to design a single SoC RF front-end system, the separate LNA is not sufficient. All the fabricated LNAs are single entity SoC. In the future, the work can be focused on integrating the designed LNAs with other RF front-end components such as antenna, mixer, oscillator, Phase-locked loop, and so on. In this way, it is possible to test the designed LNAs' functionality in real-time, and also, we can study the issues that arise during the integration.

We can incorporate other microwave techniques into the design. Also, we can do work on making the lower nm technologies such as 45nm, 22nm, 7nm, and so on. It can also focus on the sub-mm bands.

Appendix I DERIVATION

A-1 Noise Figure Derivation

To examine the whole LNA's noise figure, the following method is used to transform any noisy multi-port network to an equivalent noise-free network plus standard current and voltage noises at the input port. When all internal noise sources are represented as current noises, a general noisy multi-port can be explained as a hybrid of a noisefree multi-port and some current noises at the ports, as illustrated in Figure A.1(a). Ports 1 and 2 denote the input and output ports, respectively, while Figure A.1(b) depicts the equivalent standard multi-port network. To find equivalent current and



Figure A.1: Noisy multi-port (a) its equivalent noiseless multi-port network (b) with equivalent current and voltage noise sources at the input.

voltage source noises at the input port, we describe the network by Y matrix:

$$Y = \begin{bmatrix} Y_{11} & Y_{12} & \cdots & Y_{1N} \\ Y_{21} & Y_{22} & \cdots & Y_{2N} \\ \vdots & \vdots & \cdots & \vdots \\ Y_{N1} & Y_{N2} & \cdots & Y_{NN} \end{bmatrix}$$
(A.1)

Now we define a new matrix

$$Y'' = \begin{bmatrix} Y_{21} & Y_{23} & \cdots & Y_{2N} \\ Y_{31} & Y_{33} & \cdots & Y_{3N} \\ \vdots & \vdots & \cdots & \vdots \\ Y_{N1} & Y_{N3} & \cdots & Y_{NN} \end{bmatrix}$$
(A.2)

Equivalent current and voltage noises in Figure A.1(b) are depicted as

$$\begin{cases} I_n = \frac{\Delta_{12}I_{n1} + \Delta_{22}I_{n2} + \dots \Delta_{N2}I_{nN}}{\Delta_{12}} \\ V_n = \frac{\Delta_{11}''I_{n2} + \Delta_{21}''I_{n3} + \dots \Delta_{N-1,1}'I_{nN}}{\Delta_{12}} \end{cases}$$
(A.3)

Where Δ_{ij} and Δ''_{ij} are the determinant of the adjoint matrix of the ij the element of the Y and Y'' matrix respectively.

Using the correlation coefficient I_n and V_n is correlated.

$$c = \overline{I_n V_n^*} \tag{A.4}$$

 I_n can be divided into two terms, one is correlated to V_n and the other is uncorrelated

$$I_n = I_{nc} + I_{nu} \tag{A.5}$$

$$\overline{I_n V_n^*} = \overline{(I_{nc} + I_{nu}) V_n^*} = \overline{I_{nc} V_n^*}$$
(A.6)

The correlation admittance Y_c is defined as:

$$Y_c = G_c + jB_c = \frac{I_{nc}}{V_n} \tag{A.7}$$

$$Y_c = \frac{c}{\overline{V_n^2}} \tag{A.8}$$

Equivalent noise resistance ${\cal R}_n$ is defined as

$$R_n = \frac{\overline{V_n^2}}{4kTB} \tag{A.9}$$

Equivalent noise conductance G_u is defined as

$$G_{u} = \frac{\overline{I_{nu}^{2}}}{4kTB} = \frac{\overline{I_{n}^{2}} - \overline{I_{nu}^{2}}}{4kTB} = \frac{\overline{I_{n}^{2}} - |Y_{c}|^{2} \overline{V_{n}^{2}}}{4kTB}$$
(A.10)

This expression is used in the above equation

$$\overline{I_{nu}}^2 = \overline{I_n}^2 - \frac{|c|^2}{\overline{V_n}^2} \tag{A.11}$$

The optimum source admittance $Y_{S_{-opt}}$ for minimum noise factor is calculated as:

$$Y_{S-opt} = \sqrt{G_c^2 + \frac{G_u}{R_n}} - jB_c \tag{A.12}$$

$$Y_{S_-opt} = \sqrt{\frac{\overline{I_n^2}}{\overline{V_n^2}} - B_c^2} - jB_c \tag{A.13}$$

The minimum noise figure F_{min} is calculated as:

$$F_{\min} = 1 + \frac{G_u + R_n \left(G_{\text{Sopt}} + G_c\right)^2}{G_{\text{Sopt}}}$$
 (A.14)

Finally, the noise factor F of the multi-port is calculated as:

$$F = 1 + \frac{G_u + R_n \left| Y_S + Y_c \right|^2}{G_S}$$
(A.15)

Bibliography

- Hayati , M., F. Daryabari , and S. Zarghami (2020). Ultra-wideband complementary metal-oxide semiconductor low noise amplifier using cs-cg noisecancellation and dual resonance network techniques. *IET Circuits, Devices and Systems*, 4(2), 200–208.
- Lee, M. and I. Kwon (2018). 3-10 ghz noise-cancelling cmos lna using gm-boosting technique. *IET Circuits, Devices and Systems*, **12**(1), 12–16.
- Lu, Z., R. Guan, and J. Zhou (2016). A practical design of x-band receiver front-end in 65-nm cmos. *IET Chinese Journal of Electronics*, **25**(3), 413–417.
- Nikbakhsh, M., E. Abiri , and H. Ghasemian (2018). Two-stage current-reused variable-gain low-noise amplifier for x-band receivers in 65 nm complementary metal oxide semiconductor technology. *IET Circuits, Devices and Systems*, 12(5), 630– 637.
- Adabi, E., B. Heydari, M. Bohsali, and A. M. Niknejad, 30 ghz cmos low noise amplifier. In 2007 IEEE Radio Frequency Integrated Circuits (RFIC) Symposium. 2007.
- Agiwal, M., A. Roy, and N. Saxena (2016). Next generation 5g wireless networks: A comprehensive survey. *IEEE Communications Surveys Tutorials*, 18(3), 1617– 1655.
- Ameen, H. A., K. Abdelmonem, M. A. Elgamal, M. A. Mousa, O. Hamada,
 Y. Zakaria, and M. A. Y. Abdalla, A 28 ghz four-channel phased-array transceiver in 65-nm cmos technology for 5g applications. *In 2017 29th International Conference on Microelectronics (ICM)*. 2017.

- Aneja, A., X. J. Li, and P. H. J. Chong (2021). Design and analysis of a 1.1 and 2.4 ghz concurrent dual-band low noise amplifier for multiband radios. AEU-International Journal of Electronics and Communications, 134, 153654.
- Arsalan, M. and F. Wu, S-band lna for future satellite navigation systems. In 2019 IEEE 11th International Conference on Communication Software and Networks (ICCSN). 2019.
- Arshad, S., R. Ramzan, K. Muhammad, and Q. ul Wahab (2015). A sub-10mw, noise cancelling, wideband lna for uwb applications. AEU - International Journal of Electronics and Communications, 69(1), 109–118. ISSN 1434-8411.
- Balalem, A., A. R. Ali, S. Amari, J. Machac, and A. Omar, Realization of a microstrip triple-mode bandpass filter using a square-loop resonator. In 2009 IEEE MTT-S International Microwave Symposium Digest. 2009.
- **Baltus, P.** (2004). *Minimum power design of RF front ends*. Ph.D. thesis, Electrical Engineering.
- Bisht, R., M. Akhtar, and S. Qureshi (2020). Design of reconfigurable multiband low-noise amplifiers for 802.11 ah/b/g and dcs-1800 applications. AEU-International Journal of Electronics and Communications, 120, 153201.
- Chaghaei, J., A. Jalali, and M. Meghdadi (2020). Low-power inductorless current-reuse lnas with active and passive gm-enhancement. *Microelectronics Journal*, 97, 104699.
- Chang, W., G.-I. Jeon, Y.-R. Park, S. Lee, and J.-K. Mun, X-band low noise amplifier mmic using algan/gan hemt technology on sic substrate. In 2013 Asia-Pacific Microwave Conference Proceedings (APMC). 2013.
- Chen, Y., K. Ma, and Y. Wang (2018). A ka-band substrate integrated suspended line to rectangular waveguide transition. *IEEE Microwave and Wireless Components Letters*, 28(9), 744–746.
- Choi, H.-W., C.-Y. Kim, and S. Choi (2021). 6.7–15.3 ghz, high-performance broadband low-noise amplifier with large transistor and two-stage broadband noise matching. *IEEE Microwave and Wireless Components Letters*, **31**(8), 949–952.

- Choi, K., H. Park, M. Kim, J. Kim, and Y. Kwon (2018). A 6–18-ghz switchless reconfigurable dual-band dual-mode pa mmic using coupled-line-based diplexer. *IEEE Transactions on Microwave Theory and Techniques*, 66(12), 5685–5695.
- Chu, Q.-X. and F.-C. Chen (2008). A compact dual-band bandpass filter using meandering stepped impedance resonators. *IEEE Microwave and Wireless Components Letters*, 18(5), 320–322.
- Chu, Y., K. Ma, and Y. Wang (2019). A novel triplexer based on sisl platform. *IEEE Transactions on Microwave Theory and Techniques*, **67**(3), 997–1004.
- Collin, R., Foundations for Microwave Engineering. IEEE Press Series on Electromagnetic Wave Theory. Wiley, 2001. ISBN 9780780360310. URL https://books.google.co.in/books?id=2SxoQgAACAAJ.
- da Costa, D. B. and H.-C. Yang (2020). Grand challenges in wireless communications. *Frontiers in Communications and Networks*, 1, 1. ISSN 2673-530X. URL https://www.frontiersin.org/article/10.3389/frcmn.2020.00001.
- Dai, R. (2017). A 0.6-v sub-mw x-band rfsoi cmos lna with novel complementary current-reused technique. International Journal of Circuit Theory and Applications, 45(12), 2046–2056.
- **Das, T.** (2013). Practical considerations for low noise amplifier design. Technical report, Freescale Semiconductors.
- Davulcu, M., C. Çalışkan, Kalyoncu, and Y. Gurbuz (2018). An x-band sige bicmos triple-cascode lna with boosted gain and p1db. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 65(8), 994–998.
- Djoumessi, E. E. and K. Wu, Dual-band low-noise amplifier using step-impedance resonator (sir) technique for wireless system applications. In 2009 European Microwave Conference (EuMC). 2009.
- El-Aassar, O. and G. M. Rebeiz (2020). Design of low-power sub-2.4 db mean nf 5g lnas using forward body bias in 22 nm fdsoi. *IEEE Transactions on Microwave Theory and Techniques*, 68(10), 4445–4454.

- Elkholy, M., S. Shakib, J. Dunworth, V. Aparin, and K. Entesari (2018). A wideband variable gain lna with high oip3 for 5g using 40-nm bulk cmos. *IEEE Microwave and Wireless Components Letters*, 28(1), 64–66.
- Fang, K. and J. F. Buckwalter (2019). Efficient linear millimeter-wave distributed transceivers in cmos soi. *IEEE Transactions on Microwave Theory and Techniques*, 67(1), 295–307.
- Fouad, H. (2001). Design and optimization of silicon rf front-ends for mobile communication systems. *Cairo*.
- Fouad, H., An rf cascode bjt-lna with shunt-series input matching. In Proceedings of the Twentieth National Radio Science Conference (NRSC'2003) (IEEE Cat. No.03EX665). 2003.
- Ghaneei Aarani, S., M. Hayati, and A. H. Kazemi (2019). A novel uwb low-noise amplifierdesign using double capacitor cross-coupled feedback. AEU - International Journal of Electronics and Communications, 104, 1–9. ISSN 1434-8411.
- Gonzalez, G., Microwave Transistor Amplifiers: Analysis and Design. Prentice Hall, 1997. ISBN 9780132543354. URL https://books.google.co.in/books?id= -AVTAAAMAAJ.
- Gray, P., P. Hurst, R. Meyer, and S. Lewis, ANALYSIS AND DESIGN OF ANA-LOG INTEGRATED CIRCUITS, 4TH ED. Wiley India Pvt. Limited, 2008. ISBN 9788126515691. URL https://books.google.co.in/books?id=LE-3tAXHF7AC.
- Hagag, M. F., M. Abu Khater, and D. Peroulis, Balanced-balanced tunable filtering lna using evanescent-mode resonators. In 2019 IEEE MTT-S International Microwave Symposium (IMS). 2019.
- Hong, J.-S., H. Shaman, and Y.-H. Chun (2007). Dual-mode microstrip open-loop resonators and filters. *IEEE Transactions on Microwave Theory and Techniques*, 55(8), 1764–1770.
- Hou, Z., C. Liu, B. Zhang, R. Song, Z. Wu, J. Zhang, and D. He (2020). Dual-/tri-wideband bandpass filter with high selectivity and adjustable passband for 5g mid-band mobile communications. *Electronics*, 9(2), 205.

- Hsieh, J., T. Wang, and S. Lu (2014). A 1.5-mw, 2.4 ghz quasi-circulator with high transmitter-to-receiver isolation in cmos technology. *IEEE Microwave and Wireless Components Letters*, 24(12), 872–874.
- Hu, J., K. Ma, S. Mou, and F. Meng (2018). A seven-octave broadband lna mmic using bandwidth extension techniques and improved active load. *IEEE Transactions* on Circuits and Systems I: Regular Papers, 65(10), 3150–3161.
- Hu, J., K. Ma, S. Mou, and F. Meng (2019). Analysis and design of a 0.1–23 ghz lna mmic using frequency-dependent feedback. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 66(9), 1517–1521.
- Huang, B., K. Lin, and H. Wang (2009). Millimeter-wave low power and miniature cmos multicascode low-noise amplifiers with noise reduction topology. *IEEE Transactions on Microwave Theory and Techniques*, 57(12), 3049–3059.
- Iyer, B. and N. P. Pathak (2014). A concurrent dual-band lna for noninvasive vital sign detection system. *Microwave and Optical Technology Letters*, 56(2), 391–394.
- Jadormio, J., M. N. Afsar, and V. J. Koomson, Simulation, design and integration of 60 ghz gallium nitride transceiver into nano-ferrite circulator on-silicon cmos substrate. In 2017 IEEE 60th International Midwest Symposium on Circuits and Systems (MWSCAS). 2017.
- Jiang, W., W. Shen, T. Wang, Y. M. Huang, Y. Peng, and G. Wang (2016). Compact dual-band filter using open/short stub loaded stepped impedance resonators (oslsirs/sslsirs). *IEEE Microwave and Wireless Components Letters*, 26(9), 672–674.
- Kanar, T. and G. M. Rebeiz (2014). X- and k-band sige hbt lnas with 1.2- and 2.2- db mean noise figures. *IEEE Transactions on Microwave Theory and Techniques*, 62(10), 2381–2389.
- Karthick, M. and K. R. Kashwan, Design of irnss receiver antennae for smart city applications in india. In 2015 Global Conference on Communication Technologies (GCCT). 2015.

- Ke, Z., S. Mou, K. Ma, and F. Meng (2018). A 0.7/1.1-db ultra-low noise dualband lna based on sisl platform. *IEEE Transactions on Microwave Theory and Techniques*, 66(10), 4576–4584. ISSN 1557-9670.
- Keshavarz Hedayati, M., A. Abdipour, R. Sarraf Shirazi, C. Cetintepe, and
 R. B. Staszewski (2018). A 33-ghz ha for 5g wireless systems in 28-nm bulk cmos. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 65(10), 1460–1464.
 ISSN 1558-3791.
- Kumar, A. and N. P. Pathak, Reconfigurable concurrent dual-band low noise amplifier for noninvasive vital sign detection applications. In 2014 International Conference on Advances in Computing, Communications and Informatics (ICACCI). 2014.
- Kumar, A. and N. P. Pathak, Coupled stepped-impedance resonator (csir) based concurrent dual band filtering lna for wireless applications. In 2015 IEEE MTT-S International Microwave and RF Conference (IMaRC). 2015.
- Kumar, M. and V. K. Deolia (2019). Performance analysis of low power lna using particle swarm optimization for wide band application. AEU - International Journal of Electronics and Communications, 111, 152897. ISSN 1434-8411.
- Lee, D. and C. Nguyen (2021). Dual q/v-band sige bicmos low noise amplifiers using q-enhanced metamaterial transmission lines. *IEEE Transactions on Circuits* and Systems II: Express Briefs, 68(3), 898–902.
- Lee, J. and C. Nguyen (2018). A k and ka band concurrent dual-band singleended input to differential output low-noise amplifier employing a novel transformer feedback dual-band load. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 65(9), 2679–2690.
- Lee, J.-R., J.-H. Cho, and S.-W. Yun (2000). New compact bandpass filter using microstrip /spl lambda//4 resonators with open stub inverter. *IEEE Microwave* and Guided Wave Letters, 10(12), 526–527.
- Lee, T., The Design of CMOS Radio-Frequency Integrated Circuits. Cambridge University Press, 2004. ISBN 9780521835398. URL https://books.google.co.in/books?id=io1hL480qBsC.

- Li, C., O. El-Aassar, A. Kumar, M. Boenke, and G. M. Rebeiz, Lna design with cmos soi process-1.4db nf k/ka band lna. In 2018 IEEE/MTT-S International Microwave Symposium - IMS. 2018.
- Li, N., W. Feng, and X. Li (2017). A cmos 3–12-ghz ultrawideband low noise amplifier by dual-resonance network. *IEEE Microwave and Wireless Components Letters*, 27(4), 383–385.
- Li, Z., Y. Duan, Z. Li, Z. Wang, and Y. Zhuang (2020). A 0.3–6 ghz noise and distortion cancelling lna achieving an nf of 2.8±0.3 db in 0.18-μm sige bicmos process. *Microelectronics Journal*, 105, 104904.
- Liang, C.-J., C.-W. Chiang, J. Zhou, R. Huang, K.-A. Wen, M.-C. F. Chang, and Y.-C. Kuan, A tri (k/ka/v)-band monolithic cmos low noise amplifier with shared signal path and variable gains. In 2020 IEEE/MTT-S International Microwave Symposium (IMS). 2020.
- Liang, C.-J., Y. Zhang, C.-W. Chiang, R. Huang, J. Zhou, J. Du, K.-A. Wen, M.-C. F. Chang, and Y.-C. Kuan (2021). A k/ka/v triband single-signalpath receiver with variable-gain low-noise amplifier and constant-gain phase shifter in 28-nm cmos. *IEEE Transactions on Microwave Theory and Techniques*, 69(5), 2579–2593.
- Liao, C.-K., P.-L. Chi, and C.-Y. Chang (2007). Microstrip realization of generalized chebyshev filters with box-like coupling schemes. *IEEE Transactions on Microwave Theory and Techniques*, 55(1), 147–153.
- Liu, B., J. Zhou, and J. Mao (2011). A fully integrated low voltage (0.5 v) x-band cmos low noise amplifier. *Microwave and Optical Technology Letters*, 53, 17 20.
- Lu, Y., S.-h. Yang, and Y. Chen, The design of lna based on bjt working on 2.2–2.6ghz. In 2010 2nd International Conference on Signal Processing Systems, volume 1. 2010.
- Luo, S., L. Zhu, and S. Sun (2010). A dual-band ring-resonator bandpass filter based on two pairs of degenerate modes. *IEEE Transactions on Microwave Theory* and Techniques, 58(12), 3427–3432.

- Luo, S., L. Zhu, and S. Sun (2011). Compact dual-mode triple-band bandpass filters using three pairs of degenerate modes in a ring resonator. *IEEE Transactions* on Microwave Theory and Techniques, 59(5), 1222–1229.
- M Fan wei, K. L. Y. and J Tseng, Design and analysis of low noise amplifier for 5g wireless systems. In 2018 IEEE internal conference on consumer electronics, Taiwan. 2018.
- Ma, Z., K. Ma, S. Mou, and F. Meng (2017). Quasi-lumped-element filter based on substrate-integrated suspended line technology. *IEEE Transactions on Microwave Theory and Techniques*, 65(12), 5154–5161.
- Makimoto, M. and S. Yamashita (1980). Bandpass filters using parallel coupled stripline stepped impedance resonators. *IEEE Transactions on Microwave Theory* and Techniques, 28(12), 1413–1417.
- Medra, A., V. Giannini, D. Guermandi, and P. Wambacq, A 79ghz variable gain low-noise amplifier and power amplifier in 28nm cmos operating up to 125°c. In ESSCIRC 2014 - 40th European Solid State Circuits Conference (ESSCIRC). 2014.
- Meghdadi, M., H. Lotfi, and A. Medi (2020). A reconfigurable highly-linear cmos transceiver core chip for x-band phased arrays. AEU - International Journal of Electronics and Communications, 114, 152990. ISSN 1434-8411.
- Meghdadi, M., M. Piri, and A. Medi (2018). A highly linear dual-gain cmos low-noise amplifier for x -band. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 65(11), 1604–1608.
- Miller, G. (2008). Tuning, optimization and statistical design advanced design system manual. URL https://edadocs.software.keysight.com/display/ ads2009/Tuning.
- Molavi, R. (2005). On the design of wideband CMOS low-noise amplifiers. Ph.D. thesis, University of British Columbia. URL https://open.library.ubc.ca/collections/ubctheses/831/items/1.0065420.

- Nawaz, A. A., J. D. Albrecht, and A. Çağrı Ulusoy (2019). A ka/v bandswitchable lna with 2.8/3.4 db noise figure. *IEEE Microwave and Wireless Components Letters*, 29(10), 662–664.
- Nawaz, A. A., J. D. Albrecht, and A. Çağrı Ulusoy (2020). A 28-/60-ghz bandswitchable bidirectional amplifier for reconfigurable mm-wave transceivers. *IEEE Transactions on Microwave Theory and Techniques*, 68(7), 3197–3205.
- Nguyen, T.-K., C.-H. Kim, G.-J. Ihm, M.-S. Yang, and S.-G. Lee (2004). Cmos low-noise amplifier design optimization techniques. *Microwave Theory and Techniques*, *IEEE Transactions on*, **52**, 1433 – 1442.
- Nikandish, G. and A. Medi (2015). Transformer-feedback interstage bandwidth enhancement for mmic multistage amplifiers. *IEEE Transactions on Microwave Theory and Techniques*, **63**(2), 441–448.
- Park, S., S. Wang, and S. Hong (2021). A k-band dual-mode common gate crosssumming vg-lna with low phase variation. *IEEE Transactions on Circuits and* Systems II: Express Briefs, 68(7), 2438–2442.
- **Perrott**, M. (2005). High speed communication circuits and systems.
- Pozar, D., Microwave Engineering, 4th Edition. Wiley, 2011. ISBN 9781118213636. URL https://books.google.co.in/books?id=JegbAAAQBAJ.
- Qin, P. and Q. Xue (2017a). Compact wideband lna with gain and input matching bandwidth extensions by transformer. *IEEE Microwave and Wireless Components Letters*, 27(7), 657–659.
- Qin, P. and Q. Xue (2017b). Design of wideband lna employing cascaded complimentary common gate and common source stages. *IEEE Microwave and Wireless Components Letters*, 27(6), 587–589.
- Rappaport, T. S., S. Sun, R. Mayzus, H. Zhao, Y. Azar, K. Wang, G. N.
 Wong, J. K. Schulz, M. Samimi, and F. Gutierrez (2013). Millimeter wave mobile communications for 5g cellular: It will work! *IEEE Access*, 1, 335–349.

Razavi, B., RF Microelectronics. Pearson Education India, Second Edition, 2013.

- Reddy, K. V. (2019). Low power balun LNAs for narrow band and UWB applications. Ph.D. thesis, Electronics and Communication Engineering NITK Surathkal. URL https://idr.nitk.ac.in/jspui/handle/123456789/14495.
- Sanduleanu, M. A. T., Gang Zhang, and J. R. Long, 31-34ghz low noise amplifier with on-chip microstrip lines and inter-stage matching in 90-nm baseline cmos. In IEEE Radio Frequency Integrated Circuits (RFIC) Symposium, 2006. 2006.
- Schweber, B. (2018). Get the most out of exotic processes for 5g lnas. Technical report, Digi-Key Electronics.
- Schwindt, R., V. Kumar, O. Aktas, J.-W. Lee, and I. Adesida (2005). Algan/gan hemt-based fully monolithic x-band low noise amplifier. *physica status solidi* (c), 2, 2631 – 2634.
- Shahani, A., D. Shaeffer, and T. Lee (1997). A 12-mw wide dynamic range cmos front-end for a portable gps receiver. *IEEE Journal of Solid-State Circuits*, 32(12), 2061–2070.
- Shaheen, R. A., T. Rahkonen, and A. Pärssinen (2021). Millimeter-wave frequency reconfigurable low noise amplifiers for 5g. *IEEE Transactions on Circuits* and Systems II: Express Briefs, 68(2), 642–646.
- Shang, Y., D. Cai, W. Fei, H. Yu, and J. Ren, An 8mw ultra low power 60ghz direct-conversion receiver with 55db gain and 4.9db noise figure in 65nm cmos. In 2012 IEEE International Symposium on Radio-Frequency Integration Technology (RFIT). 2012.
- Shen, W., X.-W. Sun, and W.-Y. Yin (2009). A novel microstrip filter using three-mode stepped impedance resonator (tsir). *IEEE Microwave and Wireless Components Letters*, 19(12), 774–776.
- Shin, S.-C., J.-Y. Huang, K.-Y. Lin, and H. Wang (2008). A 1.5–9.6 ghz monolithic active quasi-circulator in 0.18 μm cmos technology. *IEEE Microwave and Wireless Components Letters*, 18(12), 797–799.
- Sun, S.-J., T. Su, K. Deng, B. Wu, and C.-H. Liang (2013). Shorted-ended stepped-impedance dual-resonance resonator and its application to bandpass filters. *IEEE Transactions on Microwave Theory and Techniques*, 61(9), 3209–3215.
- Van der Ziel, A., Noise in Solid State Devices and Circuits. Wiley, 1986. ISBN 9780471832348. URL https://books.google.co.in/books?id=RjJTAAAAMAAJ.
- Vecchi, F., M. Repossi, W. Eyssa, P. Arcioni, and F. Svelto (2009). Design of low-loss transmission lines in scaled cmos by accurate electromagnetic simulations. *IEEE Journal of Solid-State Circuits*, 44(9), 2605–2615.
- Vigilante, M. and P. Reynaert, 20.10 a 68.1-to-96.4ghz variable-gain low-noise amplifier in 28nm cmos. In 2016 IEEE International Solid-State Circuits Conference (ISSCC). 2016.
- Wang, K. and H. Zhang (2020). A 22-to-47 ghz 2-stage lna with 22.2 db peak gain by using coupled l-type interstage matching inductors. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 67(12), 4607–4617.
- Wang, S. and C.-C. Teng (2016). An x-band cmos low-noise variable-gain amplifier with interstage reflection-type attenuator. *Microwave and Optical Technology Letters*, 58, 196–199.
- Wang, Y. and K. Ma (2019). Low-loss sisl patch-based phase shifters and the mixer application. *IEEE Transactions on Microwave Theory and Techniques*, 67(6), 2302– 2312.
- Wang, Y., K. Ma, N. Yan, and L. Li (2017). A slow-wave rat-race coupler using substrate integrated suspended line technology. *IEEE Transactions on Components*, *Packaging and Manufacturing Technology*, 7(4), 630–636.
- Wolff, I. (1972). Microstrip bandpass filter using degenerate modes of a microstrip ring resonator. *IEEE Electron. Letters*, 8(12).
- Xie, C., Z. Yu, and C. Tan (2020). An x/ku dual-band switch-free reconfigurable gaas lna mmic based on coupled line. *IEEE Access*, 8, 160070–160077.
- Yang, S.-L., Q. Xue, and K.-M. Luk (2005). A wideband low noise amplifiers design incorporating the cmrc circuitry. *IEEE Microwave and Wireless Components Letters*, 15(5), 315–317.
- Yu, X. and N. M. Neihart (2013). Analysis and design of a reconfigurable multimode low-noise amplifier utilizing a multitap transformer. *IEEE Transactions on Microwave Theory and Techniques*, **61**(3), 1236–1246.

- Yu, Y., H. Liu, Y. Wu, and K. Kang (2017). A 54.4–90 ghz low-noise amplifier in 65-nm cmos. *IEEE Journal of Solid-State Circuits*, 52(11), 2892–2904.
- Yu, Y., J. Zhu, Z. Zong, P. Tang, H. Liu, C. Zhao, Y. Wu, and K. Kang (2021). A 21-to-41-ghz high-gain low noise amplifier with triple-coupled technique for multiband wireless applications. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 68(6), 1857–1861.
- Yueting, Y., Z. Qi, L. Yunlong, and J. Xiaowei, An inexact quasi-newton algorithm combined with jacobian restart technique for nonlinear equation systems. In 2012 Fifth International Joint Conference on Computational Sciences and Optimization. 2012.
- Zhang, J. and D. Zhao, A broadband 1-db noise figure gaas low-noise amplifier for millimeter-wave 5g base-stations. In 2018 International Conference on Microwave and Millimeter Wave Technology (ICMMT). 2018.
- Zhang, X.-C., Z.-Y. Yu, and J. Xu (2008). Design of microstrip dual-mode filters based on source-load coupling. *IEEE Microwave and Wireless Components Letters*, 18(10), 677–679.
- Zhang, X. Y., J.-X. Chen, Q. Xue, and S.-M. Li (2007). Dual-band bandpass filters using stub-loaded resonators. *IEEE Microwave and Wireless Components Letters*, 17(8), 583–585.
- Zhen-Qi Huang and Quan-Rang Yang (1988). U-band shield suspended-stripline gunn dro and vco. *IEEE Transactions on Microwave Theory and Techniques*, 36(12), 1685–1694.
- Zhou, J., X. Zhang, Y. Yu, L. Zhang, C. Zhao, H. Liu, and K. Kang, A 24–30 ghz cmos lna with 2.05db nf and 0.6db in-band gain ripple for 5g-applications. In 2018 IEEE MTT-S International Wireless Symposium (IWS). 2018.
- Zhu, L. and W. Menzel (2003). Compact microstrip bandpass filter with two transmission zeros using a stub-tapped half-wavelength line resonator. *IEEE Microwave* and Wireless Components Letters, 13(1), 16–18.
- Zhu, L. and K. Wu (1999). A joint field/circuit model of line-to-ring coupling structures and its application to the design of microstrip dual-mode filters and

ring resonator circuits. *IEEE Transactions on Microwave Theory and Techniques*, **47**(10), 1938–1948.

Çalışkan, C., I. Kalyoncu, M. Yazıcı, M. Kaynak, and Y. Gurbuz (2019). Ultra-low noise amplifier for x-band sige bicmos phased array applications. *IEEE Transactions on Circuits and Systems II: Express Briefs*, 66(9), 1507–1511.

Publications Based on the Thesis

Refereed International Journals:

- Vignesh R ; Pradeep Gorre ; Sandeep Kumar ; Hanjung Song "Performance Analysis of 65nm CMOS LNA using SSL Technique for 5G Cellular Front-end Receivers" International Journal of Electronics and Communications (AEU), Elsevier, Volume 127, 2020, 153470, ISSN 1434-8411. DOI: 10.1016/j.aeue.2020.153470.
- Vignesh R; Pradeep Gorre ; Sandeep Kumar ; Hanjung Song "Highly Robust X-Band Quasi Circulator Integrated LNA for High Survivability of RF Frontend Systems" International Journal of Circuit, Theory and Applications, Wiley, vol 49, issue 7, July 2021, page no 2170-2182. DOI: 10.1002/cta.3001.
- Vignesh R, Pradeep Gorre, Sandeep Kumar, "A novel wide bandwidth FB-SSIR integrated low noise amplifier for satellite navigational receiver system", Microelectronics Journal, Volume 117, 2021, 105288, ISSN 0026-2692. DOI: 10.1016/j.mejo.2021.105288.
- Vignesh R; Pradeep Gorre ; Sandeep Kumar ; Hanjung Song "A K/Kaband Switchless Reconfigurable 65nm CMOS LNA based on Suspended Substrate Coupled Line" in IEEE Access, vol. 10, pp. 33449-33464, 2022, DOI: 10.1109/ACCESS.2022.3162202.

International Conferences Proceedings:

 Vignesh R, P. Gorre, S. Kumar and H. Song, "A 28-32GHz CMOS LNA with Broadband Approach for 5G Mm-Wave Communication cells," 2019 IEEE Asia-Pacific Microwave Conference (APMC), 2019, pp. 485-487, doi: 10.1109/APMC46564.2019.9038463.

Book Chapters:

 "Techniques to improve Gain-Bandwidth 5G ICs" in book title "CMOS Analog IC Design for 5G and Beyond" ISSN 1876-1100, ISBN 978-981-15-9864-7, 2021, Contributed by Vignesh R and Sandeep Kumar, doi:10.1007/978-981-15-9865-4.

Bio-data

$\mathbf{NAME}: \mathrm{Vignesh}\ \mathrm{R}$

CONTACT DETAILS

Address	: S/o KS Rajhamanicgam, $1/288D4$,
	Vayakattu Street, Samusigapuram
	Rajapalayam via, Virudunagar Dist., Tamilnadu–626102
T	: 9677385841

⊠ : vigneshr1993@gmail.com

EDUCATIONAL QUALIFICATIONS

Doctor of Philosophy (Ph.D)		
National Institute of Technology Karnataka, Surathkal	2018–Till date	
Master of Technology (M.Tech)		
Indian Institute of Information Technology Design		
and Manufacturing, Kancheepuram	2016 - 2018	
Branch : Electronic System Design		
Bachelor of Engineering (B.E)		
Coimbatore Institute of Technology, Coimbatore	2011 - 2015	
Branch : Electronics and Communication Engineering		
Teaching Experience		
Assistant Professor	Jun–Dec 2018	
Department of Electronics and Communication Engineering		
Vignan University, Guntur, A.P.		